Max-Planck-Institut für Mathematik Bonn

Clifford invariants of line bundle-valued quadratic forms

by

Asher Auel



Max-Planck-Institut für Mathematik Preprint Series 2011 (33)

Clifford invariants of line bundle-valued quadratic forms

Asher Auel

Max-Planck-Institut für Mathematik Vivatsgasse 7 53111 Bonn Germany Department of Mathematics & CS Emory University 400 Dowman Drive NE W401 Atlanta, GA 30322 USA

CLIFFORD INVARIANTS OF LINE BUNDLE-VALUED QUADRATIC FORMS

ASHER AUEL

ABSTRACT. We construct an extension of the Clifford (also known as the Hasse–Witt or 2nd Stiefel–Whitney) invariant to similarity classes of line bundle-valued quadratic forms of even rank n = 2m and fixed discriminant δ on a scheme X (where 2 is invertible). This invariant resides in the étale cohomology group $H^2_{\acute{e}t}(X, \kappa^m_{\delta})$ with coefficients in a twisted group scheme of order four and it "interpolates" between the classical étale cohomological Clifford invariant of (\mathcal{O}_X -valued) quadratic forms and the 1st Chern class (modulo 2) of the value line bundle. We further relate this invariant to natural classes in the "involutive" Brauer group arising from the even Clifford algebra and Clifford bimodule. In rank ≤ 6 , we use the classification of line bundle-valued quadratic forms in terms of twisted norm and pfaffian forms to explicitly compute this invariant and give necessary and sufficient conditions for its vanishing.

Contents

1
5
6
12
22
35
41
47
49

INTRODUCTION

For a line bundle \mathscr{L} on a space X, the notion of a quadratic (or symmetric bilinear) form on X with values in \mathscr{L} dates back to the early 1970s. Geyer-Harder-Knebusch-Scharlau [44] introduced the notion of symmetric bilinear forms with values in the module of Kähler differentials over a global function field. This notion enabled a consistent choice of local traces in order to generalize residue theorems to nonrational function fields. For a smooth complete algebraic curve X, Mumford [77] introduced the notion of locally free \mathscr{O}_X -modules with pairings taking values in the sheaf of differentials ω_X to study theta characteristics. For a commutative ring R, Kanzaki [56] introduced the notion of a Witt group W(I) of quadratic forms with values in an invertible R-module I.

Implicit in these early developments are generalizations to algebraic varieties of the classical transfer (or trace) maps from the theory of quadratic forms over fields. The general context in which such transfer maps exist has recently been established by the work of Gille [45], Nenashev [78], [79], Calmès-Hornbostel [25], [26], and Balmer-Calmès [10]. If $f: X \to Y$ is a proper morphism of relative dimension d between connected, noetherian, regular Spec $\mathbb{Z}[\frac{1}{2}]$ -schemes of finite Krull dimension, then the total derived direct image functor gives rise to a transfer map,

$$f_*: GW^{i+d}(X, \omega_f) \to GW^i(Y, \mathscr{O}_Y),$$

between the shifted derived (or coherent) Grothendieck–Witt groups introduced by Balmer [6], [7], [8], and Walter [96]. Here, ω_f is the relative dualizing sheaf from Grothendieck duality theory. In particular, in order to define the transfer along a proper morphism $f : X \to Y$, one is forced to consider ω_f -valued forms on X. It's in this setting that Balmer [9, §1.4] and Walter [96, §10] considered the existence of cohomological invariants defined on general shifted and line bundle-valued Witt and Grothendieck-Witt groups.

Date: May 29, 2011.

Line bundle-valued forms have also emerged in the context of involutions on Azumaya algebras, initially to generalize theorems of Albert [1, Thm. 10.19] on the existence of involutions on central simple algebras. Saltman [88, Thm. 4.2] and Knus–Parimala–Srinivas [67, §3] showed that involutions on endomorphism algebras are adjoint to symmetric bilinear forms with valued in line bundles. This paved the way for a version of Jacobson's [52] (see also Tits [93, §4]) even Clifford algebra of an Azumaya algebra with orthogonal involution. In his thesis, Bichsel [18] (reported in Bichsel–Knus [17]) constructs an even Clifford algebra of a line bundle-valued form over an affine scheme, then Parimala–Sridharan [84] use étale descent to construct an even Clifford algebra of an Azumaya algebra with orthogonal involution. These constructions enable much of the theory in [69] to be generalized (with some care) to arbitrary base schemes. In this context, line bundle-valued forms should be viewed as intermediary objects between quadratic forms and algebras with orthogonal involution: they are "unsplit" versions of quadratic forms, yet they still represent orthogonal involutions on "split" algebras.

Kapranov [57, §4.1] also considered the *homogeneous* Clifford algebra of a quadratic form—the same as the *generalized* Clifford algebra (in the sense Bichsel–Knus [17]) or the *graded* Clifford algebra (in the sense of Caenepeel–van Oystaeyen [23])—to study the derived category of projective quadrics and quadric fibrations. This theme was later taken up and developed by Bondal–Orlov [20, §2], [19], Kuznetsov [71], and Bolognesi–Bernardara [12]. These authors consider line bundle-valued quadratic forms with possible degeneration.

Over Dedekind domains, regular line bundle-valued bilinear forms (so called *modular lattices*) have been studied by Bushnell [21], [22], generalizing results of Fröhlich [38]. Recent number theoretic developments have also seen the appearance of line bundle-valued forms. Bhargava's [13], [14], [15] classification of "rings of low rank" over \mathbb{Z} is used in his work [16] on the distribution of discriminants of number fields of fixed low degree. An approach to this classification over arbitrary base rings is initiated in Wood's thesis [97], where line bundle-valued forms (of higher degree) were essential to the generalization of Bhargava's results. Also see Ho's thesis [49]. In related developments, Venkata Balaji [5] and independently Voight [94], used Clifford algebras of (possibly degenerate) ternary line bundle-valued forms to classify "degenerate" quaternion algebras over arbitrary bases.

In a topological context, Holla–Nitsure [50], [51] study characteristic classes (i.e. cohomological invariants in singular cohomology) for possibly mildly degenerate line bundle-valued quadratic forms (on complex vector bundles) by computing the cohomology of the relevant classifying space.

While the cohomological invariant theory of quadratic forms is well developed, the theory for line bundle-valued forms is substantially less so. When \mathscr{L} is the trivial line bundle, one can use the work of Delzant [32], Laborde [72], or Jardine [53] to define invariants in étale cohomology analogous to the Stiefel–Whitney (or higher Hasse–Witt) invariants for quadratic forms over fields (see [91], for example). These invariants have been intensively studied by many people in various contexts, including Serre [89], Knus-Ojanguren [70], Parimala-Srinivas [85], Ojanguren-Parimala-Sridharan [80], Knus-Parimala-Sridharan [66], Esnault-Kahn-Viehweg [34], Jardine [54], Cassou-Noguès-Erez-Taylor [27], and Esnault-Kahn-Levine-Viehweg [35]. In these approaches, invariants are constructed is several ways: directly, in the case of the classical invariants; à la Grothendieck, using the étale cohomology ring of a grassmannian bundle of anisotropic vectors; or as characteristic classes, using the étale cohomology groups of the simplicial classifying scheme (or topos) of a standard orthogonal group. The étale cohomology of the anisotropic grassmannian can be computed because it's an inner form of projective space with a standard subvariety of isotropic lines removed. The étale cohomology of the classifying space can be computed by analogy with the topological case because the standard orthogonal group is the base change of a smooth affine group scheme on Spec $\mathbb{Z}[\frac{1}{2}]$ (even on Spec \mathbb{Z}). When \mathscr{L} is nontrivial, these approaches break down. When \mathscr{L} is not a square in the Picard group, there are no known calculations of the étale cohomology rings of the required spaces; the calculations of Holla–Nitsure [50] in the topological case indicate that this might be a daunting task. The orthogonal group of such an \mathscr{L} -valued form is *not* a twist of the standard orthogonal group by a cocycle representing a class in the image of the natural map $H^1_{\text{\acute{e}t}}(X, \mathbf{O}_n) \to H^1_{\text{\acute{e}t}}(X, \mathbf{PO}_n)$. As Serre famously points out, the torsors for a group have in general no relation to the torsors for such a form of the group.¹

In this work, we employ a different approach. While isometry classes of \mathscr{L} -valued forms are associated to torsors for a twisted form of the orthogonal group, their similarity classes are associated to torsors for a standard group \mathbf{GO}_n of orthogonal similitudes. One complication is the nonexistence of a natural "pin" double covering of \mathbf{GO}_{2m} (this is even true in the topological context). Our

¹ Serre [90, I, §5.5, Remarque] warns, "Par contre, $H^1(G, {}_aA)$ n'a en général aucune relation avec $H^1(G, A)$." Then later in [90, I, §5.7, Remarque 1], Serre feels the need to reiterate himself, "Ici encore, il est faux en général que $H^1(G, {}_cB)$ soit en correspondance bijective avec $H^1(G, B)$." This is indeed an important point.

CLIFFORD INVARIANTS

essential contribution is the realization that, while there is no natural double covering, there is a natural four-fold covering of \mathbf{GO}_{2m} by the Clifford group Γ_{2m} that becomes a central isogeny when restricted to the subgroup of proper similitudes, and which "interpolates" between the Kummer double covering of the multiplicative group and the classical pin covering of the orthogonal group (see §2.4). Such a central isogeny does not seem to exist in the literature (though see Joshi [55] for an ad hoc construction in rank 2). The kernel of this covering is the group scheme κ_{δ}^{m} mentioned in the abstract. The appearance of invariants with values in cohomology with coefficients in a group scheme of order 4 is perhaps natural in light of Berhuy's theory [11, Prop. 9] of descent of cohomological invariants. An example of this is the cohomology class of the Tits algebra of a central simple algebra with orthogonal involution, see [69, VII §31.11].

Finally, since our methods are mostly torsorial in nature, we restrict ourselves to the case where 2 is invertible. This eliminates the consideration of nonsmooth algebraic group schemes and allows us to work purely in the étale site. The consideration of characteristic 2 phenomena will appear elsewhere.

Main results. Let X be a noetherian separated scheme on which 2 is invertible. A line bundle-valued quadratic form on X consists of a locally free \mathcal{O}_X -module \mathscr{E} of finite rank, an invertible \mathcal{O}_X -module \mathscr{L} , and a quadratic map $q : \mathscr{E} \to \mathscr{L}$. Since we assume that 2 is invertible on X, we will not concern ourselves with the difference between line-bundle valued quadratic and symmetric bilinear forms. Consult §1 for a review of these notions.

When $\mathscr{L} = \mathscr{O}_X$, the generalization to étale cohomology of the (signed) discriminant $d_{\pm}(\mathscr{E}, q)$ and classical Clifford invariant $c(\mathscr{E}, q)$ yield important invariants of quadratic forms (\mathscr{E}, q) on schemes. When $\mathscr{L} \neq \mathscr{O}_X$, the discriminant immediately yields a generalization to line bundle-valued forms of even rank (see Parimala–Sridharan [84]), while the Clifford invariant—due to the nonexistence of a "full" Clifford algebra—does not. The main construction of this current work is an extension of the Clifford invariant to similarity classes of line bundle-valued quadratic forms.

For $n \geq 1$ and $f: Z \to X$ an étale quadratic morphism with isomorphism class $\delta \in H^1_{\text{ét}}(X, \mathbb{Z}/2\mathbb{Z}) \cong H^1_{\text{\acute{et}}}(X, \mu_2)$, denote by $\mathrm{GQ}_n^{\delta}(X)$ the set of similarity classes of regular line bundle-valued quadratic forms of rank n and discriminant δ . In §2.3, we construct a group scheme κ_n^Z on X together with canonical isomorphisms

$$\boldsymbol{\kappa}_n^Z \cong \begin{cases} \mathbf{R}_{Z/X}^1 \boldsymbol{\mu}_4 & \text{if } n \equiv 0, 1 \mod 4 \\ \mathbf{R}_{Z/X} \boldsymbol{\mu}_2 & \text{if } n \equiv 2, 3 \mod 4 \end{cases}$$

where $\mathbf{R}_{Z/X}$ and $\mathbf{R}_{Z/X}^1$ denote the Weil restriction functor and subfunctor of norm one elements (see §2.2), respectively. There is also a canonical exact sequence

(1)
$$1 \to \boldsymbol{\mu}_2 \xrightarrow{i} \boldsymbol{\kappa}_n^Z \xrightarrow{p} \boldsymbol{\mu}_2 \to 1$$

of group schemes on X. For the relationship between κ_n^Z and the center of the spin group, see §3.4.

Theorem 1 (Theorem 2.10). Let X be a scheme with 2 invertible. For each $n \ge 1$ and étale quadratic $f: Z \to X$ with class $\delta \in H^1_{\text{\acute{e}t}}(X, \mathbb{Z}/2\mathbb{Z})$, there exists a map, called the similarity Clifford invariant,

$$gc: \mathrm{GQ}^{\delta}_n(X) \to H^2_{\mathrm{\acute{e}t}}(X, \kappa^Z_n)$$

which "interpolates" between the classical étale cohomological Clifford invariant and the 1st Chern class (modulo 2) of the value line bundle. More precisely, via the exact sequence of étale cohomology groups

$$\cdots \to H^2_{\text{\'et}}(X, \boldsymbol{\mu}_2) \xrightarrow{i^2} H^2_{\text{\'et}}(X, \boldsymbol{\kappa}_n^Z) \xrightarrow{p^2} H^2_{\text{\'et}}(X, \boldsymbol{\mu}_2) \to \cdots$$

arising from (1), we have

$$i^2 c(\mathscr{E}, q) = gc(\mathscr{E}, q, \mathscr{O}_X), \qquad p^2 gc(\mathscr{E}, q, \mathscr{L}) = c_1(\mathscr{L}, \mu_2)$$

for every isometry class of \mathcal{O}_X -valued quadratic form (\mathcal{E}, q) and every similarity class of line bundlevalued quadratic form $(\mathcal{E}, q, \mathcal{L})$ of rank n and discriminant δ .

The similarity Clifford invariant is constructed as the coboundary map in nonabelian étale cohomology associated to a short exact sequence of sheaves of groups

$$1 \to \kappa_n^Z \to \mathbf{S} \mathbf{\Gamma}_n^Z \to \mathbf{G} \mathbf{S} \mathbf{O}_n^Z \to 1$$

where \mathbf{Sr}_n^Z and \mathbf{GSO}_n^Z are the even Clifford group and group of proper orthogonal similitudes of a fixed choice of \mathscr{O}_X -valued form of rank n and discriminant δ (see §2.4). This particular central isogeny does not seems to have a place in the existing literature, except in small ranks (see Joshi [55], for example). Here we use étale descent to identify the category of \mathbf{GSO}_n^Z -torsors on the étale site $X_{\text{\acute{e}t}}$ with the category of regular oriented (see §1.10) line bundle-valued quadratic forms of rank n and discriminant δ on X.

We prove that the invariant respects Grothendieck–Witt equivalence, and so defines an étale cohomological invariant on the second part of the "fundamental" filtration of the total quadratic Grothendieck–Witt group (see §2.7). Let $GW(X, \mathscr{L})$ denote the Grothendieck–Witt group of \mathscr{L} -valued quadratic forms on X and $GW^{\text{tot}}(X) = \bigoplus_{\mathscr{L}} GW(X, \mathscr{L})$ denote the total quadratic Grothendieck–Witt group (which is actually a group fibered over the Picard groupoid of X). The signed discriminant defines a (surjective) cohomological invariant

$$d_{\pm}: GW^{\mathrm{tot}}(X) \to H^1_{\mathrm{\acute{e}t}}(X, \boldsymbol{\mu}_2).$$

Define $GI_2^{\text{tot}}(X) = \bigoplus_{\mathscr{L}} GI_2(X, \mathscr{L})$ as the subgroup of $GW^{\text{tot}}(X)$ of classes of quadratic forms of rank $n \equiv 0 \mod 4$ and trivial discriminant.

Theorem 2 (Theorem 2.11). Let X be a scheme with 2 invertible. The similarity Clifford invariant respects Grothendieck–Witt equivalence and so defines an invariant

$$gc: GI_2^{\text{tot}}(X) \to H^2_{\text{\'et}}(X, \mu_4).$$

Furthermore, if $(\mathscr{E}, q, \mathscr{L})$ is a quadratic form of rank $n \equiv 0 \mod 4$ and trivial discriminant on X, then

$$gc((\mathscr{E}, q, \mathscr{L}) \perp H_{\mathscr{L}}(\mathscr{V})) = c(\mathscr{E}, q, \mathscr{L}) + i^2 c_1(\mathscr{V}, \mu_2)$$

for any \mathscr{L} -valued hyperbolic form $H_{\mathscr{L}}(\mathscr{V})$ with lagrangian \mathscr{V} of even rank.

In §3 we relate the similarity Clifford invariant to "Brauer classes" associated to the even Clifford algebra of line bundle-valued form defined in Bichsel [18] and Bichsel–Knus [17]. Parimala–Srinivas [85] define an "involutive" Brauer group $\operatorname{Br}^*(X)$ of Azumaya \mathscr{O}_X -algebras with involution of the first kind as well as a "Z/X-unitary involutive" Brauer group $\operatorname{Br}^*(Z/X)$ of Azumaya \mathscr{O}_Z -algebras with Z/X-unitary involution for $f: Z \to X$ étale quadratic (see §3.3). If $(\mathscr{E}, q, \mathscr{L})$ is a regular line bundle-valued quadratic form of even rank n, then its even Clifford algebra $\mathscr{C}_0(\mathscr{E}, q, \mathscr{L})$ —together with its canonical Clifford involution τ_0 —defines a class $[\widetilde{\mathscr{C}}_0(\mathscr{E}, q, \mathscr{L}), \tau_0]$ in $\operatorname{Br}^*(Z)$ if $n \equiv 0 \mod 4$ or in $\operatorname{Br}^*(Z/X)$ if $n \equiv 2 \mod 4$ (see 3.4). Here, the étale quadratic $f: Z \to X$ is the given by the center of the even Clifford algebra.

Theorem 3 (Theorems 3.13, 3.17). Let X be a scheme with 2 invertible, $(\mathscr{E}, q, \mathscr{L})$ a regular line bundle-valued quadratic form of even rank n and étale quadratic $f : Z \to X$ given by the center of its even Clifford algebra.

a) If $n \equiv 0 \mod 4$, then

$$\varphi^2 gc(\mathscr{E}, q, \mathscr{L}) = i^2 [\widetilde{\mathscr{C}}_0(\mathscr{E}, q, \mathscr{L}), \tau_0] + c_1(f^* \mathscr{L}, \boldsymbol{\mu}_4)$$

in $H^2_{\text{\'et}}(Z, \mu_4)$, where $i^2 : \operatorname{Br}^*(Z) \to H^2_{\text{\'et}}(Z, \mu_4)$ and $\varphi^2 : H^2_{\text{\'et}}(X, \kappa_n^Z) \to H^2_{\text{\'et}}(Z, \mu_4)$ are certain canonical comparison maps.

b) If $n \equiv 2 \mod 4$, then

$$\phi^2 gc(\mathscr{E}, q, \mathscr{L}) = [\mathscr{C}_0(\mathscr{E}, q, \mathscr{L})]$$

in $H^2_{\text{\acute{e}t}}(Z, \mathbf{G}_m)$, where $\phi^2 : H^2_{\text{\acute{e}t}}(X, \kappa_n^Z) \to H^2_{\text{\acute{e}t}}(Z, \mathbf{G}_m)$ is a certain canonical comparison map.

In §4 we proceed to exactly calculate the similarity Clifford invariant in terms of "involutive" cohomology classes arising from what we call *Clifford data*. A Clifford datum mixes the canonical involution of the even Clifford algebra together with so-called *torsion data* arising from the Clifford bimodule. One should think of a torsion datum as a generalization of an involution. A Clifford datum $(Z/X, \mathscr{A}, \sigma, \mathscr{P}, \varphi)$ combines an Azumaya algebra with involution (\mathscr{A}, σ) of the first kind (resp. of unitary type) together with a torsion datum $(\mathscr{A}, \mathscr{P}, \varphi)$ of unitary type (resp. of the first kind). The interaction of both these structures enables the explicit calculation of the similarity Clifford invariant.

A torsion datum $(\mathscr{A}, \mathscr{P}, \varphi)$ on X gives rise to a cohomology class $[\mathscr{A}, \mathscr{P}, \varphi]$ in $H^2_{\text{ét}}(Z, \mu_2)$ or in $H^2_{\text{ét}}(X, \mathbf{R}^1_{Z/X} \mathbf{G}_m)$, depending on whether it is of the first type or of unitary type, respectively.

Theorem 4. Let X be a scheme with 2 invertible and $(\mathcal{E}, q, \mathcal{L})$ a regular line bundle-valued quadratic form of even rank n and Clifford datum $(Z/X, \mathcal{A}, \sigma, \mathcal{P}, \varphi)$.

- a) If $n \equiv 2 \mod 4$ then $gc(\mathscr{E}, q, \mathscr{L}) = [\mathscr{A}, \mathscr{P}, \varphi]$ in $H^2_{\text{\'et}}(X, \kappa_n^Z) \cong H^2_{\text{\'et}}(Z, \mu_2)$
- b) If $n \equiv 0 \mod 4$ then $i^2gc(\mathscr{E}, q, \mathscr{L}) = [\mathscr{A}, \mathscr{P}, \varphi]$ in $H^2_{\text{\'et}}(X, \mathbf{R}^1_{Z/X}\mathbf{G}_m)$.

where $i^2: H^2_{\text{ét}}(X, \kappa_n^Z) \to H^2_{\text{\acute{e}t}}(X, \mathbf{R}^1_{Z/X} \mathbf{G}_m)$ is a certain canonical comparison map.

In the case where m is even, we also have an exact formula (see Theorem 4.12) expressing the similarity Clifford invariant in terms of Clifford data. To achieve this, we establish a concrete presentation for the cohomology group $H^2_{\text{ét}}(X, \mathbf{R}^1_{Z/X} \boldsymbol{\mu}_4)$, generalizing Colliot-Thélène–Gille–Parimala [28, Prop. 2.10], as a certain fibered product of involutive Brauer groups.

The results in §3 and §4 indicate that the (refined) Tits algebra captures the even Clifford algebra with it's canonical involution while the similarity Clifford invariant constructed here captures the Clifford bimodule and its associated involutive structures.

Finally, in §5 we use the classification of line bundle-valued quadratic forms of rank 2, 4, and 6 in terms of twisted norm and pfaffian functors to explicitly compute the similarity Clifford invariant and give necessary and sufficient conditions as to its vanishing. The classification of regular quadratic forms of low rank over rings was initiated by Kneser, Knus, Ojanguren, Parimala, Paques, and Sridharan [60], [62], [63], [61], [64], [65]. Bichsel [18] and Bichsel–Knus [17] provide an extension of this theory to line bundle-valued forms of trivial discriminant over rings. In the context of quadratic forms over schemes, *low rank* usually means of rank ≤ 6 . In this interval, the exceptional isomorphisms of Dynkin diagrams $A_1 = B_1 = C_1$, $D_2 = A_1^2$, $B_2 = C_2$, and $A_3 = D_3$, have beautiful reverberations in the theory of quadratic forms of rank 3, 4, 5, and 6, respectively. Now, a standard reference on this work is Knus [68, Ch. V]. Over fields, a wonderful reference is [69, IV §15]. Much of the work in §5 can be viewed as generalization, from fields to schemes, of facts from [69]. While much of the existing theory over rings can be globalized, a unified treatment using the category of Clifford data (see §4.3) appears in Auel [4].

Theorem 5 (Corollaries 5.6, 5.10, 5.14). Let X be a scheme with 2 invertible and $(\mathscr{E}, q, \mathscr{L})$ a regular line bundle-valued quadratic form of even rank n and étale quadratic $f : Z \to X$ arising from the center of the even Clifford algebra with ι the nontrivial element of the Galois group of Z/X.

- If n = 2 then $(\mathscr{E}, q, \mathscr{L})$ has trivial discriminant if and only if it's similar to a hyperbolic form $H_{\mathscr{L}}(\mathscr{V})$ for some invertible \mathscr{O}_X -module \mathscr{V} . Furthermore, $gc(\mathscr{E}, q, \mathscr{L})$ is trivial if and only if both \mathscr{L} and \mathscr{V} are squares in $\operatorname{Pic}(X)$.
- If n = 4 then $gc(\mathscr{E}, q, \mathscr{L})$ is trivial if and only if there exists a locally free \mathscr{O}_Z -module \mathscr{V} of rank 2 satisfying det $\mathscr{V} \cong \iota^* \det \mathscr{V}$ such that $(\mathscr{E}, q, \mathscr{L})$ is similar to the norm $N_{Z/X}(\mathscr{V}, \wedge, \det \mathscr{V})$ of the canonical skew-symmetric form $\mathscr{V} \otimes \mathscr{V} \xrightarrow{\wedge} \det \mathscr{V}$. Furthermore, if $(\mathscr{E}, q, \mathscr{L})$ has trivial discriminant, then it's similar to the canonical quadratic form $\mathscr{W} \otimes \mathscr{W}' \xrightarrow{\wedge \otimes \wedge} \det \mathscr{W} \otimes \det \mathscr{W}'$ for locally free \mathscr{O}_X -modules \mathscr{W} and \mathscr{W}' of rank 2 with det $\mathscr{W} \cong \det \mathscr{W}'$.
- If n = 6 then $gc(\mathscr{E}, q, \mathscr{L})$ is trivial if and only if there exists a regular line bundle-valued Z/X-hermitian form $(\mathscr{H}, h, \mathscr{M})$ of rank 4 with trivial hermitian discriminant such that $f^*(\mathscr{E}, q, \mathscr{L})$ is similar (with the pullback Z/X-hermitian structure) to the canonical quadratic form $\bigwedge^2 \mathscr{H} \xrightarrow{\wedge} \det \mathscr{H}$. Furthermore, if $(\mathscr{E}, q, \mathscr{L})$ has trivial discriminant, then it's similar to the canonical quadratic form $\bigwedge^2 \mathscr{H} \xrightarrow{\wedge} \det \mathscr{H}$ for a locally free \mathscr{O}_X -module \mathscr{W} of rank 4 with $\det \mathscr{W}$ a square in $\operatorname{Pic}(X)$.

While much of the motivation to construct such a theory of cohomological invariants for line bundle-valued quadratic forms (and eventually Azumaya algebras with orthogonal involution) comes from classification problems, there are many other potential applications. As a complement to a series of papers by Parimala, Scharlau, and Sridharan [82], [83], [81], the author [2] uses the similarity Clifford invariant to prove the validity of Merkurjev's theorem for line bundle-valued quadratic forms, where it was known to fail for (\mathcal{O}_X -valued) quadratic forms. There are also applications to the "orthogonal Riemann-Roch" problem on computing cohomological invariants of pushforwards in derived Grothendieck–Witt groups. Finally, a version of the four-fold covering of the orthogonal similitude group can be adapted to Galois representations—analogously to Deligne's [31] use of the spin group—to investigate obstructions to embedding problems and local constants associated to essentially self-dual Galois representations of orthogonal type.

Acknowledgments

Some of this material is adapted from the author's Ph.D. thesis under the supervision of Ted Chinburg, who is heartily thanked. Most of the final preparation of this document was accomplished at Emory University and at the Max Planck Institute for Mathematics in Bonn. At both institutions, the author enjoyed excellent working conditions and much stimulating conversation. This work has also been greatly encouraged by (and tremendously improved as a result of conversations with) Philippe Cassou-Noguès, Skip Garibaldi, Max Knus, Manuel Ojanguren, R. Parimala, V. Suresh, Martin Taylor, and Jean-Pierre Tignol. Author partially supported by National Science Foundation Mathematical Sciences Postdoctoral Research Fellowship grant DMS-0903039.

ASHER AUEL

1. Torsorial theory of line bundle-valued forms

Let X be a scheme and \mathscr{L} and invertible \mathscr{O}_X -module. An $(\mathscr{L}\text{-valued})$ bilinear form on X is a triple $(\mathscr{E}, b, \mathscr{L})$, where \mathscr{E} is a locally free \mathscr{O}_X -module of finite rank and $b: T^2\mathscr{E} = \mathscr{E} \otimes \mathscr{E} \to \mathscr{L}$ is an \mathscr{O}_X -module morphism, equivalently, b is a global section of $\mathscr{H}om(T^2\mathscr{E}, \mathscr{L})$. An \mathscr{L} -valued bilinear form is symmetric if b factors through the canonical epimorphism $T^2\mathscr{E} \to S^2\mathscr{E}$, equivalently, b is a section of $\mathscr{H}om(S^2\mathscr{E}, \mathscr{L}) \subset \mathscr{H}om(T^2\mathscr{E}, \mathscr{L})$. An \mathscr{L} -valued bilinear form is skew-symmetric if b factors through the canonical epimorphism $T^2\mathscr{E} \to \bigwedge^2\mathscr{E}$, equivalently, b is a section of $\mathscr{H}om(\bigwedge^2\mathscr{E}, \mathscr{L}) \subset$ $\mathscr{H}om(T^2\mathscr{E}, \mathscr{L})$. An $(\mathscr{L}\text{-valued})$ quadratic form on X is a triple $(\mathscr{E}, q, \mathscr{L})$, where \mathscr{E} is a locally free \mathscr{O}_X -module and $q: S_2\mathscr{E} \to \mathscr{L}$ is an \mathscr{O}_X -module morphism, equivalently, q is a global section of $\mathscr{H}om(S_2\mathscr{E}, \mathscr{L}) \cong S^2(\mathscr{E}^{\vee}) \otimes \mathscr{L}$. Here $S^2\mathscr{E}$ and $S_2\mathscr{E}$ denote the second symmetric power and submodule of symmetric second tensors of \mathscr{E} , respectively.

Conventions. Denote by $X_{\text{ét}}$ the (large) étale site of X. We consider many sheaves of groups on $X_{\text{ét}}$, such as $\mathbf{GL}(\mathscr{E})$ and $\mathbf{O}(\mathscr{E}, q, \mathscr{L})$, that depend on a regular line bundle-valued bilinear form $(\mathscr{E}, q, \mathscr{L})$. When writing exact sequences and commutative diagrams of such groups, we will often suppress this dependence. We will also suppress the dependence on the base X, when writing common sheaves of groups, such as \mathbf{G}_{m} and $\boldsymbol{\mu}_n$, that are defined over Spec Z.

If $\alpha : \mathbf{G} \to \mathbf{G}$ is a homomorphism of sheaves of groups on $X_{\text{\acute{e}t}}$, then denote by $\alpha^i : H^i_{\text{\acute{e}t}}(X, \mathbf{G}) \to H^i_{\text{\acute{e}t}}(X, \mathbf{G}')$ the induced map on étale cohomology, where we assume \mathbf{G} and \mathbf{G}' are abelian for $i \geq 2$.

All algebras and sheaves of algebras will be unital and associative. The sheaf of algebra automorphisms of an \mathcal{O}_X -algebra \mathscr{A} will be denoted by $\operatorname{Aut}(\mathscr{A})$. For a commutative \mathcal{O}_X -algebra \mathscr{Z} , and a \mathscr{Z} -module (resp. algebra) \mathscr{M} , we will denote by $\widetilde{\mathscr{M}}$ the associated \mathcal{O}_Z -module (resp. algebra) on $Z = \operatorname{Spec} \mathscr{Z}$. A finite étale morphism $f: Z \to X$ of degree 2 will be called *étale quadratic*.

1.1. Quadratic and symmetric bilinear forms. Over an arbitrary scheme, the above notion of quadratic form agrees with the "classical" notion. See also Swan [92, Lemma 2.1] and Wood [97, §2.6].

Lemma 1.1. Let $\mathcal{Q}uad(\mathscr{E}, \mathscr{L})$ be the Zariski sheaf associated with the presheaf assigning to $U \to X$ the abelian group of maps $q : \mathscr{E}|_U \to \mathscr{L}|_U$ that satisfy

- $q(a v) = a^2 q(v)$ on local sections a of \mathcal{O}_U and v of $\mathcal{E}|_U$, and
- the map $b_q: \mathscr{E}|_U \times \mathscr{E}|_U \to \mathscr{L}|_U$, defined locally on sections by b(v, w) = q(v+w) q(v) q(w), is \mathcal{O}_U -bilinear.

Then there is a canonical isomorphism of \mathscr{O}_X -modules $\mathscr{Q}uad(\mathscr{E},\mathscr{L}) \to \mathscr{H}om(S_2\mathscr{E},\mathscr{L}).$

There's also a canonical \mathscr{O}_X -module morphism $\Gamma^2 \mathscr{E} \to S_2 \mathscr{E}$ from the module of second divided powers, which is an isomorphism is \mathscr{E} is locally free, see Deligne [30, 5.5.2.5]. Composing the isomorphism constructed in Lemma 1.1 with the dual of this isomorphism yields an isomorphism $\mathscr{Q}uad(\mathscr{E},\mathscr{L}) \to \mathscr{H}om(\Gamma^2 \mathscr{E},\mathscr{L})$, yielding an additional general tensorial interpretation of quadratic forms. Also, there's a canonical isomorphism $S^2(\mathscr{E}^{\vee}) \otimes \mathscr{L} \to \mathscr{H}om(S_2 \mathscr{E},\mathscr{L})$.

As in the classical case, if 2 is invertible on X, then every quadratic form is the associated quadratic form of a (symmetric) bilinear form on X, i.e. the canonical \mathscr{O}_X -module morphism $\mathscr{H}om(S^2\mathscr{E}, \mathscr{L}) \to \mathscr{H}om(S_2\mathscr{E}, \mathscr{L})$ is an isomorphism. Note that this morphism is *not* in general equivariant with respect to the natural $\mathbf{GL}(\mathscr{E}) \times \mathbf{GL}(\mathscr{L})$ actions.

1.2. Adjoint morphism. An \mathscr{L} -valued bilinear form $b : T^2 \mathscr{E} \to \mathscr{L}$ has an \mathscr{O}_X -module *adjoint* morphism $\psi_b : \mathscr{E} \to \mathscr{H}om(\mathscr{E}, \mathscr{L})$ defined on sections by $v \mapsto (w \mapsto b(v \otimes w))$. A bilinear form b is called *regular* if ψ_b is an isomorphism of \mathscr{O}_X -modules. An \mathscr{L} -valued quadratic form $q : \mathscr{E} \to \mathscr{L}$ is called regular if its associated bilinear form b_q is regular.

On the category of coherent \mathscr{O}_X -modules, denote by $(-)^{\vee \mathscr{L}}$ the exact (contravariant) functor $\mathscr{H}om(-,\mathscr{L})$. There is a canonical evaluation morphism of functors

$$\mathrm{ev}^{\mathscr{L}}:\mathrm{id}\to ((-)^{\vee\mathscr{L}})^{\vee\mathscr{L}}$$

which is an isomorphism on the subcategory VB(X) of locally free \mathscr{O}_X -modules of finite rank. The triple $(VB(X), (-)^{\vee \mathscr{L}}, \operatorname{ev}^{\mathscr{L}})$ forms an *exact category with duality* in the language of Balmer [6].

1.3. Similarities and isometries. A similarity (transformation) or similitude between bilinear forms $(\mathscr{E}, b, \mathscr{L})$ and $(\mathscr{E}', b', \mathscr{L}')$ is a pair (φ, λ) of \mathscr{O}_X -module isomorphisms $\varphi : \mathscr{E} \to \mathscr{E}'$ and

 $\lambda: \mathscr{L} \to \mathscr{L}'$ such that either of the following (equivalent) diagrams,

(2)
$$\begin{array}{cccc} T^{2}\mathscr{E} \xrightarrow{b} \mathscr{L} & \mathscr{E} \xrightarrow{\psi_{b}} \mathscr{H}om(\mathscr{E},\mathscr{L}) \\ T^{2}\varphi & & & & & & & \\ T^{2}\varphi' & & & & & & & & \\ T^{2}\mathscr{E}' \xrightarrow{b'} \mathscr{L}' & & & & & & & & \\ T^{2}\mathscr{E}' \xrightarrow{b'} \mathscr{L}' & & & & & & & & & & & \\ \end{array}$$

of \mathscr{O}_X -modules commute, where $\lambda^{-1}\varphi^{\vee\mathscr{L}}(\psi) = \lambda^{-1}\circ\psi\circ\varphi$ on sections. Note that the commutativity of the left-hand diagram (2) takes on the familiar formula $b'(\varphi(v),\varphi(w)) = \lambda \circ b(v,w)$ on sections, and can be adapted to define similarity transformations between quadratic forms. A similarity transformation (φ, λ) is an *isometry* if $\mathscr{L} = \mathscr{L}'$ and λ is the identity map.

Denote by $\operatorname{Sim}(\mathscr{E}, b, \mathscr{L})$ (resp. $\operatorname{Iso}(\mathscr{E}, b, \mathscr{L})$) the presheaf of similitudes (resp. isometries) of a regular \mathscr{L} -valued quadratic or bilinear form $(\mathscr{E}, b, \mathscr{L})$. In fact, these presheaves are sheaves on $X_{\operatorname{\acute{e}t}}$ and are representable by smooth affine reductive group schemes over X (see Demazure–Gabriel [33, III §5.2.3]). When b is quadratic or symmetric, we denote this group by $\operatorname{GO}(\mathscr{E}, b, \mathscr{L})$ (resp. $\operatorname{O}(\mathscr{E}, b, \mathscr{L})$) and call them the orthogonal similitudes group (resp. orthogonal group) of $(\mathscr{E}, b, \mathscr{L})$. When b is skewsymmetric, we denote these groups by $\operatorname{GSp}(\mathscr{E}, b, \mathscr{L})$ (resp. $\operatorname{Sp}(\mathscr{E}, b, \mathscr{L})$). For $\epsilon \in H^0_{\operatorname{\acute{e}t}}(X, \mu_2)$ we will say that an \mathscr{L} -valued bilinear form is ϵ -symmetric if on each connected component X' of X, it is either symmetric and $\epsilon|_{X'} = 1$ or skew-symmetric and $\epsilon|_{X'} = -1$. We will often omit the dependence of the groups of similitudes and isometries on the form $(\mathscr{E}, q, \mathscr{L})$ when no confusion may arise. Even when these sheaves of groups are representable by schemes over X, we will still think of them as sheaves of groups on $X_{\operatorname{\acute{e}t}}$.

1.4. Torsor interpretations. For the abstract notion of a (right) torsor for a sheaf of groups on a site, see Giraud [47].

Proposition 1.2. Let X be a scheme. Let $(\mathcal{E}, b, \mathcal{L})$ be a fixed \mathcal{L} -valued ϵ -symmetric bilinear form of rank n on X.

- a) The groupoid of $\mathbf{Iso}(\mathscr{E}, b, \mathscr{L})$ -torsors is equivalent to the groupoid whose objects are regular \mathscr{L} -valued ϵ -symmetric bilinear forms of rank n and whose morphisms are isometries.
- b) The groupoid of $Sim(\mathcal{E}, b, \mathcal{L})$ -torsors is equivalent to the groupoid whose objects are regular line bundle-valued ϵ -symmetric bilinear forms of rank n and whose morphisms are similarity transformations.

Proof. For a), the statement is a twist of [33, III, §5.2.1]. For further details, and for b), see Auel [3, Appendix A]. \Box

Remark 1.3. If \mathscr{E} is a locally free \mathscr{O}_X -module of rank n, we will denote by $\mathbf{GL}(\mathscr{E})$ its general linear group on $X_{\text{\acute{e}t}}$. The sheaves of groups $\mathbf{GL}(\mathscr{E})$ and $\mathbf{GL}(\mathscr{E}')$ are isomorphic on $X_{\text{\acute{e}t}}$ if and only if $\mathscr{E}' \cong \mathscr{E} \otimes \mathscr{L}$ for some invertible \mathscr{O}_X -module \mathscr{L} . For any invertible \mathscr{O}_X -module \mathscr{L} , there's a canonical isomorphism $\mathbf{G}_m \cong \mathbf{GL}(\mathscr{L})$, through which we shall identify these groups. A subtlety inherent in this identification arises when considering the pointed set of isomorphism classes of $\mathbf{GL}(\mathscr{L})$ -torsors, for which the distinguished point is canonically associated to the isomorphism class of \mathscr{L} . A similar remark should be made for the identification $\mu_2 = \mathbf{O}(\mathscr{E}, b, \mathscr{L})$ where $(\mathscr{E}, b, \mathscr{L})$ is a regular form of rank 1. In what follows, we will keep track of these identifications, if not in our notation, then in any statement concerning torsors.

1.5. The multiplier sequence. The map assigning $(\varphi, \lambda) \mapsto \lambda$ on sections defines the *multiplier* coefficient homomorphism $\mu : \mathbf{Sim}(\mathscr{E}, b, \mathscr{L}) \to \mathbf{GL}(\mathscr{L}) = \mathbf{G}_{\mathbf{m}}.$

Proposition 1.4. For any scheme X with 2 invertible, the sequence of sheaves of groups,

 $1 \to \mathbf{Iso}(\mathscr{E}, b, \mathscr{L}) \to \mathbf{Sim}(\mathscr{E}, b, \mathscr{L}) \xrightarrow{\mu} \mathbf{G}_m \to 1,$

is exact on $X_{\text{\acute{e}t}}$ and is called the multiplier sequence.

Proof. The only part needing an argument is that μ is an epimorphism. This follows from the fact that μ restricted to the central subgroup of homotheties is the squaring map, i.e. there's a commutative diagram



of sheaves of groups on $X_{\text{ét}}$ with exact rows. Of course, the (top horizontal) Kummer sequence in exact on $X_{\text{ét}}$ if 2 is invertible.

Remark 1.5. The interpretation of the multiplier sequence on cohomology is as follows. If $(\mathscr{E}, b, \mathscr{L})$ is a fixed \mathscr{L} -valued ϵ -symmetric bilinear form of rank n on X, then the map

$$H^1_{\text{\acute{e}t}}(X, \mathbf{Iso}(\mathscr{E}, b, \mathscr{L})) \to H^1_{\text{\acute{e}t}}(X, \mathbf{Sim}(\mathscr{E}, b, \mathscr{L}))$$

takes the isometry class of a regular \mathscr{L} -valued ϵ -symmetric bilinear form of rank n on X to its similarity class. Under the identification $\mathbf{GL}(\mathscr{L}) = \mathbf{G}_{\mathrm{m}}$ the map

$$H^1_{\text{\acute{e}t}}(X, \operatorname{Sim}(\mathscr{E}, b, \mathscr{L})) \to H^1_{\text{\acute{e}t}}(X, \mathbf{G}_{\mathrm{m}}) \cong \operatorname{Pic}(X)$$

takes the similarity class of a regular \mathscr{L}' -valued symmetric bilinear form of rank n on X to the class of \mathscr{L}' in $\operatorname{Pic}(X)$.

1.6. Forms of odd rank. The theory of regular line bundle-valued bilinear forms of odd rank essentially reduces to the study of regular \mathcal{O}_X -valued bilinear forms.

Lemma 1.6. If \mathscr{L} is not a square in the Picard group $\operatorname{Pic}(X)$, then any regular \mathscr{L} -valued bilinear form has even rank.

Proof. Let $(\mathscr{E}, b, \mathscr{L})$ be a regular bilinear form of rank n. Comparing determinants yields an isomorphism $\det(E)^{\otimes 2} \cong \mathscr{L}^{\otimes n}$ of line bundles. Thus we see that either n must be even or \mathscr{L} is a square in $\operatorname{Pic}(X)$ (up to an n-torsion element, which is itself a square). An alternate proof in the symmetric case, using the even Clifford algebra (see §1.8), can be found in [17, Theorem 3.7].

Thus every regular line bundle-valued bilinear form of odd rank has values in the square of some line bundle. In fact, for symmetric forms, there is even a way to take a canonical square root of the value bundle, see Auel [3, §1.3]. This can also be seen as a consequence of the direct product structures $\mathbf{G}_{\mathbf{m}} \times \mathbf{SO}(\mathscr{E}, b, \mathscr{L}) \to \mathbf{GO}(\mathscr{E}, b, \mathscr{L})$ of orthogonal similitude groups in odd rank. In view of this fact, we will only consider regular line bundle-valued forms of even rank.

1.7. Metabolic forms. The notion of a metabolic form—the correct generalization to schemes of the notion of hyperbolic form—is necessary to the construction of the Grothendieck–Witt and Witt groups of schemes. The formalism of \mathcal{O}_X -valued metabolic forms over arbitrary schemes was first introduced by Knebusch [58, I §3] and then extended to the context of triangulated and exact categories with duality by Balmer [6], [7], and [8]. We will summarize the formalism of metabolic forms in the context of line bundle-valued symmetric bilinear forms.

Definition 1.7. Let $(\mathscr{E}, b, \mathscr{L})$ be a line bundle-valued symmetric bilinear form on X and $\mathscr{V} \xrightarrow{j} \mathscr{E}$ a locally free \mathscr{O}_X -submodule with locally free quotient. The $(\mathscr{L}$ -valued) orthogonal complement $\mathscr{V}^{\perp} \xrightarrow{j\perp} \mathscr{E}$ is the kernel of the composition $\mathscr{E} \xrightarrow{j^{\vee \mathscr{L}} \circ \psi_b} \mathscr{H}om(\mathscr{V}, \mathscr{L}).$

Note that the restriction of the adjoint map yields an isomorphism $\mathscr{V}^{\perp} \xrightarrow{\psi_b|_{\mathscr{V}^{\perp}}} \mathscr{H}om(\mathscr{E}/\mathscr{V},\mathscr{L}).$

Definition 1.8. Let $(\mathscr{E}, b, \mathscr{L})$ be a line bundle-valued symmetric bilinear form on X and $\mathscr{V} \xrightarrow{j} \mathscr{E}$ a locally free \mathscr{O}_X -submodule. Then \mathscr{V} is an $(\mathscr{L}$ -valued) lagrangian if $\mathscr{V} = \mathscr{V}^{\perp}$, i.e. if there's an isomorphism of short exact sequences,

of locally free \mathscr{O}_X -modules. An \mathscr{L} -valued symmetric bilinear form on X is called *metabolic* if it has an \mathscr{L} -valued lagrangian. An \mathscr{L} -valued quadratic form on X is called *metabolic* if it's associated bilinear form is metabolic. Any metabolic form is regular.

The \mathscr{L} -valued hyperbolic quadratic form $H_{\mathscr{L}}(\mathscr{V})$ with lagrangian \mathscr{V} is the quadratic form $(\mathscr{V} \oplus \mathscr{H}om(\mathscr{V}, \mathscr{L}), h, \mathscr{L})$ where h is the canonical evaluation pairing h(v, f) = f(v) on sections. We will often consider the associated \mathscr{L} -valued hyperbolic symmetric bilinear form as well.

We call an \mathscr{L} -valued metabolic form $(\mathscr{E}, b, \mathscr{L})$ a *split metabolic form* if some lagrangian $\mathscr{V} \to \mathscr{E}$ is a direct summand. If 2 is invertible on X, then any split metabolic form $(\mathscr{E}, b, \mathscr{L})$ with direct summand lagrangian \mathscr{V} is isometric to the hyperbolic form $H_{\mathscr{L}}(\mathscr{V})$, see [58, §3 Prop. 1].

Splitting principle for metabolic forms. Over an affine scheme, every (line bundle-valued) metabolic form is split. This follows by an adaptation of Knebusch [58, I §3, Corollary 1]. Over a general scheme, this is no longer the case. However, there's a splitting principle for metabolic forms, in analogy with the classical splitting principle for locally free sheaves.

Theorem 1.9. Let X be a scheme and let $(\mathscr{E}, b, \mathscr{L})$ be a metabolic form on X with lagrangian $\mathscr{V} \xrightarrow{j} \mathscr{E}$.

- a) There exists a morphism of schemes $f: Y \to X$ such that $f^*: H^i_{\text{ét}}(X, \mu_2) \to H^i_{\text{ét}}(X, \mu_2)$ is injective and $f^*(\mathscr{E}, b, \mathscr{L})$ is a split metabolic form with lagrangian $f^*\mathscr{V}$.
- b) If 2 is invertible on X, then there exists a morphism of schemes $f: Y \to X$ such that $f^*: H^i_{\text{\'et}}(X, \mu_2) \to H^i_{\text{\'et}}(Y, \mu_2)$ is injective and $f^*(\mathscr{E}, b, \mathscr{L})$ is isometric to an orthogonal sum of hyperbolic planes

$$H_{f^*\mathscr{L}}(\mathscr{V}_1) \perp \cdots \perp H_{f^*\mathscr{L}}(\mathscr{V}_m),$$

for invertible \mathcal{O}_X -modules $\mathcal{V}_1, \ldots, \mathcal{V}_m$ on Y.

Proof. As for a), this is a classical construction. Following Fulton [40, §2], let $\mathscr{V} \xrightarrow{j} \mathscr{E}$ be a lagrangian and let $\mathscr{P} \subset \mathscr{H}om(\mathscr{E}, \mathscr{V})$ be the subbundle whose sections over $U \to X$ are \mathscr{O}_U -module morphisms $\varphi : \mathscr{E}|_U \to \mathscr{V}|_U$ such that $j|_U \circ \varphi = \mathrm{id}_U$. Let $Y = \mathbb{V}(\mathscr{P})$ be the corresponding affine bundle, and $f: Y \to X$ the natural projection. Then $f^*\mathscr{E}$ has a tautological projection to $f^*\mathscr{V}$, which is a section of f^*j , hence $f^*(\mathscr{E}, b, \mathscr{L})$ is a split metabolic form with lagrangian $f^*\mathscr{V}$. It is a standard fact that f^* is an injection on cohomology for a Zariski locally trivial affine bundle.

Now b is a direct result of a) and the classical splitting principle for locally free sheaves. Also see Fulton [40, §2] for a proof using the isotropic flag bundle of a metabolic bundle. Pulling back to this flag variety yields the required splitting. Also see Esnault–Kahn–Viehweg [34, §5] for a proof using iterated projective bundles.

1.8. Even Clifford algebra. In his thesis, Bichsel [18] constructed an even Clifford algebra of a line bundle-valued quadratic form on an affine scheme. Bichsel–Knus [17], Caenepeel–van Oystaeyen [23] and Parimala–Sridharan [84, §4] gave alternate constructions, all of which we shall recall below. Let $(\mathscr{E}, q, \mathscr{L})$ be a line bundle-valued quadratic form on X, not necessarily assumed to be regular.

Splitting construction. Let $\mathcal{LL} = \bigoplus_{n \in \mathbb{Z}} \mathscr{L}^{\otimes n}$ (called the Laurent polynomial algebra or generalized Rees ring in the literature) and define $Y = \operatorname{Spec} \mathcal{LL}$. Then $p: Y \to X$ is identified with the total space of the line bundle \mathscr{L} on X with the zero section removed. There is a canonical identification $p^*\mathscr{L} = \mathscr{O}_Y$ defined using the multiplication in the Rees ring, with respect to which, $p^*(\mathscr{E}, q, \mathscr{L})$ has a natural structure of $(\mathscr{O}_Y$ -valued) quadratic form on Y. The quasi-coherent sheaf of \mathscr{O}_X -algebras $\mathscr{C}'(\mathscr{E}, q, \mathscr{L}) = p_*\mathscr{C}(p^*(\mathscr{E}, q, \mathscr{L}))$, where \mathscr{C} is the standard Clifford algebra functor, is called the generalized Clifford algebra in [17]. As an \mathcal{LL} -algebra, $\mathscr{C}'(\mathscr{E}, q, \mathscr{L})$ inherits a natural \mathbb{Z} -graded \mathscr{O}_X algebra structure. The even Clifford algebra $\mathscr{C}_0(\mathscr{E}, b, \mathscr{L})$ is defined to be its 0th degree submodule. This construction is due to Caenepeel–van Oystaeyen [23].

Tensorial construction. Let $T\mathscr{E} = \bigoplus_{n\geq 0} \mathscr{E}^{\otimes n}$ be the tensor algebra on \mathscr{E} . Define an ideal $\mathscr{J} \subset T\mathscr{E} \otimes_{\mathscr{O}_X} L\mathscr{L}$, locally generated by $v \otimes v \otimes 1_{L\mathscr{L}} - 1_{T\mathscr{E}} \otimes q(v)$ for sections v of \mathscr{E} . Define a \mathbb{Z} -grading on $T\mathscr{E} \otimes_{\mathscr{O}_X} L\mathscr{L}$ by putting \mathscr{E} in degree 1 and \mathscr{L} in degree 2. Then \mathscr{J} is a 2 \mathbb{Z} -graded ideal and $\mathscr{C}'(\mathscr{E}, q, \mathscr{L}) = T\mathscr{E} \otimes_{\mathscr{O}_X} L\mathscr{L}/\mathscr{J}$ is a \mathbb{Z} -graded (as well as a $\mathbb{Z}/2\mathbb{Z}$ -graded) \mathscr{O}_X -algebra, defining the generalized Clifford algebra. The even Clifford algebra is defined to be its 0th degree submodule. This construction is due to Bichsel–Knus [17, §3].

Alternatively, following [69, II Lemma 8.1], we can directly define ideals \mathcal{J}_1 and \mathcal{J}_2 of $T(\mathscr{E} \otimes \mathscr{E} \otimes \mathscr{L}^{\vee})$ locally generated by

$$v\otimes v\otimes f-f(q(v)), \quad ext{and} \quad u\otimes v\otimes f\otimes v\otimes w\otimes g-u\otimes w\otimes f(q(v))\cdot g,$$

respectively, for sections u, v, w of \mathscr{E} and f, g of \mathscr{L}^{\vee} . Then $\mathscr{C}_0(\mathscr{E}, q, \mathscr{L}) = T(\mathscr{E} \otimes \mathscr{E} \otimes \mathscr{L}^{\vee})/(\mathscr{J}_1 + \mathscr{J}_2)$.

Gluing construction. Choose a Zariski affine open cover, $\mathscr{U} = \{U_i\}_{i \in I}$, of X trivializing \mathscr{L} via $\varphi_i : \mathscr{O}_{U_i} \simeq \mathscr{L}|_{U_i}$. Then $a_{ij} = \varphi_{ij}^{-1} \varphi_{ij} \in \mathbf{G}_m(U_{ij})$ for $(i, j) \in I^2$ (where $U_{ij} = U_i \times_X U_j$ and $\varphi_{ij} = \varphi_i|_{U_{ij}}$, etc.) is a Čech 1-cocycle representing \mathscr{L} . For each $i \in I$, the composition,

$$\varphi_i^{-1} \circ b|_{U_i} : \mathscr{E}|_{U_i} \to \mathscr{L}|_{U_i} \to \mathscr{O}_{U_i}.$$

defines an \mathcal{O}_{U_i} -valued quadratic form on U_i , and for each $(i, j) \in I^2$, one checks that the identity map on $\mathscr{E}|_{U_{i,i}}$ defines a similarity transformation,

$$(\mathrm{id}|_{U_{ij}}, a_{ij}) : (\mathscr{E}|_{U_{ij}}, \varphi_j^{-1} \circ q|_{U_{ij}}, \mathscr{O}_{U_{ij}}) \to (\mathscr{E}|_{U_{ij}}, \varphi_i^{-1} \circ q|_{U_{ij}}, \mathscr{O}_{U_{ij}}).$$

These similarities lift to $\mathcal{O}_{U_{ij}}$ -algebra isomorphisms of the standard even Clifford algebras (see Knus [68, IV Prop. 7.1.1]),

$$\mathscr{C}_{0}(\mathrm{id}|_{U_{ij}}, a_{ij}) : \mathscr{C}_{0}(\mathscr{E}|_{U_{ij}}, \varphi_{j}^{-1} \circ q|_{U_{ij}}) \to \mathscr{C}_{0}(\mathscr{E}|_{U_{ij}}, \varphi_{i}^{-1} \circ q|_{U_{ij}}),$$

yielding a gluing datum constructing the even Clifford algebra $\mathscr{C}_0(\mathscr{E}, q, \mathscr{L})$. This construction is due to Parimala–Sridharan [84, §4]. The original construction in Bichsel [18], in the affine case, uses faithfully flat descent.

Remark 1.10. For a regular (\mathscr{O}_X -valued) quadratic form (\mathscr{H}, b) of rank n and trivial Clifford algebra, the homomorphism of sheaves of groups, $\mathbf{GO}(\mathscr{H}, b) \to \mathbf{Aut}(\mathscr{C}_0(\mathscr{H}, b))$ defined by lifting similarities to automorphisms of the even Clifford algebra, induces a map

$$H^1_{\text{\acute{e}t}}(X, \mathbf{GO}(\mathscr{H}, b)) \to H^1_{\text{\acute{e}t}}(X, \mathbf{Aut}(\mathscr{C}_0(\mathscr{H}, b)))$$

with the following interpretation: the class associated (by Proposition 1.2 and Proposition 3.3) to the similarity class of a line bundle-valued quadratic form of rank n maps to the isomorphism class of its even Clifford algebra.

Functorial properties. Some consequence of the splitting construction and the classical properties of the even Clifford algebra are (assuming that $(\mathscr{E}, q, \mathscr{L})$ is regular and of rank n on X):

- a) If n is odd, $\mathscr{C}_0(\mathscr{E}, q, \mathscr{L})$ is a central \mathscr{O}_X -algebra. If n = 2m is even, the center $\mathscr{Z}(\mathscr{E}, q, \mathscr{L})$ of $\mathscr{C}_0(\mathscr{E}, q, \mathscr{L})$ is an étale quadratic \mathscr{O}_X -algebra (i.e. the morphism $f : \operatorname{Spec} \mathscr{Z} = Z \to X$ is étale quadratic).
- b) If n is odd, $\mathscr{C}_0(\mathscr{E}, q, \mathscr{L})$ is an Azumaya \mathscr{O}_X -algebra of rank 2^{n-1} ; if n is even, $\mathscr{C}_0(\mathscr{E}, q, \mathscr{L})$ gives rise to an Azumaya \mathscr{O}_Z -algebra of rank 2^{n-2}
- c) There is a canonical embedding of locally free \mathcal{O}_X -modules

$$i: \mathscr{E} \otimes \mathscr{E} \otimes \mathscr{L}^{\vee} \to \mathscr{C}_0(\mathscr{E}, q, \mathscr{L})$$

and a unique canonical involution τ_0 of $\mathscr{C}_0(\mathscr{E}, q, \mathscr{L})$ that restricts to the naïve switch map on (the first two tensor factors of) $\mathscr{E} \otimes \mathscr{E} \otimes \mathscr{L}^{\vee}$.

d) Any similarity transformation $(\varphi, \lambda) : (\mathscr{E}, q, \mathscr{L}) \to (\mathscr{E}', q', \mathscr{L}')$ induces an \mathscr{O}_X -algebra isomorphism

$$\mathscr{C}_0(\varphi,\lambda):\mathscr{C}_0(\mathscr{E},q,\mathscr{L})\to\mathscr{C}_0(\mathscr{E}',q',\mathscr{L}').$$

e) Any regular bilinear form $(\mathcal{N}, n, \mathcal{N}^{\otimes 2})$ of rank 1, induces an \mathcal{O}_X -algebra isomorphism

$$\mathscr{C}_0(n \otimes \mathrm{id}) : \mathscr{C}_0(\mathscr{E}, q, \mathscr{L}) \to \mathscr{C}_0(\mathscr{N} \otimes \mathscr{E}, n \otimes q, \mathscr{N}^{\otimes 2} \otimes \mathscr{L})$$

f) For any morphism of schemes $g: X' \to X$, there's a canonical \mathcal{O}_X -module isomorphism

$$g^*\mathscr{C}_0(\mathscr{E},q,\mathscr{L}) \simeq \mathscr{C}_0(g^*(\mathscr{E},q,\mathscr{L})).$$

1.9. The discriminant and Arf invariant. The classical signed discriminant of a quadratic form generalizes to line bundle-valued bilinear forms of even rank, see Parimala–Srinivas [85, §2.2] and Parimala–Sridharan [84, §4]. We will review this construction here.

By a discriminant module (\mathcal{N}, b) on X we mean a regular \mathcal{O}_X -valued bilinear form $b: T^2 \mathcal{N} \to \mathcal{O}_X$ of rank 1. The group (under tensor product and with identity $\langle 1 \rangle : T^2 \mathcal{O}_X \to \mathcal{O}_X$ given by multiplication) of isometry classes of discriminant modules on X is canonically isomorphic to $H^1_{\text{ét}}(X, \mu_2)$, see Milne [76, III §4]. Given an étale quadratic $f: Z \to X$, let \mathcal{N} be the kernel of the trace map $f_* \mathcal{O}_Z \xrightarrow{\text{Tr}} \mathcal{O}_X$. Then the multiplication in the \mathcal{O}_X -algebra $f_* \mathcal{O}_Z$ induces a bilinear form $m|_{\mathcal{N}}: T^2 \mathcal{N} \to \mathcal{O}_X$, yielding a discriminant module on X, see Knus [68, III §4.2]. The X-isomorphism classes of étale quadratic $f: Z \to X$ is canonically isomorphic to the group $H^1_{\text{ét}}(X, \mathbb{Z}/2\mathbb{Z})$, and we've just defined a map $\chi^1: H^1_{\text{ét}}(X, \mathbb{Z}/2\mathbb{Z}) \to H^1_{\text{ét}}(X, \mu_2)$. This map is the homomorphism induced on cohomology from the canonical homomorphism of group schemes $\chi: \mathbb{Z}/2\mathbb{Z} \to \mu_2$ and is an isomorphism if 2 is invertible on X, see [68, III Prop. 4.2.4].

Let $(\mathscr{E}, b, \mathscr{L})$ be a regular \mathscr{L} -valued bilinear form of rank n on X. Applying the determinant functor to the adjoint morphism yields an \mathscr{O}_X -module isomorphism

$$\det \mathscr{E} \xrightarrow{\det \psi_b} \det \mathscr{H}om(\mathscr{E}, \mathscr{L}) \xrightarrow{\operatorname{can}} \mathscr{H}om(\det \mathscr{E}, \mathscr{L}^{\otimes n}),$$

of \mathcal{O}_X -modules, where the canonical isomorphism at right given by

 v_1

(3)
$$f_1 \wedge \dots \wedge f_n \mapsto \left(v_1 \wedge \dots \wedge v_n \mapsto \det(f_i(v_j))_{ij} \right)$$

on sections. Write det b for the $\mathscr{L}^{\otimes n}$ -valued bilinear form of rank 1 on X whose adjoint morphism is the composition can \circ det ψ_b . Note that under the above identification, det b is given by

$$\det \mathscr{E} \otimes \det \mathscr{E} \qquad \xrightarrow{\det b} \qquad \mathscr{L}^{\otimes n}$$
$$\wedge \dots \wedge v_n \otimes w_1 \wedge \dots \wedge w_n \longmapsto \det(b(v_i, w_j))_{ij}$$

on sections.

Definition 1.11. Let $(\mathscr{E}, q, \mathscr{L})$ be a regular \mathscr{L} -valued bilinear form of even rank n = 2m on X. The *unsigned discriminant form* of $(\mathscr{E}, b, \mathscr{L})$ is the discriminant module $d(\mathscr{E}, b, \mathscr{L}) = (\mathscr{L}^{\vee \otimes m} \otimes \det \mathscr{E}, d(b))$ given by the composition

$$d(b): T^2(\mathscr{L}^{\vee \otimes m} \otimes \det \mathscr{E}) \to \mathscr{L}^{\vee \otimes n} \otimes T^2 \det \mathscr{E} \xrightarrow{\operatorname{id} \otimes \det b} \mathscr{L}^{\vee \otimes n} \otimes \mathscr{L}^{\otimes n} \xrightarrow{\operatorname{ev}} \mathscr{O}_X,$$

The signed discriminant form is the discriminant module $d_{\pm}(\mathscr{E}, b, \mathscr{L}) = (\mathscr{L}^{\vee \otimes m} \otimes \det \mathscr{E}, (-1)^m d(b))$. The signed discriminant invariant is the isometry class of the signed discriminant form in $H^1_{\acute{e}t}(X, \mu_2)$.

Definition 1.12. The Arf covering of a line bundle-valued quadratic form $(\mathscr{E}, q, \mathscr{L})$ will refer to the morphism $f : \operatorname{Spec} \mathscr{L}(\mathscr{E}, q, \mathscr{L}) = Z \to X$. The Arf covering is étale quadratic if and only if $(\mathscr{E}, q, \mathscr{L})$ is regular (see [68, IV Prop. 4.8.9]), and then its isomorphism class [Z/X] in $H^1_{\text{ét}}(X, \mathbb{Z}/2\mathbb{Z})$ is called the Arf invariant.

Proposition 1.13. Let X be a scheme and $(\mathscr{E}, q, \mathscr{L})$ be a regular line bundle-valued quadratic form of even rank on X with Arf covering $f: Z \to X$. Then $\chi^1[Z/X] = d_{\pm}(\mathscr{E}, q, \mathscr{L})$.

Proof. The statement holds on any Zariski open cover of X trivializing \mathscr{L} by [68, IV Prop. 4.6.3]. Since the transition functions are similarity transformations, the statement glues via Knus [68, IV Prop. 7.1.2].

Any similitude $(\varphi, \lambda) : (\mathscr{E}, b, \mathscr{L}) \to (\mathscr{E}', b', \mathscr{L}')$ between forms of even rank on X induces an isometry $d(\varphi, \lambda) : d(\mathscr{E}, b, \mathscr{L}) \to d(\mathscr{E}', b', \mathscr{L}')$ between discriminant forms and an \mathscr{O}_X -algebra isomorphism $\mathscr{Z}(\varphi, \lambda) : \mathscr{Z}(\mathscr{E}, q, \mathscr{L}) \to \mathscr{Z}(\mathscr{E}', q', \mathscr{L}')$ between even Clifford algebra centers. These give rise to homomorphisms det $/\mu^m : \mathbf{GO}(\mathscr{E}, b, \mathscr{L}) \to \mathbf{O}(d(\mathscr{E}, b, \mathscr{L})) = \mu_2$ and $\Delta : \mathbf{GO}(\mathscr{E}, q, \mathscr{L}) \to \mathbf{Aut}(\mathscr{Z}(\mathscr{E}, q, \mathscr{L})) = \mathbb{Z}/2\mathbb{Z}$, respectively. The induced maps

$$(\det/\mu^m)^1 : H^1_{\acute{e}t}(X, \mathbf{GO}(\mathscr{E}, b, \mathscr{L})) \to H^1_{\acute{e}t}(X, \boldsymbol{\mu}_2)$$
$$\Delta^1 : H^1_{\acute{e}t}(X, \mathbf{GO}(\mathscr{E}, q, \mathscr{L})) \to H^1_{\acute{e}t}(X, \mathbb{Z}/2\mathbb{Z})$$

then have the following interpretation: $(\mathscr{E}', b', \mathscr{L}') \mapsto d_{\pm}(\mathscr{E}', b', \mathscr{L}') - d_{\pm}(\mathscr{E}, q, \mathscr{L})$ and $(\mathscr{E}', b', \mathscr{L}') \mapsto \mathscr{L}(\mathscr{E}', b', \mathscr{L}') - \mathscr{L}(\mathscr{E}, q, \mathscr{L})$, respectively.

Define the group of *proper* similitudes $\mathbf{GSO}(\mathscr{E}, q, \mathscr{L})$ to be the sheaf kernel of the homomorphism Δ . Then $\chi \circ \Delta = \det / \mu^m$, so that when 2 is invertible on X, $\mathbf{GSO}(\mathscr{E}, b, \mathscr{L})$ is also equal to the sheaf kernel of \det / μ^m .

When 2 is invertible on X, there's a short exact sequence of sheaves of groups

(4)
$$1 \to \mathbf{GSO}(\mathscr{E}, b, \mathscr{L}) \to \mathbf{GO}(\mathscr{E}, b, \mathscr{L}) \xrightarrow{\det/\mu} \mu_2 \to 1$$

called the *similitude discriminant sequence* fitting into a commutative diagram with exact rows and columns,



of sheaves of groups on $X_{\text{\acute{e}t}}$.

Lemma 1.14. Any \mathscr{L} -valued metabolic quadratic form $(\mathscr{E}, b, \mathscr{L})$ of rank n = 2m on X has trivial Arf invariant. Moreover, any choice of lagrangian $\mathscr{V} \xrightarrow{j} \mathscr{E}$ induces an \mathscr{O}_X -algebra isomorphism $\zeta_{\mathscr{V}} : \mathscr{O}_X \times \mathscr{O}_X \to \mathscr{L}(\mathscr{E}, q, \mathscr{L}).$

Proof. In the case that 2 is invertible on X, we can appeal to Proposition 1.13, then use a straightforward adaptation of Knebusch [58, IV Proposition 3.2] or [59, Satz 4.1.2] to line bundle-valued metabolic forms.

1.10. Proper torsor interpretation. Let \mathscr{Z} be an étale quadratic \mathscr{O}_X -algebra. An \mathscr{Z} -orientation on a line bundle-valued quadratic form $(\mathscr{E}', q', \mathscr{L}')$ is an \mathscr{O}_X -algebra isomorphism $\zeta' : \mathscr{Z} \to \mathscr{Z}(\mathscr{E}', q', \mathscr{L}')$. If $f : Z \to X$ is étale quadratic, we also say use the term Z/X-orientation to mean an $f_*\mathscr{O}_Z$ orientation. When 2 is invertible on X and $\mathscr{Z} = \mathscr{Z}(\mathscr{E}, q, \mathscr{L})$, then a \mathscr{Z} -orientation on $(\mathscr{E}', q', \mathscr{L}')$ is equivalent to an isometry $\zeta' : d_{\pm}(\mathscr{E}, q, \mathscr{L}) \to d_{\pm}(\mathscr{E}', q', \mathscr{L}')$ of discriminant modules, by Proposition 1.13.

Proposition 1.15. Let X be a scheme and $(\mathcal{E}, b, \mathcal{L})$ a regular \mathcal{L} -valued quadratic form of even rank n on X.

- a) The groupoid of $\mathbf{SO}(\mathscr{E}, b, \mathscr{L})$ -torsors is equivalent to the category whose objects are $\mathscr{Z}(\mathscr{E}, q, \mathscr{L})$ oriented regular \mathscr{L} -valued quadratic forms $((\mathscr{E}', b', \mathscr{L}), \zeta')$ of rank n, and whose morphisms between objects $((\mathscr{E}', b', \mathscr{L}), \zeta')$ and $((\mathscr{E}'', b'', \mathscr{L}), \zeta'')$ are isometries $\varphi : (\mathscr{E}', b', \mathscr{L}) \to (\mathscr{E}'', b'', \mathscr{L})$ such that $\zeta'' = \mathscr{Z}(\varphi) \circ \zeta'$.
- b) The groupoid of **GSO**(\mathscr{E} , b, \mathscr{L})-torsors is equivalent to the category whose objects are $\mathscr{Z}(\mathscr{E}, q, \mathscr{L})$ oriented regular line bundle-valued quadratic forms $((\mathscr{E}', b', \mathscr{L}'), \zeta')$ of rank n, and whose morphisms between objects $((\mathscr{E}', b', \mathscr{L}'), \zeta')$ and $((\mathscr{E}'', b'', \mathscr{L}''), \zeta'')$ are similarities $\varphi : (\mathscr{E}', b', \mathscr{L}') \to$ $(\mathscr{E}'', b'', \mathscr{L}'')$ such that $\zeta'' = \mathscr{Z}(\varphi) \circ \zeta'$.

Proof. This is a straightforward adaptation of Proposition 1.2.

2. Clifford invariants of line bundle-valued forms

In this section, after reviewing étale cohomological invariants related to 1st Chern classes (in §2.1), we will construct the four-fold covering of the orthogonal similitude group (in §2.3) that gives rise to an étale cohomological invariant of line bundle-valued quadratic forms extending the classical Clifford invariant. We then show that our invariant respects Grothendieck–Witt equivalence of line bundle-valued forms (in §2.7).

2.1. Chern classes. Let $l \geq 2$ be such that l is invertible on X. Recall Grothendieck's [48] construction of *Chern classes modulo* l in étale cohomology, $c_i(\mathscr{E}, \boldsymbol{\mu}_l) \in H^{2i}_{\text{\acute{e}t}}(X, \boldsymbol{\mu}_l^{\otimes i})$, of a locally free \mathscr{O}_X -module \mathscr{E} of finite rank. If \mathscr{E} is invertible, define $c_1(\mathscr{E}, \boldsymbol{\mu}_l) \in H^{2i}_{\text{\acute{e}t}}(X, \boldsymbol{\mu}_l)$ to be the image of the isomorphism class of \mathscr{E} under the coboundary map $\operatorname{Pic}(X) \cong H^{1}_{\text{\acute{e}t}}(X, \mathbf{G}_m) \to H^{2}_{\text{\acute{e}t}}(X, \boldsymbol{\mu}_l)$ arising from the Kummer sequence,

$$1 \rightarrow \boldsymbol{\mu}_l \rightarrow \mathbf{G}_{\mathrm{m}} \xrightarrow{l} \mathbf{G}_{\mathrm{m}} \rightarrow 1,$$

on $X_{\text{\acute{e}t}}$. For a general locally free \mathscr{O}_X -module, the construction is reduced to the case of invertible modules by the splitting principle. In particular, $c_1(\mathscr{E}, \mu_l) = c_1(\det \mathscr{E}, \mu_l) \in H^2_{\text{\acute{e}t}}(X, \mu_l)$. Note that in view of the long exact sequence in étale cohomology associated to the Kummer sequence, $c_1(\mathscr{E}, \mu_l)$ is trivial if and only if det \mathscr{E} is an *l*th power in the Picard group Pic(X).

For the rest of this section we will assume that 2 is invertible on X. The scalar multiplication homomorphism $m : \mathbf{GL}(\mathscr{L}) \times \mathbf{O}(\mathscr{E}, q, \mathscr{L}) \to \mathbf{GO}(\mathscr{E}, q, \mathscr{L})$, given by $(\lambda, \varphi) \mapsto (\lambda \circ \varphi, \lambda^2)$ on sections, yields a commutative diagram of sheaves of groups with exact rows

on $X_{\text{\acute{e}t}}$, where $\boldsymbol{\mu}_2 \to \mathbf{G}_{\text{m}} \times \mathbf{O}(\mathscr{E}, q, \mathscr{L})$ is the diagonal inclusion and the center vertical map is the projection onto the first factor. The commutativity of diagram (6) implies that the coboundary map of the first row $H^1_{\text{\acute{e}t}}(X, \mathbf{GO}(\mathscr{E}, q, \mathscr{L})) \to H^2_{\text{\acute{e}t}}(X, \boldsymbol{\mu}_2)$ has the following interpretation (via Proposition 1.2): the class associated to $(\mathscr{E}', b', \mathscr{L}')$ is mapped to $c_1(\mathscr{L}', \boldsymbol{\mu}_2) - c_1(\mathscr{L}, \boldsymbol{\mu}_2)$.

Remark 2.1. What about the 1st Chern class $c_1(\mathscr{E}, \mu_2)$ of the underlying locally free module? For regular line bundle-valued symmetric bilinear forms of even rank n = 2m, the discriminant and 1st Chern class (modulo 2) of the value line bundle determine the 1st Chern class of \mathscr{E} . Indeed, we have the equations det $\mathscr{E} = \mathscr{L}^m \otimes d(\mathscr{E}, b, \mathscr{L})$ in Pic(X) (see the proof of Lemma 1.6), and also

$$c_1(d(\mathscr{E}, b, \mathscr{L}), \boldsymbol{\mu}_2) = d(\mathscr{E}, b, \mathscr{L}) \cup (d(\mathscr{E}, b, \mathscr{L}) + (-1))$$

in $H^2_{\text{ét}}(X, \mu_2) \cong H^2_{\text{ét}}(X, \mu_2^{\otimes 2})$ due to Esnault–Kahn–Viehweg [34, Lemma 5.3].

CLIFFORD INVARIANTS

2.2. Weil restriction. Let $f : Z \to X$ be finite étale and **G** a sheaf of groups on $Z_{\text{ét}}$ represented by a Z-scheme, which by abuse of notation, we shall also denote by **G**. Then $f_*\mathbf{G}$ is a sheaf of groups on $X_{\text{ét}}$ represented by an X-scheme $\mathbf{R}_{Z/X}\mathbf{G}$, called the *Weil restriction* or *corestriction* of **G** through $f : Z \to X$. By further abuse of notation, we often write $\mathbf{R}_{Z/X}\mathbf{G}$ for the sheaf of groups $f_*\mathbf{G}$. A general reference on Weil restriction is [33, I §1.6.6].

The counit of adjunction provides a canonical morphism $\varepsilon_{\mathbf{G}} : f^{-1}\mathbf{R}_{Z/X}\mathbf{G} \to \mathbf{G}$ of sheaves of groups on $Z_{\text{\acute{e}t}}$ (or of Z-schemes), and the Weil restriction is universal for this property: given any sheaf of groups on $X_{\text{\acute{e}t}}$ (or X-scheme) T and a morphism $\phi : f^{-1}T \to \mathbf{G}$, there exists a unique morphism $\varphi : T \to \mathbf{R}_{Z/X}\mathbf{G}$ of sheaves of groups on $X_{\text{\acute{e}t}}$ (or X-schemes) such that $\varepsilon_{\mathbf{G}} \circ f^{-1}\varphi = \phi$.

If **G** is a sheaf of groups on $X_{\acute{e}t}$ representable by an X-scheme, then $f^{-1}\mathbf{G}$ is a sheaf of groups on $Z_{\acute{e}t}$ representable by the Z-scheme $Z \times_X \mathbf{G}$, and the unit of adjunction provides a canonical morphism of sheaves of groups on $X_{\acute{e}t}$ (or X-schemes) $\eta_{\mathbf{G}} : \mathbf{G} \to \mathbf{R}_{Z/X} f^{-1}\mathbf{G}$.

When **G** is abelian, there's a *norm* homomorphism $N = N_{Z/X} : \mathbf{R}_{Z/X} f^{-1}\mathbf{G} \to \mathbf{G}$ of sheaves of groups on $X_{\text{\acute{e}t}}$. The sheaf kernel of the norm homomorphism is denoted by $\mathbf{R}_{Z/X}^1 f^{-1}\mathbf{G}$, and there's an exact sequence

(7)
$$1 \to \mathbf{R}^1_{Z/X} f^{-1} \mathbf{G} \to \mathbf{R}_{Z/X} f^{-1} \mathbf{G} \xrightarrow{N} \mathbf{G} \to 1$$

of sheaves of abelian groups on $X_{\text{ét}}$.

Now let $f : Z \to X$ be étale quadratic with nontrivial automorphism ι . Then the cokernel of the counit of adjunction can be identified with $\mathbf{R}^1_{Z/X}\mathbf{G}_m$, and we have an exact sequence

(8)
$$1 \to \mathbf{G} \xrightarrow{\eta_{\mathbf{G}}} \mathbf{R}_{Z/X} f^{-1} \mathbf{G} \xrightarrow{\mathrm{id}/\iota} \mathbf{R}_{Z/X}^{1} f^{-1} \mathbf{G} \to 1$$

of sheaves of abelian groups on $X_{\text{\acute{e}t}}$, where we can identify the cokernel map with $x \mapsto x/\iota(x)$ for sections x of $\mathbf{R}_{Z/X} f^{-1} \mathbf{G}$.

2.3. Finite coverings of orthogonal similitude groups. We now construct the four-fold covering of the orthogonal similitude group by the Clifford group that will be used in defining the similarity Clifford invariant. Fixing a regular \mathcal{O}_X -valued quadratic form (\mathcal{H}, b) of even rank n = 2m, let $\mathscr{Z} = \mathscr{Z}(\mathcal{H}, b)$ with associated étale quadratic Arf covering $f : Z = \operatorname{Spec} \mathscr{Z} \to X$.

Let $\Gamma = \Gamma(\mathcal{H}, b)$ be the Clifford group of (\mathcal{H}, b) (see [34, §1.9]) and recall the vector representation $r : \Gamma \to \mathbf{O}$ and the Clifford norm $N : \Gamma \to \mathbf{G}_{\mathrm{m}}$. The cartesian product of the Clifford norm and the vector representation yields an exact sequence,

(9)
$$1 \to \boldsymbol{\mu}_2 \to \boldsymbol{\Gamma} \xrightarrow{N \times r} \mathbf{G}_{\mathrm{m}} \times \mathbf{O} \to 1,$$

of sheaves of groups on $X_{\text{\acute{e}t}}$, where $\mu_2 \to \Gamma$ denotes the canonical inclusion of constants. The exactness at right may be checked locally in the étale topology. Finally, the composite homomorphism $s = m \circ (N \times r) : \Gamma \to \mathbf{GO}$ is an epimorphism of sheaves of groups on $X_{\text{\acute{e}t}}$.

Definition 2.2. Define $\kappa = \kappa(\mathscr{H}, b)$ to be the sheaf kernel of the homomorphism $s : \Gamma \to \mathbf{GO}$. Equivalently, thinking of $\mu_2 \hookrightarrow \mathbf{O}$ as the subgroup of homotheties, κ is the sheaf defined by

$$\kappa(U) = \{ x \in \Gamma(h)(U) : N(x) \in \mu_2(U), \ r(x) \in \mu_2(U), \ N(x) = r(x) \}$$

on $X_{\text{ét}}$. The resulting exact sequence,

(10)
$$1 \to \boldsymbol{\kappa} \to \boldsymbol{\Gamma} \xrightarrow{s} \mathbf{GO} \to 1,$$

on $X_{\text{ét}}$ is called the *similitude Clifford sequence*.

By construction, the similitude Clifford sequence is an extension of the Kummer sequence by the pinor sequence, forming a *fundamental Clifford diagram*

of sheaves of groups with exact rows and columns on $X_{\text{\acute{e}t}}$. In particular, κ is a étale group scheme of order 4, which we shall now identify.

Proposition 2.3. Let (\mathcal{H}, b) be a regular \mathcal{O}_X -valued quadratic form of even rank n and Arf covering $f: Z \to X$. Then there's a group scheme isomorphism

$$\boldsymbol{\kappa} \cong \begin{cases} \mathbf{R}_{Z/X}^1 \boldsymbol{\mu}_4 & \text{if } n \equiv 0 \mod 4 \\ \mathbf{R}_{Z/X} \boldsymbol{\mu}_2 & \text{if } n \equiv 2 \mod 4 \end{cases}$$

Proof. Every symmetric bilinear form over a scheme with 2 invertible is locally diagonalizable in the Zariski topology. We can assume, without loss of generality, that X is connected. Let $\mathscr{U} = \{U_i \rightarrow X\}_{i \in I}$ be a Zariski open cover of X diagonalizing the symmetric bilinear form (\mathscr{H}, b) via isometries

$$\Phi_i: (\mathscr{O}_{U_i}^n, \langle b_i^1, \dots, b_i^n \rangle) \to (\mathscr{H}|_{U_i}, h|_{U_i})$$

for $b_i^1, \ldots, b_i^n \in \mathbf{G}_m(U_i)$ corresponding to orthogonal bases $e_i^1, \ldots, e_i^n \in \mathscr{O}_X^n(U_i)$. Let $d_i = b_i^1 \cdots b_i^n \in \mathbf{G}_m(U_i)$, so that there are isometries

$$\det \Phi_i : (\mathscr{O}_{U_i}, \langle d_i \rangle) \to \det(\mathscr{H}, b)|_{U_i}$$

For each $i, j \in I$, there are isometries

$$\varphi_{ij} = \Phi_{\underline{ij}}^{-1} \circ \Phi_{\underline{ij}} : (\mathscr{O}_{U_{ij}}^n, \langle b_j^1, \dots, b_j^n \rangle) \to (\mathscr{O}_{U_{ij}}^n, \langle b_i^1, \dots, b_i^n \rangle)$$

and

$$\det \varphi_{ij} = \det \Phi_{\underline{ij}}^{-1} \circ \det \Phi_{\underline{ij}} : (\mathscr{O}_{U_{ij}}, \langle d_j \rangle) \to (\mathscr{O}_{U_{ij}}, \langle d_i \rangle)$$

yielding the equality

(12)
$$(\det \varphi_{ij})^2 d_i = d_j$$

in $\mathbf{G}_{\mathrm{m}}(U_{ij})$.

Now let $p: \widetilde{X} \to X$ be the étale quadratic covering defined by the class of

$$\delta' = -(-1)^{n(n-1)/2} \delta \in H^1_{\text{ét}}(X, \mu_2)$$

For each $i \in I$, fix an isomorphism of $p^{-1}(U_i)$ with $\widetilde{U}_i = U_i \times_{\operatorname{Spec} \mathscr{O}_X(U_i)} \operatorname{Spec} \mathscr{O}_X(U_i)[\sqrt{-d_i}]$. Then $\widetilde{\mathscr{U}} = {\widetilde{U}_i}_{i \in I}$ is a Zariski open cover of \widetilde{X} and also can be considered as an étale cover of X. We'll identify $\widetilde{U}_{ij} = \widetilde{U}_i \times_X \widetilde{U}_j$ with $U_{ij} \times_{\operatorname{Spec} \mathscr{O}_X(U_{ij})} \operatorname{Spec} \mathscr{O}_X(U_{ij})[\sqrt{-d_i}, \sqrt{-d_j}]$, where $U_{ij} = U_i \cap U_j$. Then

$$\delta'_{ij} = -\det \varphi_{ij} \, d_j^{-1} \otimes \sqrt{d_i d_j} \in \boldsymbol{\mu}_2(\widetilde{U}_{ij})$$

defines a Čech 1-cocycle representing δ' . Clearly, any étale cover of X splitting \mathcal{H} and δ' is a refinement of such a cover. Finally, for each $i \in I$, define

$$\epsilon_i = \frac{1}{d_i} e_i^1 \cdots e_i^n \otimes \sqrt{-d_i} \in \Gamma_i(\widetilde{U}_i),$$

where Γ_i is the Clifford group of the diagonal form $(\mathscr{O}_{U_i}, \langle b_i^1, \ldots, b_i^n \rangle)$. We'll argue that

(13)
$$N(\epsilon_i) = -1 \in \mathbf{G}_{\mathbf{m}}(\widetilde{U}_i), \quad r(\epsilon_i) = -\mathrm{id} \in \mathbf{O}_i(\widetilde{U}_i),$$

so that $\epsilon_i \in \kappa_i(\widetilde{U}_i)$, where \mathbf{O}_i and κ_i are associated to the diagonal form. Recalling that in the Clifford algebra,

$$e_i^k e_i^l = \left\{ \begin{array}{cc} d_i & \text{if } k = l \\ -e_i^l e_i^k & \text{if } k \neq l \end{array} \right.$$

we now compute

$$N(\epsilon_i) = \epsilon_i \, \sigma(\epsilon_i) = \frac{1}{d_i^2} \, e_i^1 \cdots e_i^n \, e_i^n \cdots e_i^1 \otimes -d_i = -1 \otimes 1.$$

Also, for each $k = 1, \ldots, n$, we have

$$\epsilon_i e_i^k = \frac{1}{d_i} e_i^1 \cdots e_i^n e_i^k \otimes \sqrt{-d_i} = (-1)^{n-1} e_i^k \epsilon_i,$$

so that

$$r(\epsilon_i)(e_i^k) = I(\epsilon_i) e_i^k \epsilon_i^{-1} = (-1)^n \epsilon_i e_i^k \epsilon^{-1} = (-1)^{n-1} (-1)^n e_i^k \epsilon \epsilon^{-1} = -e_i,$$

hence $r(\epsilon_i) = -\mathrm{id}$.

Thus -1 and ϵ_i generate $\kappa_i(\widetilde{U}_i)$. Using the identity $\sigma(\epsilon) = (-1)^{n(n-1)/2}$, and equation (13), we find that

$$\epsilon^2 = -(-1)^{n(n-1)/2}$$

so that

$$\boldsymbol{\kappa}_i(\widetilde{U}_i) \cong \begin{cases} \mathbb{Z}/4\mathbb{Z} & \text{if } n \equiv 0,1 \mod 4\\ \mathbb{Z}/2\mathbb{Z} \times \mathbb{Z}/2\mathbb{Z} & \text{if } n \equiv 2,3 \mod 4 \end{cases}$$

CLIFFORD INVARIANTS

via either $\epsilon_i \mapsto 1$ or $\epsilon_i \mapsto (0,1)$ and $-1 \mapsto (1,0)$, depending on the rank modulo 4.

The isometry Φ_i induces an algebra isomorphism $\mathscr{C}(\Phi_i)$ of Clifford algebras, hence commutes with N and r and induces group scheme isomorphisms $\kappa_i \simeq \kappa|_{\widetilde{U}_i}$ and $\mathbf{O}_i \simeq \mathbf{O}|_{\widetilde{U}_i}$. We have deduced that κ is a form of the corresponding constant group of order 4, which is split by the étale cover $\widetilde{\mathscr{U}}$. We'll now argue that κ is split by $p: \widetilde{X} \to X$ by showing that the sections $\mathscr{C}(\Phi_i)(\epsilon_i) \in \kappa(\widetilde{U})$ glue to a section $\epsilon \in \kappa(\widetilde{X})$.

Lemma 2.4. Let \mathscr{H} be a free \mathscr{O}_X -module of rank n on X, (\mathscr{H}, b_1) and (\mathscr{H}, b_2) be regular quadratic forms with orthogonal bases e_1^1, \ldots, e_1^n and e_2^1, \ldots, e_2^n , and $\varphi : (\mathscr{H}, b_1) \to (\mathscr{H}, b_2)$ an isometry. Then

$$\mathscr{C}(\varphi)(e_1^1\cdots e_1^n) = \varphi(e_1^1)\cdots \varphi(e_1^n) = \det \varphi \, e_2^1\cdots e_2^n.$$

in the Clifford algebra of (\mathcal{H}, b_2) .

Now using equation (12) and Lemma 2.4, compute

$$\begin{aligned} \mathscr{C}(\varphi_{ij})(\epsilon_{\underline{i}j}) &= \frac{1}{d_{\underline{i}j}}\varphi_{ij}(e_{\underline{i}j}^{1})\cdots\varphi_{ij}(e_{\underline{i}j}^{n})\otimes\sqrt{-d_{\underline{i}j}} \\ &= \frac{1}{(\det\varphi_{ij})^{2}d_{\underline{i}j}}\det\varphi_{ij}e_{\underline{i}j}^{1}\cdots e_{\underline{i}j}^{n}\otimes\sqrt{-(\det\varphi_{ij})^{2}d_{\underline{i}j}} \\ &= \frac{1}{d_{\underline{i}\underline{j}}}e_{\underline{i}\underline{j}}^{1}\cdots e_{\underline{i}\underline{j}}^{n}\otimes\sqrt{-d_{\underline{i}\underline{j}}} = \epsilon_{\underline{i}\underline{j}} \end{aligned}$$

so that we have the equality

$$\mathscr{C}(\Phi_{\underline{i}\underline{j}})(\epsilon_{\underline{i}\underline{j}}) = \mathscr{C}(\Phi_{\underline{i}\underline{j}})(\mathscr{C}(\varphi_{ij})(\epsilon_{\underline{i}\underline{j}})) = \mathscr{C}(\Phi_{\underline{i}\underline{j}})(\epsilon_{\underline{i}\underline{j}})$$

of sections in $\Gamma(\widetilde{U}_{ij})$. Thus the sections $\mathscr{C}(\Phi_i)(\epsilon_i) \in \kappa(\widetilde{U}_i)$ glue to a section $\epsilon \in \kappa(\widetilde{X})$ which splits the group scheme κ over the étale quadratic covering $p: \widetilde{X} \to X$. Clearly, the Galois action on the section $\epsilon \in \kappa(\widetilde{X})$ is trivial if and only if the class δ' is trivial. Since any étale quadratic extension splits a unique nontrivial form of $\mathbb{Z}/4\mathbb{Z}$ or $\mathbb{Z}/2\mathbb{Z} \times \mathbb{Z}/2\mathbb{Z}$, the proposition follows. On the étale cover $\widetilde{\mathscr{U}}$, the isomorphism of group schemes constructed above descends.

Alternatively, we can also view the above equality as

$$\mathscr{C}(\Phi_{\underline{i}j})(\epsilon_{\underline{i}j}) = \left(\det\varphi_{ij}\,d_{\underline{i}j}^{-1} \otimes \sqrt{d_{\underline{i}\underline{j}}d_{\underline{i}j}}\right)\,\mathscr{C}(\Phi_{\underline{i}\underline{j}})(\epsilon_{\underline{i}\underline{j}}) = \delta'_{ij}\mathscr{C}(\Phi_{\underline{i}\underline{j}})(\epsilon_{\underline{i}\underline{j}})$$

so that $\boldsymbol{\kappa}$ is identified with the twist of $\mathbb{Z}/4\mathbb{Z}$ or $\mathbb{Z}/2\mathbb{Z} \times \mathbb{Z}/2\mathbb{Z}$ by the the Čech 1-cocyle δ'_{ij} on the étale cover $\widetilde{\mathscr{U}}$ on X.

Remark 2.5. For regular \mathcal{O}_X -valued quadratic forms of odd rank, we have

$$\boldsymbol{\kappa} \cong \begin{cases} \mathbf{R}_{Z/X}^1 \boldsymbol{\mu}_4 & \text{if } n \equiv 1 \mod 4 \\ \mathbf{R}_{Z/X} \boldsymbol{\mu}_2 & \text{if } n \equiv 3 \mod 4 \end{cases}$$

see Auel [3, Thm. 2.11]. This formula in the odd rank case depends on our convention for the Clifford norm N. We take the convention of Fröhlich [39, Appendix I]. The other convention, taken by Knus [68, IV §6.1], defines N via the *standard Clifford involution* induced by negation on \mathcal{H} .

Example 2.6. We continue to assume, without loss of generality, that X is connected. For certain standard quadratic forms, we now make the isomorphism in Proposition 2.3. Let $(\mathscr{H}, b) = H_{\mathscr{O}_X}(\mathscr{O}_X^m)$ be the hyperbolic quadratic form on X with trivial lagrangian, and $e_1, \ldots, e_m, f_1, \ldots, f_m$ a choice of global sections forming a hyperbolic basis, i.e. $h(f_l, e_k) = \delta_{lk}$. An explicit representation of $\kappa = \kappa(H_{\mathscr{O}_X}(\mathscr{O}_X^m))$ is as follows.

For *m* odd, a straightforward calculation shows that the global section $\epsilon = \prod_{l=1}^{m} (1 - 2e_l f_l) \in \Gamma(X)$ of the Clifford group generates, together with $-1 \in \Gamma(X)$, the group scheme κ . This yields a global isomorphism $\mu_2 \times \mu_2 = \langle -1 \rangle \times \langle \epsilon \rangle \cong \kappa$ of group schemes on $X_{\acute{e}t}$, depending on our particular choice of ϵ .

For *m* even, the section $\tilde{\epsilon} = (-1)^{m/2} \prod_{l=1}^{m} (1 - 2e_l f_l) \otimes \sqrt{-1} \in \Gamma(\tilde{X})$ of the Clifford group over the étale double covering $\tilde{X} = X \times_{\text{Spec } \mathscr{O}_X(X)} \text{Spec } \mathscr{O}_X(X)[\sqrt{-1}] \to X$ generates $\kappa|_{\tilde{X}}$. By descent there's an isomorphism $\mu_4 \simeq \kappa$ of group schemes on $X_{\text{\acute{e}t}}$. This isomorphism also depends on our particular choice of ϵ .

More generally, for each even $n = 2m \ge 2$ and each étale quadratic $f: Z \to X$, define $(\mathscr{H}, b) = h_n^Z = (f_* \mathscr{O}_Z, h^Z) \perp H_{\mathscr{O}_X}(\mathscr{O}_X^{m-1})$, where h^Z is the norm form associated to $f: Z \to X$ (see §5.1). Then h_n^Z has rank n, Arf invariant canonically isomorphic to $f: Z \to X$, and trivial classical Clifford invariant (see [68, V §2.3]). There's a canonical isometry $f^*h_n^Z \to H_{\mathscr{O}_Z}(\mathscr{O}_Z^m)$ (indeed, for m = 1 the canonical \mathscr{O}_Z -algebra isomorphism $f^*f_*\mathscr{O}_Z \to \mathscr{O}_Z \times \mathscr{O}_Z$ is an isometry of the quadratic forms $f^*(f_*\mathscr{O}_Z, h^Z) \to H_{\mathscr{O}_Z}(\mathscr{O}_Z)$, which extends, for general m, to the remaining hyperbolic factors by the identity). Hence we have isomorphisms $f^{-1}\kappa(h_n^Z) \cong \kappa(f^*h_n^Z) \cong \kappa(H_{\mathscr{O}_Z}(\mathscr{O}_Z^m)).$

For *m* odd, we compose with the projection onto the subgroup $\kappa(H_{\mathscr{O}_Z}(\mathscr{O}_Z^m)) \cong \mu_2 \times \mu_2 \to \langle -1 \rangle$ (this depends on our choice of ϵ , as above), yielding a morphism $f^{-1}\kappa(h_n^Z) \to \mu_2$, and hence by adjunction, a morphism $\kappa(h_n^Z) \to \mathbf{R}_{Z/X}\mu_2$, which we can check, locally using Proposition 2.3, is an isomorphism.

For m even, we have an isomorphism $f^{-1}\kappa(h_n^Z) \to \mu_4$, which by adjunction yields a morphism $\kappa(h_n^Z) \to \mathbf{R}_{Z/X} \mu_4$, which is seen to be in the kernel of the norm map $\mathbf{R}_{Z/X} \mu_4 \to \mu_4$. Thus we have a map $\kappa(h_n^Z) \to \mathbf{R}^1_{Z/X} \mu_4$, which again we can check is an isomorphism locally. Thus for the form h_n^Z , we have an explicit choice of isomorphism in Proposition 2.3.

2.4. The similarity Clifford invariant. By the calculations in the proof of Proposition 2.3, we see that any section of κ not in the canonical subgroup $\mu_2 \hookrightarrow \kappa$ has degree (in the Clifford group) the parity n. The structure theory of the center of the Clifford algebra then implies that the similitude Clifford sequence (10) is central for n odd. For n even, its restriction to the even Clifford group,

(14)
$$1 \to \kappa \to \mathbf{S}\Gamma \xrightarrow{s} \mathbf{G}\mathbf{S}\mathbf{O} \to 1,$$

is central and we consider the restriction of the fundamental diagram (11) to the even Clifford group: 1

1

$$1 \longrightarrow \mu_{2} \longrightarrow \operatorname{Spin} \xrightarrow{r} \operatorname{SO} \longrightarrow 1$$

$$i \downarrow \qquad \downarrow \qquad \downarrow \qquad \downarrow$$

$$1 \longrightarrow \kappa \longrightarrow \operatorname{S\Gamma} \xrightarrow{s} \operatorname{GSO} \longrightarrow 1$$

$$p \downarrow \qquad N \downarrow \qquad \downarrow \qquad \downarrow$$

$$1 \longrightarrow \mu_{2} \longrightarrow \operatorname{G_{m}} \xrightarrow{2} \operatorname{G_{m}} \longrightarrow 1$$

$$\downarrow \qquad \downarrow \qquad \downarrow \qquad \downarrow$$

$$1 \longrightarrow \mu_{2} \longrightarrow \operatorname{I} \qquad 1$$

1

Definition 2.7. For each n = 2m, étale quadratic $f : Z \to X$, and each sheaf of groups **G** depending on a quadratic form, write $\mathbf{G}_n^Z = \mathbf{G}(h_n^Z)$.

Now, we are ready to define the cohomological invariant we'll primarily be concerned with.

Definition 2.8. Let $f: Z \to X$ be étale quadratic. For a regular line bundle-valued quadratic form $(\mathscr{E}, q, \mathscr{L})$ of rank n = 2m and Arf invariant [Z/X], define the similarity Clifford invariant $gc(q) = gc(\mathscr{E}, q, \mathscr{L}) \in H^2_{\text{\'et}}(X, \kappa_n^Z)$ to be the image of $(\mathscr{E}, q, \mathscr{L}, \zeta)$ under the coboundary map $H^1_{\text{\acute{e}t}}(X, \mathbf{GSO}^Z_n) \to H^2_{\text{\acute{e}t}}(X, \kappa^Z_n)$ arising from the even Clifford sequence (14), for any choice of Z/Xorientation ζ . By Lemma 5.15, gc(q) does not depend on the choice of orientation ζ .

Remark 2.9. An analogous similarity Clifford invariant can also be defined for line bundle-valued quadratic forms of odd rank. By Lemma 1.6, the value line bundle is a square, and in fact, has a canonical square root. This can be used to compute the invariants odd rank forms in terms of classical invariants, see Auel $[3, \S 2.2.1]$.

2.5. Interpolation property. The similarity Clifford invariant "interpolates" between the classical étale cohomological Clifford invariant (see Parimala–Srinivas [85, Lemma 6] or Knus–Parimala– Sridharan [66, $\S3$]) and the 1st Chern class (modulo 2) (see $\S2.1$) of the value line bundle.

Theorem 2.10. Let X be a scheme with 2 invertible. Let $(\mathcal{E}, q, \mathcal{L})$ be a regular \mathcal{L} -valued quadratic form of rank n = 2m and Arf invariant [Z/X]. We have an exact sequence

$$\cdots \to H^2_{\text{\'et}}(X, \boldsymbol{\mu}_2) \xrightarrow{i^2} H^2_{\text{\'et}}(X, \boldsymbol{\kappa}_n^Z) \xrightarrow{p^2} H^2_{\text{\'et}}(X, \boldsymbol{\mu}_2) \to \cdots$$

of étale cohomology groups.

- a) Then $p^2gc(\mathscr{E},q,\mathscr{L}) = c_1(\mathscr{L},\mu_2)$, i.e. the similarity Clifford invariant maps to the 1st Chern class (modulo 2) of \mathscr{L} .
- b) If $\mathscr{L} = \mathscr{O}_X$ then $gc(\mathscr{E}, q, \mathscr{O}_X) = i^2 c(\mathscr{E}, q)$, i.e. the classical Clifford invariant of a quadratic form maps to the similarity Clifford invariant of its similarity class.

Proof. In light of the cohomology interpretation of Remark 1.5, one needs only to consider the implications on étale cohomology of the fundamental diagram (15). For b, it remains to be verified that the coboundary map $H^1_{\text{ét}}(X, \mathbf{SO}_n^Z) \to H^2_{\text{\acute{e}t}}(X, \boldsymbol{\mu}_2)$ agrees with the Clifford invariant of a quadratic form (\mathscr{E}, q) (together with an irrelevant choice of orientation) with Arf invariant [Z/X]. This is verified using a generalization to $X_{\text{\acute{e}t}}$ of formulas from Lam [73, V Prop. 3.20] and Serre [90, III App. 2, §2.2].

2.6. Functoriality. Let $f: Y \to X$ be a morphism of schemes. If $(\mathscr{E}, q, \mathscr{L})$ is an \mathscr{L} -valued quadratic form of rank n on X, then there's a canonical $f^*\mathscr{L}$ -valued quadratic form $f^*(\mathscr{E}, q, \mathscr{L})$ of rank n on Y. By the functoriality of the Clifford group, there's also a canonical isomorphism of group schemes $\kappa_n^{Y \times XZ} \simeq f^{-1}\kappa_n^Z$, where f^{-1} is the pullback in the category of sheaves of groups on $X_{\text{\acute{e}t}}$ (see Milne [76, II §2]). Finally, we have

$$gc(f^*q) = f^*gc(q)$$

in
$$H^2_{\text{\'et}}(Y, \boldsymbol{\kappa}^m_{Y \times_X Z/Y}) \simeq H^2_{\text{\'et}}(Y, f^{-1} \boldsymbol{\kappa}^Z_n)$$

2.7. Grothendieck–Witt equivalence. Now let $f : Z \to X$ be split, so that $h_n^Z = H_{\mathscr{O}_X}(\mathscr{O}_X^m)$. Write $\mathbf{G}_{m,m}$ instead of \mathbf{G}_n^Z for any sheaf of groups depending on a quadratic form. If n is divisible by four then $gc(q) \in H^2_{\mathrm{\acute{e}t}}(X, \boldsymbol{\mu}_4)$ via the identification $\boldsymbol{\mu}_4 \simeq \boldsymbol{\kappa}$ of Remark 2.6.

Theorem 2.11. Let X be a scheme with 2 invertible. Let $(\mathscr{E}, q, \mathscr{L})$ be a regular quadratic form of rank n = 2m divisible by four and trivial Arf invariant on X.

a) If \mathscr{V} is a locally free \mathscr{O}_X -module of even rank, then

$$gc(q \perp H_{\mathscr{L}}(\mathscr{V})) = gc(q) + i^2 c_1(\mathscr{V}, \boldsymbol{\mu}_2).$$

b) If $(\mathscr{E}, q, \mathscr{L})$ is metabolic with lagrangian $\mathscr{V} \to \mathscr{E}$, then

$$gc(q) = c(H_{\mathscr{L}}(\mathscr{V})) = c_1(\mathscr{L}, \boldsymbol{\mu}_4) + i^2 c_1(\mathscr{V}, \boldsymbol{\mu}_2).$$

Proof. Consider the homomorphism $H: \mathbf{G}_m \times \mathbf{GL}_m \to \mathbf{GSO}_{m,m}$ defined by

$$H(l,\varphi) = \begin{pmatrix} \varphi & 0\\ 0 & l \, (\varphi^{-1})^{\vee} \end{pmatrix}.$$

Lemma 2.12. The induced mapping $H^1 : H^1_{\text{ét}}(X, \mathbf{G}_m \times \mathbf{GL}_m) \to H^1_{\text{\acute{e}t}}(X, \mathbf{GSO}_{m,m})$ is interpreted as $(\mathscr{L}, \mathscr{V}) \mapsto (H_{\mathscr{L}}(\mathscr{V}), \zeta_{\mathscr{V}})$, where $\zeta_{\mathscr{V}}$ is the orientation from Lemma 1.14.

Proof. Let $\mathscr{U} = \{U_i\}_{i \in I}$ be a Zariski open cover of X splitting \mathscr{L} and \mathscr{V} via \mathscr{O}_{U_i} -module isomorphisms

 $\lambda_i: \mathscr{O}_{U_i} \simeq \mathscr{L}|_{U_i}, \quad \phi_i: \mathscr{O}_{U_i}^m \simeq \mathscr{V}|_{U_i},$

for each $i \in I$. The \mathcal{O}_{U_i} -module morphism

$$\begin{array}{rcl} \psi_i: H_{\mathscr{O}_{U_i}}(\mathscr{O}_{U_i}) & \to & H_{\mathscr{L}}(\mathscr{V})|_{U_i} \\ (v, f) & \mapsto & (\phi_i(v), \lambda_i \circ f \circ \phi_i^{-1}) \end{array}$$

is a proper similarity transformation with similarity factor λ_i . For each $(i, j) \in I^2$, the collections

$$l_{ij} = \lambda_{\underline{ij}}^{-1} \circ \lambda_{\underline{ij}} \in \mathbf{G}_{\mathbf{m}}(U_{ij}), \qquad \varphi_{ij} = \phi_{\underline{ij}}^{-1} \circ \phi_{\underline{ij}} \in \mathbf{GL}_{m}(U_{ij})$$

and

$$b_{ij} = \psi_{ij}^{-1} \circ \psi_{\underline{i}j} \in \mathbf{GO}_{m,m}(U_{ij}),$$

form Čech étale 1-cocycles representing the classes associated to \mathscr{L} , \mathscr{V} , and $H_{\mathscr{L}}(\mathscr{V})$ in $H^1_{\text{\acute{e}t}}(\mathscr{U}, \mathbf{G}_m)$, $H^1_{\text{\acute{e}t}}(\mathscr{U}, \mathbf{GL}_m)$, and $H^1_{\text{\acute{e}t}}(\mathscr{U}, \mathbf{GO}_{m,m})$, respectively. With respect to the decomposition $H_{\mathscr{O}_X}(\mathscr{O}_X^m) = \mathscr{O}_X^m \oplus \mathscr{O}_X^{m\vee}$, the cocycle b_{ij} is represented by the matrices

$$H(l_{ij},\varphi_{ij}) = \begin{pmatrix} \varphi_{ij} & 0\\ 0 & l_{ij} (\varphi_{ij}^{-1})^{\vee} \end{pmatrix}$$

proving the claim. Thus we've shown that $H^1_{\acute{e}t}(X, \mathbf{G}_m \times \mathbf{GL}_m) \to H^1_{\acute{e}t}(X, \mathbf{GO}_{m,m})$ is interpreted by $(\mathscr{L}, \mathscr{V}) \mapsto H_{\mathscr{L}}(\mathscr{V})$. All that remains is the verification of the orientation statement. If 2 is invertible on X, this is equivalent to the equality

$$(\det/\mu^m)(H(l_{ij},\varphi_{ij})) \circ \zeta_{\mathscr{V}|_{U_{ij}}} = \zeta_{\mathscr{V}|_{U_{ij}}} : \mathscr{O}_{U_{ij}} = d_{\pm}(H_{\mathscr{O}_{U_{ij}}}(\mathscr{O}_{U_{ij}}^m)) \to d_{\pm}(H_{\mathscr{O}_{U_{ij}}}(\mathscr{V}|_{U_{ij}})).$$

This is a straightforward process of unwinding the definition of $\zeta_{\mathscr{V}}$ from Lemma 1.14.

Furthermore, consider the restriction of H to the maximal torus $\mathbf{G}_{\mathrm{m}} \times \mathbf{G}_{\mathrm{m}}^{\times m} \to \mathbf{GSO}_{m,m}$, which similarly to Lemma 2.12, is interpreted as $(\mathscr{L}, \mathscr{V}_1, \ldots, \mathscr{V}_m) \mapsto H_{\mathscr{L}}(\mathscr{V}_1 \oplus \cdots \oplus \mathscr{V}_m)$, where the orientation from Lemma 2.12 is understood. We now construct a commutative diagram with exact rows

lifting H to the even Clifford group: the top row is the product of appropriate Kummer sequences, the bottom row is the even Clifford sequence (14) for the split similitude group, Π is the homomorphism given by multiplying a tuple of the elements together, and

$$\tilde{H}(l, a_1, \dots, a_m) = l^{m+1} \prod_{j=1}^m a_j^{-1} \Big(1 - \big(1 - (l^{-1}a_j)^2 \big) e_j f_j \Big)$$

on sections, where $e_1, \ldots, e_m, f_1, \ldots, f_m$ are global sections of $H_{\mathscr{O}_X}(\mathscr{O}_X^m)$ forming a hyperbolic basis, i.e. $h(e_i, e_j) = 0$, $h(f_i, f_j) = 0$, and $h(e_i, f_j) = \delta_{ij}$, where δ_{ij} is the Kronecker δ -function.

To verify the commutativity of diagram (16), we will decompose \tilde{H} into a product of homomorphisms. Calculations in the Clifford algebra show that for each $1 \leq j \leq m$, the map $\chi_j : \mathbf{G}_m \to \mathbf{S}\Gamma_{m,m}$ defined on sections by

$$\chi_{j}(x) = 1 - (1 - x)e_{j}f_{j}$$

forms a one parameter subgroup of $\mathbf{S}\Gamma_{m,m}$ satisfying

$$N(\chi_j(x)) = x, \qquad r(\chi_j(x)): \begin{array}{ccc} e_k & \mapsto & \left(1 - \delta_{jk}(1-x)\right)e_k \\ f_k & \mapsto & \left(1 - \delta_{jk}(1-x^{-1})\right)f_k \end{array}$$

In particular,

$$N(\tilde{H}(l, a_1, \dots, a_m)) = l^2, \qquad r(\tilde{H}(l, a_1, \dots, a_m)) = l^{-2}H(l^4, a_1^2, \dots, a_m^2)$$

so that \hat{H} is a homomorphism and the rightmost square of diagram (16) is commutative. The fact that the leftmost square is commutative results from the identity

$$\tilde{H}(\sqrt{-1}, 1, \dots, 1) = (-1)^{m/2}\sqrt{-1} \prod_{j=1}^{m} (1 - e_j f_j)$$

and the identification $\mu_4 \simeq \kappa(h)$ of Remark 2.6.

Finally, by the splitting principle for metabolic forms (see Theorem 1.9) and the fact that the formulas of Theorem 2.11 all commute with pullbacks, to prove b) we can reduce to the case where $(\mathscr{E}, q, \mathscr{L})$ is a hyperbolic form $H_{\mathscr{L}}(\mathscr{V})$ with lagrangian $\mathscr{V} = \mathscr{V}_1 \oplus \cdots \oplus \mathscr{V}_m$ a sum of invertible modules. The coboundary map of the top sequence of diagram (16), composed with the map Π^2 induced from the leftmost vertical map, yields

$$(\mathscr{L},\mathscr{V}_1,\ldots,\mathscr{V}_m)\mapsto (c_1(\mathscr{L},\boldsymbol{\mu}_4),c_1(\mathscr{V}_1,\boldsymbol{\mu}_2),\ldots,c_1(\mathscr{V}_m,\boldsymbol{\mu}_2))\mapsto c_1(\mathscr{L},\boldsymbol{\mu}_4)+i^2c_1(\mathscr{V},\boldsymbol{\mu}_2)$$

and thus $H_{\mathscr{L}}(\mathscr{V}) \mapsto c_1(\mathscr{L}, \mu_4) + i^2 c_1(\mathscr{V}, \mu_2)$ under the coboundary map of the Clifford sequence for the similitude group, verifying the formula in b).

As for a), let m' be even and consider the commutative diagram with exact rows

(17)
$$1 \xrightarrow{} \mu_{4} \times \mu_{2}^{\times m'} \xrightarrow{} \mathbf{S}\Gamma_{m,m} \times \mathbf{G}_{m}^{\times m'} \xrightarrow{} \mathbf{GSO}_{m,m} \times \mathbf{G}_{m}^{\times m'} \xrightarrow{} 1$$
$$\prod_{i} \xrightarrow{\tilde{H}'_{i}} \xrightarrow{H'_{i}} H'_{i} \xrightarrow{H'_{i}} 1$$
$$1 \xrightarrow{} \mu_{4} \xrightarrow{} \mathbf{S}\Gamma_{m+m',m+m'} \xrightarrow{} \mathbf{GSO}_{m+m',m+m'} \xrightarrow{} 1$$

where $H'((\varphi, l), a_1, \dots, a_{m'}) = (\varphi, l) \perp H(l, a_1, \dots, a_{m'})$ and

$$\tilde{H}'(x, a_1, \dots, a_{m'}) = N(x)^{m'/2} x \prod_{j=1}^{m'} a_j^{-1} \chi_j(N(x)^{-1} a_j^2)$$

The commutativity of the diagram is checked as above. Finally, H'^1 is interpreted as

$$((\mathscr{E},q,\mathscr{L}),\mathscr{V}_1,\ldots,\mathscr{V}_{m'})\mapsto (\mathscr{E},q,\mathscr{L})\perp H_{\mathscr{L}}(\mathscr{V}_1\oplus\cdots\oplus\mathscr{V}_{m'}),$$

while the coboundary map of the top sequence is interpreted as

$$((\mathscr{E}, q, \mathscr{L}), \mathscr{V}_1, \dots, \mathscr{V}_{m'}) \mapsto (c(\mathscr{E}, q, \mathscr{L}), c_1(\mathscr{V}_1, \boldsymbol{\mu}_2), \dots, c_1(\mathscr{V}_{m'}, \boldsymbol{\mu}_2))$$

so that in total $gc(q \perp H_{\mathscr{L}}(\mathscr{V})) = c(q) + i^2 c_1(\mathscr{V}, \mu_2)$, verifying the formula in a).

Corollary 2.13. Let $(\mathscr{E}_0, q_0, \mathscr{L})$, $(\mathscr{E}_1, q_1, \mathscr{L})$, and $(\mathscr{E}_2, q_2, \mathscr{L})$ be \mathscr{L} -valued quadratic forms of rank divisible by 4 and trivial Arf invariant. If there's an isometry $q_0 \perp q_1 \cong q_0 \perp q_2$ then $gc(q_1) = gc(q_2)$.

Proof. If $q_0 \perp q_1 \cong q_0 \perp q_2$ then $(-q_0 \perp q_0) \perp q_1 \cong (-q_0 \perp q_0) \perp q_2$. But $(\mathscr{E}_0, -q_0, \mathscr{L}) \perp (\mathscr{E}_0, q_0, \mathscr{L})$ is \mathscr{L} -valued metabolic with lagrangian \mathscr{E}_0 , so that by Theorem 2.11,

$$gc(q_1) + i^2c_1(\mathscr{E}_0, \mu_2) = gc(q_2) + i^2c_1(\mathscr{E}_0, \mu_2)$$

in $H^2_{\text{ét}}(X, \mu_4)$, proving the corollary.

Let $GW(X, \mathscr{L})$ be the Grothendieck–Witt group of \mathscr{L} -valued quadratic forms on X and $GW^{\text{tot}}(X) = \bigoplus_{\mathscr{L}} GW(X, \mathscr{L})$ the total quadratic Grothendieck-Witt group. Letting $GI_1^{\text{tot}}(X) = \bigoplus_{\mathscr{L}} GI_1(X, \mathscr{L})$ be the subgroup of quadratic forms of even rank, the signed discriminant defines a (surjective) cohomological invariant (in a sense generalizing Garibaldi–Merkurjev–Serre [43])

$$d_{\pm}: GI_1^{\text{tot}}(X) \to H^1_{\text{\acute{e}t}}(X, \boldsymbol{\mu}_2).$$

Letting $GI_2^{\text{tot}}(X) = \bigoplus_{\mathscr{L}} GI_2(X, \mathscr{L})$ be the subgroup of quadratic forms of rank $n \equiv 0 \mod 4$ and trivial discriminant, we may interpret Theorem 2.11 as the statement that the similarity Clifford invariant descends to a cohomological invariant

$$gc: GI^{\text{tot}}(X) \to H^2_{\text{\'et}}(X, \mu_4).$$

2.8. Oriented invariants. We now restrict our attention to line bundle-valued quadratic forms with trivial Arf invariant. As noted in Definition 2.8, the invariant $gc(\mathscr{E}, q, \mathscr{L}) \in H^2_{\text{ét}}(X, \kappa_{m,m})$ does not depend on an orientation of $(\mathscr{E}, q, \mathscr{L})$. In this section, we construct related invariants in $H^2_{\text{ét}}(X, \mu_2)$, but which are orientation dependent.

Definition 2.14. Let (\mathcal{H}, b) be a regular quadratic form of even rank n = 2m on X. Denote by $\mathbf{z} = \mathbf{z}(\mathcal{H}, b)$ be the subgroup scheme of $\mathbf{S\Gamma}(\mathcal{H}, b)$ defined by

$$\mathbf{z}(U) = \{ x \in \mathbf{S\Gamma}(\mathscr{H}, b)(U) : N(x) = 1, \ r(x) \in \boldsymbol{\mu}_2(U) \}$$

Equivalently, \mathbf{z} is the kernel in the short exact sequence of sheaves of groups

(18)
$$1 \rightarrow \mathbf{z} \rightarrow \mathbf{Spin} \rightarrow \mathbf{PSO} \rightarrow 1$$

on $X_{\text{ét}}$ (here $\mathbf{PSO} = \mathbf{SO}/\mu_2$, see §3.1). In fact, \mathbf{z} is the center of the simply connected affine algebraic group **Spin**. It's a classical fact, see [69, VI.26.A] for example, that there are isomorphisms of group schemes

$$\mathbf{z} \cong \begin{cases} \mathbf{R}_{Z/X} \boldsymbol{\mu}_2 & \text{if } n \equiv 0 \mod 4 \\ \mathbf{R}_{Z/X}^1 \boldsymbol{\mu}_4 & \text{if } n \equiv 2 \mod 4 \end{cases}$$

where $Z \to X$ is the Arf cover of (\mathcal{H}, b) .

We retain the notation of §2.7 and assume, for simplicity of exposition, that X is connected. For each $m \geq 1$, the choice of global section $\varepsilon = \prod_{i=1}^{m} (1 - 2e_i f_i)$ of $\mathbf{S}\Gamma_{m,m}$ fixes a group scheme isomorphism $\mu_2 \times \mu_2 \simeq \mathbf{z}_{m,m}$ for m even and $\mu_2 \times \mu_2 \simeq \kappa_{m,m}$ for m odd. Indeed, if m is even, then ε is a global section of $\mathbf{Spin}_{m,m}$, which together with -1, generate $\mathbf{z}_{m,m}$. If m is odd, see Example 2.6.

For *m* even, define $\Gamma'_{m,m} = \mathbf{G}_m \times \mathbf{Spin}_{m,m}$ and $\gamma^+_{m,m}$ to be the subgroupscheme generated by the global section $(-1, \varepsilon)$. For *m* odd, define $\Gamma'_{m,m} = \mathbf{S}\Gamma_{m,m}$ and $\gamma^+_{m,m}$ to be the subgroupscheme generated by the global section ε .

Definition 2.15. For $m \geq 1$, define $\operatorname{\mathbf{GSpin}}_{m,m}^+ = \Gamma'_{m,m}/\gamma_{m,m}^+$. Replacing ε by $-\varepsilon$ yields an analogous subgroup $\gamma_{m,m}^-$ and quotient $\operatorname{\mathbf{GSpin}}_{m,m}^-$.

Remark 2.16. For m odd, $\operatorname{\mathbf{GSpin}}_{m,m}^{\pm}$ is precisely the image of $\operatorname{\mathbf{S\Gamma}}_{m,m}$ under the "half-spin" representations $\rho^{\pm} : \operatorname{\mathbf{S\Gamma}}_{m,m} \to \operatorname{\mathbf{GL}}_{2^{2(m-1)}}$. In the literature, $\operatorname{\mathbf{GSpin}}(q)$ is often another name for the even Clifford group $\operatorname{\mathbf{S\Gamma}}(q)$ of a quadratic form. Our notation is somewhat justified since $\operatorname{\mathbf{GSpin}}_{m,m}^+$ becomes isomorphic to $\operatorname{\mathbf{S\Gamma}}_{m,m}$ upon replacing ε by $(-1)^{m-1}$ in the definition.

The compositions $\mathbf{G}_{\mathrm{m}} \times \mathbf{Spin}_{m,m} \xrightarrow{pr_1} \mathbf{G}_{\mathrm{m}} \xrightarrow{2} \mathbf{G}_{\mathrm{m}}$ and $\mathbf{S\Gamma}_{m,m} \xrightarrow{N} \mathbf{G}_{\mathrm{m}} \xrightarrow{2} \mathbf{G}_{\mathrm{m}}$ descend to homomorphisms $\mu^{\pm} : \mathbf{GSpin}_{m,m}^{\pm} \to \mathbf{G}_{\mathrm{m}}$. The kernel of μ^{\pm} is isomorphic to $\mathbf{Spin}_{m,m}$, as seen by the Nine Lemma 5.16 applied to the commutative diagram



of sheaves of groups on $X_{\text{ét}}$, with exact columns and bottom rows.

Also, the homomorphisms $\mathbf{G}_m \times \mathbf{Spin}_{m,m} \to \mathbf{GSO}_{m,m}$ given by $(a, x) \mapsto (ar(x), a^2)$ and $s : \mathbf{S\Gamma}_{m,m} \to \mathbf{GSO}_{m,m}$ from §2.4) factor through homomorphisms $s^{\pm} : \mathbf{GSpin}_{m,m}^{\pm} \to \mathbf{GSO}_{m,m}$, yielding short exact sequences

(19)
$$1 \to \boldsymbol{\mu}_2 \to \mathbf{GSpin}_{m,m}^{\pm} \xrightarrow{s^{\pm}} \mathbf{GSO}_{m,m} \to 1$$

fitting into commutative diagrams with exact rows and columns

of sheaves of groups on $X_{\text{\acute{e}t}}$.

Definition 2.17. For an oriented regular line bundle-valued quadratic form $(\mathscr{E}, q, \mathscr{L}, \zeta)$ of rank n = 2m and trivial Arf invariant, define a pair of invariants $gc^{\pm}(q, \zeta) = gc^{\pm}(\mathscr{E}, q, \mathscr{L}, \zeta) \in H^2_{\text{ét}}(X, \mu_2)$ to be the images of $(\mathscr{E}, q, \mathscr{L}, \zeta)$ under the coboundary maps $H^1_{\text{ét}}(X, \mathbf{GSpin}_{m,m}^{\pm}) \to H^2_{\text{ét}}(X, \mu_2)$ arising from sequence (19).

Proposition 2.18. Let $(\mathscr{E}, q, \mathscr{L})$ be a regular line bundle-valued quadratic form of rank n = 2m and trivial Arf invariant. Let ζ be any orientation of $(\mathscr{E}, q, \mathscr{L})$. Then we have:

- a) $gc^{\pm}(q,\zeta) = c(q)$ if $\mathscr{L} = \mathscr{O}_X$,
- b) $gc^+(q,\zeta) + gc^-(q,\zeta) = c_1(\mathscr{L}, \mu_2),$
- c) $gc^{\pm}(q,-\zeta) = gc^{\mp}(q,\zeta),$
- d) $gc^+(q,\zeta_{\mathscr{V}}) = c_1(\mathscr{V},\boldsymbol{\mu}_2), \text{ if } \mathscr{V} \to \mathscr{E} \text{ is a lagrangian of } (\mathscr{E},q,\mathscr{L}) \text{ and } \zeta_{\mathscr{V}} \text{ is the induced orientation from Lemma 1.14.}$

Proof. For a, we simply consider the interpretation on cohomology of the top rows of diagram (20).

For b), denote by $\gamma_{m,m}$ the subgroupscheme of $\Gamma'_{m,m}$ generated by $\gamma^+_{m,m}$ and $\gamma^-_{m,m}$. Note that when m is odd, $\gamma_{m,m} = \kappa_{m,m}$. We have a fixed isomorphism $\mu_2 \times \mu_2 \simeq \gamma_{m,m}$ and a labeling $\gamma^{\pm}_{m,m}$ and $\gamma^0_{m,m}$ of the three subgroupschemes of $\gamma_{m,m}$. For $\bullet \in \{-,0,+\}$, denote by p^{\bullet} the quotient homomorphism in the exact sequence

$$1 \to \boldsymbol{\gamma}^{ullet}_{m,m} \to \boldsymbol{\gamma}_{m,m} \xrightarrow{p^{ullet}} \boldsymbol{\mu}_2 \to 1.$$

For each $m \ge 1$ and each $\bullet \in \{-, 0, +\}$, it's a straightforward verification that the diagram



of sheaves of groups on $X_{\acute{e}t}$ is commutative with exact rows and columns. Here we've used the following notation: for $\bullet \in \{\pm\}$, the map $q^{\bullet} : \Gamma'_{m,m} \to \mathbf{GSpin}^{\bullet}_{m,m}$ is the quotient homomorphism; for $\bullet = 0$, we set $s^0 = s$ as defined in Definition 2.2, $\mathbf{GSpin}^0_{m,m} = \mathbf{G}_m \times \mathbf{SO}_{m,m}$, and $q^0 = N \times r : \Gamma'_{m,m} = \mathbf{S}\Gamma_{m,m} \to \mathbf{G}_m \times \mathbf{SO}_{m,m}$ when m is odd while $q^0 = \mathrm{id} \times r : \Gamma'_{m,m} = \mathbf{G}_m \times \mathbf{Spin}_{m,m} \to \mathbf{G}_m \times \mathbf{SO}_{m,m}$ when m is even.

For each $m \ge 1$, denote by $\delta_m : H^1_{\text{ét}}(X, \mathbf{GSO}_{m,m}) \to H^2_{\text{ét}}(X, \gamma_{m,m})$ the coboundary map associated to the central row of diagram (21). Note that for m odd, $\delta_m = gc$, by Definition 2.8.

Consider the split exact sequence of group schemes,

$$1 \to \boldsymbol{\gamma}_{m,m} \xrightarrow{p^- \times p^0 \times p^+} \boldsymbol{\mu}_2^{\times 3} \xrightarrow{\Pi} \boldsymbol{\mu}_2 \to 1,$$

uniquely defined up to ordering, where Π is the total multiplication homomorphism. On étale cohomology, there are thus (split) exact sequences,

(22)
$$0 \to H^i_{\text{ét}}(X, \gamma_{m,m}) \xrightarrow{p^{-,i} \oplus p^{0,i} \oplus p^{+,i}} H^i_{\text{ét}}(X, \mu_2)^{\oplus 3} \to H^i_{\text{ét}}(X, \mu_2) \to 0$$

of abelian groups for each $i \ge 0$.

By the commutativity of the bottom rows of diagram (21), we have

$$p^{\bullet,2}\delta_m(\mathscr{E},q,\mathscr{L},\zeta) = \begin{cases} gc^{\bullet}(\mathscr{E},q,\mathscr{L},\zeta) & \text{for } \bullet \in \{\pm\} \\ c_1(\mathscr{L},\boldsymbol{\mu}_2) & \text{for } \bullet = 0 \end{cases}$$

by the definition of gc^{\pm} for $\bullet \in \{\pm\}$ and by the commutativity of diagram (6) for $\bullet = 0$. Note that for *m* odd and $\bullet = 0$, this is a restatement of Theorem 2.10*a*). In particular, by exact sequence (22) of cohomology groups (for i = 2),

$$gc^{+}(\mathscr{E},q,\mathscr{L}) + gc^{-}(\mathscr{E},q,\mathscr{L}) = p^{-,2}\delta_{m}(\mathscr{E},b,\mathscr{L},\zeta) + p^{+,2}\delta_{m}(\mathscr{E},b,\mathscr{L},\zeta) = c_{1}(\mathscr{L},\boldsymbol{\mu}_{2})$$

in $H^2_{\text{\'et}}(X, \boldsymbol{\mu}_2)$.

Part c) follows immediately from the definition of $\mathbf{GSpin}_{m,m}^{\pm}$. For d), we can adapt the proof of Theorem 2.11

Remark 2.19. By Proposition 2.18c), if \mathscr{L} is not a square in $\operatorname{Pic}(X)$, then $gw^{\pm}(q,\zeta)$ depends on the orientation ζ . In particular, if $\operatorname{Pic}(X)$ is not 2-divisible, then the invariants gc^{\pm} are in general orientation dependent and hence do not descend to $GI_2^{\operatorname{tot}}(X)$.

Example 2.20. If m = 3, then $\mathbf{GSpin}_{3,3}^{\pm} \cong \mathbf{GL}_4$ and exact sequence (19) is isomorphic to the familiar sequence

$$1 \rightarrow \mu_2 \rightarrow \mathbf{GL}_4 \xrightarrow{\wedge^2} \mathbf{GSO}_{3,3} \rightarrow 1_3$$

for an appropriate choice of \pm and ϵ , see §5.4 for more details.

As an application of Example 2.20, we can give a characterization of locally free \mathscr{O}_X -modules of rank 6 that are *exterior squares*, i.e. isomorphic to $\bigwedge^2 \mathscr{V}$ for a locally free \mathscr{O}_X -module \mathscr{V} of rank 4.

Proposition 2.21. Let X be a scheme with 2 invertible. Then a locally free \mathcal{O}_X -module \mathscr{E} of rank 6 is an exterior square if and only if \mathscr{E} supports a regular line bundle-valued quadratic form $(\mathscr{E}, q, \mathscr{L})$ with trivial Arf invariant and trivial $gc^+(\mathscr{E}, q, \mathscr{L}, \zeta)$ for some orientation ζ .

Proof. If $\mathscr{E} \cong \bigwedge^2 \mathscr{V}$, then \mathscr{E} supports a regular line bundle-valued quadratic form via the canonical "wedging" form $\bigwedge^2 \mathscr{V} \xrightarrow{\wedge} \det \mathscr{V}$). That such a form has trivial Arf invariant follows from its interpretation as a reduced pfaffian and the explicit calculation of the even Clifford algebra of a reduced pfaffian, see §5.4 and [17, Prop. 4.8]. Since 2 is invertible on X we can argue directly, by showing that the associated symmetric bilinear form has trivial discriminant. Indeed, when \mathscr{V} is free with basis e_1, \ldots, e_4 , an \mathscr{O}_X -module morphism $\zeta : \det(\bigwedge^2 \mathscr{V}) \to (\det \mathscr{V})^{\otimes 3}$ can be given by

$$\bigwedge_{1 \le i < j \le 4} (e_i \land e_j) \quad \mapsto \quad (e_1 \land \dots \land e_4)^{\otimes 3}.$$

A standard computation (compute the similarity factor on the standard middle exterior power of the fundamental representation of \mathbf{GL}_r) shows that this morphism does not depend on the choice of basis. Hence for a general locally free \mathscr{V} , this morphism patches over a Zariski open cover of X splitting \mathscr{V} . Finally, scaling ζ by $(\det \mathscr{V}^{\vee})^{\otimes 3}$ yields an \mathscr{O}_X -module morphism $\zeta_{\mathscr{V}} : d_{\pm}(\bigwedge^2 \mathscr{V}, \wedge, \det \mathscr{V}) \to <-1>$, which can be checked to be an isometry.

Now the map $\mathbf{GL}_4 \xrightarrow{\wedge^2} \mathbf{GSO}_{3,3}$ can be reinterpreted as the homomorphism of automorphism sheaves (when applied to $\mathscr{V} = \mathscr{O}_X^4$) induced from the functor $\mathscr{V} \mapsto (\bigwedge^2 \mathscr{V}, \land, \det \mathscr{V})$. Hence, the exact sequence of cohomology sets

$$H^{1}_{\text{\acute{e}t}}(X, \mathbf{GL}_{4}) \xrightarrow{(\wedge^{2})^{1}} H^{1}_{\text{\acute{e}t}}(X, \mathbf{GSO}_{3,3}) \xrightarrow{gc^{+}} H^{2}_{\text{\acute{e}t}}(X, \boldsymbol{\mu}_{2})$$

has the following interpretation: $(\wedge^2)^1$ takes the isomorphism class of a locally free \mathscr{O}_X -module \mathscr{V} of rank 4 to $(\bigwedge^2 \mathscr{V}, \wedge, \det \mathscr{V})$ with the above orientation $\zeta_{\mathscr{V}}$. In particular, $gc^+(\bigwedge^2 \mathscr{V}, \wedge, \det \mathscr{V}, \zeta_{\mathscr{V}})$ is trivial.

Now suppose that \mathscr{E} supports a regular line bundle-valued quadratic form $(\mathscr{E}, q, \mathscr{L})$ with trivial Arf invariant. Then the similarity class of $(\mathscr{E}, q, \mathscr{L})$ defines a class in $H^1_{\text{\acute{e}t}}(X, \mathbf{GSO}_{3,3})$ and by the

ASHER AUEL

exactness of the above sequence, $gc^+(\mathscr{E}, q, \mathscr{L}, \zeta)$ is trivial for some ζ if and only if $(\mathscr{E}, q, \mathscr{L}, \zeta)$ is in the image of $(\wedge^2)^1$. In particular, $\mathscr{E} \cong \bigwedge^2 \mathscr{V}$ for some locally free \mathscr{O}_X -module \mathscr{V} of rank 4. \Box

3. Relationship to the even Clifford Algebra

In this section we relate the similarity Clifford invariant to classes in étale cohomology arising from the (generalized) even Clifford algebra construction of Bichsel–Knus [17].

3.1. Involutions on endomorphism algebras and projective similarity. An antiautomorphism of an \mathscr{O}_X -algebra \mathscr{A} is defined to be a \mathscr{O}_X -algebra isomorphism $\sigma : \mathscr{A} \to \mathscr{A}^{\mathrm{op}}$, where $\mathscr{A}^{\mathrm{op}}$ is the opposite algebra. An antiautomorphism σ is called an *involution* (of the first kind) if $\sigma^{\mathrm{op}} \circ \sigma = \mathrm{id}_{\mathscr{A}}$, identifying $\mathscr{A} = (\mathscr{A}^{\mathrm{op}})^{\mathrm{op}}$. A morphism $\varphi : (\mathscr{A}, \sigma) \to (\mathscr{A}', \sigma')$ between \mathscr{O}_X -algebras with involution consists of an \mathscr{O}_X -algebra homomorphism $\varphi : \mathscr{A} \to \mathscr{A}'$ such that $\varphi \circ \sigma' = \sigma \circ \varphi^{\mathrm{op}}$. If $(\mathscr{E}, b, \mathscr{L})$ is a regular \mathscr{L} -valued bilinear form, the \mathscr{O}_X -algebra anti-automorphism defined by

$$\sigma_b: \mathscr{E}nd(\mathscr{E}) \to \mathscr{E}nd(\mathscr{E})^{\operatorname{op}}$$
$$\varphi \mapsto \psi_{\scriptscriptstyle h}^{-1} \circ \varphi^{\vee \mathscr{L}} \circ \psi_b$$

is an involution (called the *adjoint involution*) if and only if b is ϵ -symmetric (see 1.3), for some $\epsilon \in H^0_{\acute{e}t}(X, \mu_2)$. Conversely, any involution (of the first kind) on $\mathscr{E}nd(\mathscr{E})$ arises this way.

Proposition 3.1 (Saltman [88, Thm. 4.2a], Knus-Parimala–Srinivas [67]). Let X be a scheme and \mathscr{E} a locally free \mathscr{O}_X -module of finite rank. If σ is an involution (of the first kind) of $\mathscr{E}nd(\mathscr{E})$ then there exists a regular ϵ -symmetric bilinear form $(\mathscr{E}, b, \mathscr{L})$ such that $\sigma = \sigma_b$. Moreover, the similarity class of $(\mathscr{E}, b, \mathscr{L})$ is uniquely determined by \mathscr{E} and σ .

Remark 3.2. Note however, that the \mathscr{O}_X -algebras $\mathscr{E}nd(\mathscr{N} \otimes \mathscr{E})$ and $\mathscr{E}nd(\mathscr{E})$ are isomorphic for any invertible \mathscr{O}_X -module \mathscr{N} (see Remark 1.3). Thus an involution σ on $\mathscr{E}nd(\mathscr{E})$ that corresponds to a bilinear form $(\mathscr{E}, b, \mathscr{L})$ will give rise be an involution on $\mathscr{E}nd(\mathscr{N} \otimes \mathscr{E})$ that corresponds to the bilinear form $(\mathscr{N} \otimes \mathscr{E}, b', \mathscr{N}^{\otimes 2} \otimes \mathscr{L})$. Thus the isomorphism class (as \mathscr{O}_X -algebras with involution) of $(\mathscr{E}nd(\mathscr{E}), \sigma_b)$ only determines the *projective similarity* class of the bilinear form $(\mathscr{E}, b, \mathscr{L})$, i.e. the set of similarity classes of bilinear forms $(\mathscr{N} \otimes \mathscr{E}, n \otimes b, \mathscr{N}^{\otimes 2} \otimes \mathscr{L})$, where $(\mathscr{N}, n, \mathscr{N}^{\otimes 2})$ is a bilinear form of rank 1.

If $(\mathscr{E}, b, \mathscr{L})$ is a regular ϵ -symmetric bilinear form, the *projective similitude* group $\mathbf{PSim}(\mathscr{E}, b, \mathscr{L})$ is defined as the sheaf of automorphism (as algebras with involution) groups $\mathbf{Aut}(\mathscr{E}nd(\mathscr{E}), \sigma_b)$. There's a natural homomorphism $\mathbf{GO}(\mathscr{E}, q, \mathscr{L}) \to \mathbf{PSim}(\mathscr{E}, q, \mathscr{L})$ of sheaves of groups on $X_{\acute{e}t}$, with kernel the central embedding $\mathbf{G}_{m} \to \mathbf{Sim}(\mathscr{E}, b, \mathscr{L})$ given by homotheties There is an exact sequence

(23)
$$1 \to \mathbf{G}_{\mathrm{m}} \to \mathbf{Sim}(\mathscr{E}, b, \mathscr{L}) \to \mathbf{PSim}(\mathscr{E}, b, \mathscr{L}) \to 1.$$

of sheaves of groups on $X_{\text{\acute{e}t}}$. The embedding of homotheties restricts to a central embedding $\mu_2 \rightarrow \text{Isom}(\mathscr{E}, b, \mathscr{L})$, the sheaf cokernel of which defines the *projective orthogonal* group $\text{PIsom}(\mathscr{E}, b, \mathscr{L})$. In total, we have a commutative diagram with exact columns and rows

J

of sheaves of groups on $X_{\text{\acute{e}t}}$ by the Nine Lemma 5.16.

A locally free \mathcal{O}_X -algebra \mathscr{A} of finite rank is an Azumaya \mathcal{O}_X -algebra if the canonical \mathcal{O}_X -algebra homomorphism

$$\overset{\mathscr{A}}{a} \otimes \overset{\mathscr{A}^{\mathrm{op}}}{b} \to \mathscr{E}\!\mathit{nd}(\mathscr{A}) \ a \otimes b \mapsto c \mapsto acb$$

is an isomorphism, where $\mathscr{E}nd(\mathscr{A})$ is the algebra of \mathscr{O}_X -module endomorphisms of \mathscr{A} . Every Azumaya algebra is locally isomorphic on $X_{\acute{e}t}$ to an endomorphism algebra of a locally free sheaf, see Milne [76, IV Proposition 2.3]. In particular, locally on $X_{\acute{e}t}$, an Azumaya algebra has rank n^2 for some n, giving rise to a global section deg $(\mathscr{A}) \in H^0_{\acute{e}t}(X,\mathbb{Z})$ called the *degree* of \mathscr{A} .

An involution of the first kind σ on an Azumaya \mathscr{O}_X -algebra \mathscr{A} is locally isomorphic on $X_{\text{\acute{e}t}}$, by Proposition 3.1, to an adjoint involution associated to an ϵ -symmetric bilinear form for some $\epsilon \in H^0_{\text{\acute{e}t}}(X, \mu_2)$, called the *type* of the involution. On each connected component, the type is called *orthogonal* or *symplectic* depending on the sign of ϵ .

Proposition 3.3. Let X be a scheme with 2 invertible and $(\mathcal{E}, q, \mathcal{L})$ be a regular ϵ -symmetric bilinear form of rank n on X. Then the groupoid of $\mathbf{PSim}(\mathcal{E}, b, \mathcal{L})$ -torsors is equivalent to the category of Azumaya \mathcal{O}_X -algebras of degree n with involution of type ϵ together with isomorphisms of algebras with involution.

Remark 3.4. The map on cohomology

 $H^1_{\text{\'et}}(X, \mathbf{Sim}(\mathscr{E}, b, \mathscr{L})) \to H^1_{\text{\'et}}(X, \mathbf{PSim}(\mathscr{E}, b, \mathscr{L}))$

associated to sequence (23) has the following interpretation: the similarity class of a line bundlevalued ϵ -symmetric bilinear form is mapped to the adjoint involution on its endomorphism algebra. The coboundary map

$$H^1_{\text{\'et}}(X, \mathbf{PSim}(\mathscr{E}, b, \mathscr{L})) \to H^2_{\text{\'et}}(X, \mathbf{G}_{\mathrm{m}})$$

takes an Azumaya algebra with with involution of type ϵ to the Brauer equivalence class of the algebra. The Brauer class of any Azumaya \mathcal{O}_X -algebra \mathscr{A} supporting an involution of the first kind is 2-torsion. A refinement of this coboundary map is given via the *involutive Brauer group*, see 3.3.

3.2. Unitary involutions and hermitian forms. Let $f: Z \to X$ be étale quadratic and ι the nontrivial X-automorphism of Z. Since ι is an automorphism of order 2, $\iota_*^{-1} = \iota_*$ and ι^* are isomorphic functors on the category of \mathscr{O}_Z -modules. Indeed, for any \mathscr{O}_Z -module \mathscr{H} , a canonical isomorphism $\alpha_{\mathscr{H}}: \iota^* \mathscr{H} \to \iota_* \mathscr{H}$ is given by adjunction applied to the identity map $\mathscr{H} \to \iota_* \iota_* \mathscr{H}$.

For any \mathscr{O}_X -module \mathscr{L} , the unit of adjunction defines an \mathscr{O}_Z -module isomorphism $f^*\mathscr{L} \xrightarrow{\eta} \iota_*\iota^*f^*\mathscr{L} = \iota_*f^*\mathscr{L}$, which we shall (by abuse of notation) also denote by $\iota^{\sharp} : f^*\mathscr{L} \to \iota_*f^*\mathscr{L}$. Note that this also coincides with the composition

$$f^*\mathscr{L} \cong f^*\mathscr{L} \otimes \mathscr{O}_Z \xrightarrow{\mathrm{id}_{f^*\mathscr{L}} \otimes \iota^{\sharp}} f^*\mathscr{L} \otimes \iota_*\mathscr{O}_Z \xrightarrow{\simeq} \iota_*(\iota^*f^*\mathscr{L} \otimes \mathscr{O}_Z) \cong \iota_*f^*\mathscr{L},$$

where the third isomorphism is from the projection formula.

Hermitian forms. A line bundle-valued Z/X-hermitian form is a triple $(\mathcal{H}, h, \mathcal{L})$, where \mathcal{H} is a locally free \mathcal{O}_Z -module of finite rank, \mathcal{L} is an invertible \mathcal{O}_X -module, and $h : \mathcal{H} \otimes \iota_* \mathcal{H} \to f^* \mathcal{L}$ is an \mathcal{O}_Z -module morphism such that the following diagram of \mathcal{O}_Z -modules commute

(25)
$$\begin{aligned} \mathcal{H} \otimes \iota_* \mathcal{H} & \xrightarrow{h} f^* \mathcal{L} \\ & \bigvee_{\iota^{\sharp}} & \downarrow_{\iota^{\sharp}} \\ \iota_* (\mathcal{H} \otimes \iota_* \mathcal{H}) \xrightarrow{\iota_* h} \iota_* f^* \mathcal{L} \end{aligned}$$

where the left vertical map is the composition

$$\mathscr{H} \otimes \iota_* \mathscr{H} \to \iota_* \mathscr{H} \otimes \mathscr{H} \to \iota_* (\mathscr{H} \otimes \iota^* \mathscr{H}) \xrightarrow{\iota_* (\mathrm{id}_{\mathscr{H}} \otimes \alpha_{\mathscr{H}})} \iota_* (\mathscr{H} \otimes \iota_* \mathscr{H})$$

of canonical \mathscr{O}_Z -module morphisms: naïve tensor switch, projection formula, and induced from α . The commutativity of diagram (25) represents the usual formula " $\overline{h(x,y)} = h(y,x)$ " defining a hermitian form.

As in the bilinear form case, a Z/X-hermitian form $(\mathcal{H}, h, \mathcal{L})$ has an \mathcal{O}_Z -module *adjoint* morphism $\psi_h : \mathcal{H} \to \mathcal{H}om(\iota_*\mathcal{H}, f^*\mathcal{L})$. It is called *regular* if ψ_h is an \mathcal{O}_Z -module isomorphism. The commutativity of diagram (25) is equivalent to $\psi_h = \psi_h^{\dagger}$, where ψ_h^{\dagger} is defined as the composition

$$\psi_h^{\dagger}: \mathscr{H} \xrightarrow{\operatorname{ev}^{*\mathscr{L}}} \mathscr{H}om(\iota_*\mathscr{H}om(\iota_*\mathscr{H}, f^*\mathscr{L}), f^*\mathscr{L}) \xrightarrow{\psi_h^{*\mathscr{L}}} \mathscr{H}om(\iota_*\mathscr{H}, f^*\mathscr{L}).$$

Here, $(-)^{\star \mathscr{L}}$ is the contravariant functor $\mathscr{H}om(\iota_*(-), f^*\mathscr{L})$ and

$$\operatorname{ev}^{\star\mathscr{L}}: \operatorname{id} \to ((-)^{\star\mathscr{L}})^{\star\mathscr{L}}$$

is the canonical morphism of functors on the category of coherent \mathscr{O}_Z -modules, which is an isomorphism on the subcategory $\mathsf{VB}(Z)$ of locally free \mathscr{O}_Z -modules. The triple $(\mathsf{VB}(Z), (-)^{\star\mathscr{L}}, \mathrm{ev}^{\star\mathscr{L}})$ forms an exact category with duality, the Witt group of which is studied in Gille [46]. This definition of hermitian form is equivalent to the one found in [67, §3] and (upon taking ι_*) [85, §1.2].

Unitary groups. A similarity (transformation) or similarity between line bundle-valued Z/X-hermitian forms $(\mathcal{H}, h, \mathcal{L})$ and $(\mathcal{H}', h', \mathcal{L}')$ is a pair (φ, λ) , where $\varphi : \mathcal{H} \to \mathcal{H}'$ is an \mathcal{O}_Z -module isomorphism and $\lambda : \mathcal{L} \to \mathcal{L}'$ is an \mathcal{O}_X -module isomorphism such that either of the following (equivalent) diagrams,

(26)
$$\begin{array}{ccc} \mathscr{H} \otimes \iota_{*}\mathscr{H} \xrightarrow{h} f^{*}\mathscr{L} & \mathscr{H} \xrightarrow{\psi_{h}} \mathscr{H}om(\iota_{*}\mathscr{H}, f^{*}\mathscr{L}) \\ \varphi \otimes \iota_{*}\varphi \middle| & & & \downarrow f^{*}\lambda & \varphi \middle| & & & \downarrow f^{*}\lambda^{-1}\varphi^{*\mathscr{L}} \\ \mathscr{H}' \otimes \iota_{*}\mathscr{H}' \xrightarrow{h'} f^{*}\mathscr{L}' & & & & & & & & & \\ \end{array}$$

of \mathscr{O}_Z -modules commute, where $f^*\lambda^{-1}\varphi^{*\mathscr{L}}(\psi) = f^*\lambda^{-1}\circ\psi\circ\varphi$ on sections. A similarity transformation (φ, λ) is an *isometry* if $\mathscr{L} = \mathscr{L}'$ and λ is the identity map.

For $U \to X$, denote by $(\mathscr{H}, h, \mathscr{L})|_U$ the $Z \times_X U/U$ -hermitian form $(\mathscr{H}|_{f^{-1}(U)}, h|_{f^{-1}(U)}, \mathscr{L}|_U)$. Denote by $\mathbf{GU}(\mathscr{H}, h, \mathscr{L})$ (resp. $\mathbf{U}(\mathscr{H}, h, \mathscr{L})$) the presheaf on $X_{\text{\acute{e}t}}$ of similarities (resp. isometries)

 $U \mapsto \{\text{similarities (resp. isometries) } (\varphi, \lambda) : (\mathcal{H}, h, \mathcal{L})|_U \to (\mathcal{H}, h, \mathcal{L})|_U \}$

of the regular \mathscr{L} -valued Z/X-hermitian form $(\mathscr{H}, h, \mathscr{L})$. In fact, these presheaves are sheaves on $X_{\text{\acute{e}t}}$ and are representable by smooth affine reductive group schemes over X, called the *general unitary* (resp. *unitary*) group of the line bundle-valued Z/X-hermitian form $(\mathscr{H}, h, \mathscr{L})$. We will often omit the dependence of these groups on the form $(\mathscr{H}, h, \mathscr{L})$ when no confusion may arise. Even though these sheaves of groups are representable by schemes over X, we will still think of them as sheaves of groups on $X_{\text{\acute{e}t}}$.

The map assigning $(\varphi, \lambda) \mapsto \lambda$ on sections defines the *multiplier coefficient* homomorphism μ : $\mathbf{GU}(\mathscr{H}, h, \mathscr{L}) \to \mathbf{GL}(\mathscr{L}) = \mathbf{G}_{\mathrm{m}}$. As in Proposition 1.4, for any scheme X with 2 invertible, the sequence of sheaves of groups,

$$1 \to \mathbf{U}(\mathscr{H}, h, \mathscr{L}) \to \mathbf{GU}(\mathscr{H}, h, \mathscr{L}) \xrightarrow{\mu} \mathbf{G}_{\mathrm{m}} \to 1,$$

is exact on $X_{\text{ét}}$ and is called the *multiplier sequence*.

Proposition 3.5. Let X be a scheme, $f : Z \to X$ étale quadratic, and $(\mathcal{H}, h, \mathcal{L})$ a fixed regular \mathcal{L} -valued Z/X-hermitian form of rank n.

- a) The groupoid of $U(\mathcal{H}, h, \mathcal{L})$ -torsors is equivalent to the groupoid whose objects are regular \mathcal{L} -valued Z/X-hermitian forms of rank n and whose morphisms are isometries.
- b) The groupoid of GU(&, b, L)-torsors is equivalent to the groupoid whose objects are regular line bundle-valued Z/X-hermitian forms of rank n and whose morphisms are similarity transformations.

Remark 3.6. The multiplier sequence has a cohomological interpretation in analogy with Remark 1.5. If $(\mathcal{H}, h, \mathcal{L})$ is a fixed \mathcal{L} -valued Z/X-hermitian form of rank n, then the map

$$H^1_{\text{\acute{e}t}}(X, \mathbf{U}(\mathscr{H}, h, \mathscr{L})) \to H^1_{\text{\acute{e}t}}(X, \mathbf{GU}(\mathscr{H}, h, \mathscr{L}))$$

takes the isometry class of a regular \mathscr{L} -valued Z/X-hermitian form of rank n to its similarity class. Under the identification $\mathbf{GL}(\mathscr{L}) = \mathbf{G}_{\mathrm{m}}$ the map

$$H^1_{\text{\acute{e}t}}(X, \mathbf{GU}(\mathscr{H}, h, \mathscr{L})) \to H^1_{\text{\acute{e}t}}(X, \mathbf{G}_{\mathrm{m}}) \cong \operatorname{Pic}(X)$$

takes the similarity class of a regular \mathscr{L}' -valued of rank n to the class of \mathscr{L}' in $\operatorname{Pic}(X)$.

Unitary involutions. If \mathscr{B} is an \mathscr{O}_Z -algebra, then an ι -semilinear antiautomorphism of \mathscr{B} is defined to be an \mathscr{O}_Z -algebra isomorphism $\sigma : \mathscr{B} \to \iota_* \mathscr{B}^{\mathrm{op}}$ such that the restriction $\sigma|_{\mathscr{O}_Z}$ equals the canonical sheaf isomorphism $\iota^{\sharp} : \mathscr{O}_Z \to \iota_* \mathscr{O}_Z$. An ι -semilinear antiautomorphism σ is called a Z/X-unitary involution if $\iota_* \sigma^{\mathrm{op}} \circ \sigma = \mathrm{id}_{\mathscr{B}}$. An \mathscr{O}_Z -algebra isomorphism $\sigma : \mathscr{B} \to \iota^* \mathscr{B}^{\mathrm{op}}$ specifies an ι -semilinear antiautomorphism $\alpha_{\mathscr{B}^{\mathrm{op}}} \circ \sigma$ of \mathscr{B} .

A morphism $\varphi : (\mathscr{B}, \sigma) \to (\mathscr{B}', \sigma')$ of \mathscr{O}_Z -algebras with Z/X-unitary involution consists of an \mathscr{O}_Z -algebra homomorphism $\varphi : \mathscr{B} \to \mathscr{B}'$ such that $\sigma' \circ \varphi = \iota_* \varphi^{\mathrm{op}} \circ \sigma$.

Given a regular line bundle-valued Z/X-hermitian form, the ι -semilinear antiautomorphism defined by

$$\sigma_h : \mathscr{E}nd(\mathscr{H}) \to \mathscr{E}nd(\iota_*\mathscr{H})^{\operatorname{op}} \cong \iota_*\mathscr{E}nd(\mathscr{H})^{\operatorname{op}}$$
$$\varphi \mapsto \psi_h^{-1} \circ \varphi^{*\mathscr{L}} \circ \psi_h$$

is a Z/X-unitary involution called the *adjoint involution* of $(\mathcal{H}, h, \mathcal{L})$. Conversely, every Z/Xunitary involution on $\mathscr{E}nd(\mathscr{H})$ arises this way. **Proposition 3.7** (Saltman [88, Thm. 4.2b] or Knus–Parimala–Srinivas [67]). Let X be a scheme, $f: Z \to X$ be étale quadratic, and \mathscr{H} a locally free \mathscr{O}_Z -module of finite rank. If σ is a Z/X-unitary involution on $\mathscr{E}nd(\mathscr{H})$ then there exists a regular Z/X-hermitian form $(\mathscr{H}, h, \mathscr{L})$ such that $\sigma = \sigma_h$. Moreover, the similarity class of $(\mathscr{H}, h, \mathscr{L})$ is uniquely determined by \mathscr{H} and σ .

Remark 3.8. Similarly to Remark 3.2, the isomorphism class of the Z/X-unitary involution $(\mathscr{E}nd(\mathscr{H}), \sigma_h)$ determines the projective similarity class of $(\mathscr{H}, h, \mathscr{L})$.

Let $(\mathscr{H}, h, \mathscr{L})$ be a regular Z/X-hermitian form. For every étale $U \to X$, denote by $(\mathscr{E}nd(\mathscr{H}), \sigma_h)|_U$ the \mathscr{O}_U -algebra with $Z \times_X U/U$ -unitary involution $(\mathscr{E}nd(\mathscr{H}|_{f^{-1}(U)}), \sigma_h|_{f^{-1}(U)})$. Denote by $\mathbf{PGU}(\mathscr{H}, h, \mathscr{L})$ the presheaf of automorphism groups

$$U \mapsto \{Z \times_X U/U$$
-unitary automorphisms $\varphi : (\mathscr{E}nd(\mathscr{H}), \sigma_h)|_U \to (\mathscr{E}nd(\mathscr{H}), \sigma_h)|_U \}$

on $X_{\text{\acute{e}t}}$. In fact, this presheaves is a sheaves on $X_{\text{\acute{e}t}}$, representable as an affine reductive algebraic group, called the *projective unitary similitude* group $\mathbf{PGU}(\mathscr{H}, h, \mathscr{L}) = \mathbf{Aut}_{Z/X}(\mathscr{E}nd(\mathscr{H}), \sigma_h)$. There's a natural homomorphism $\mathbf{GU}(\mathscr{H}, h, \mathscr{L}) \to \mathbf{PGU}(\mathscr{H}, q, \mathscr{L})$ of sheaves of groups on $X_{\text{\acute{e}t}}$, with kernel the central embedding $\mathbf{R}_{Z/X}\mathbf{G}_{\mathrm{m}} \to \mathbf{GU}(\mathscr{E}, b, \mathscr{L})$ given by homotheties. There is an exact sequence

$$1 \to \mathbf{R}_{Z/X}\mathbf{G}_{\mathrm{m}} \to \mathbf{GU}(\mathscr{H}, h, \mathscr{L}) \to \mathbf{PGU}(\mathscr{H}, h, \mathscr{L}) \to 1.$$

of sheaves of groups on $X_{\text{\acute{e}t}}$. The multiplier of a homothety is its norm, thus the inclusion of homotheties restricts to a central inclusion $\mathbf{R}^1_{Z/X}\mathbf{G}_m \to \mathbf{U}(\mathscr{H}, h, \mathscr{L})$, the sheaf cokernel of which defines the *projective unitary* group $\mathbf{PU}(\mathscr{H}, h, \mathscr{L})$. In total, we have a commutative diagram with exact columns and rows

$$(27) \qquad \begin{array}{c} 1 & 1 \\ \downarrow & \downarrow \\ 1 \longrightarrow \mathbf{R}_{Z/X}^{1} \mathbf{G}_{m} \longrightarrow \mathbf{R}_{Z/X} \mathbf{G}_{m} \xrightarrow{N} \mathbf{G}_{m} \longrightarrow 1 \\ \downarrow & \downarrow \\ 1 \longrightarrow \mathbf{U} \longrightarrow \mathbf{GU} \xrightarrow{\mu} \mathbf{G}_{m} \longrightarrow 1 \\ \downarrow & \downarrow \\ \mathbf{PU} \longrightarrow \mathbf{PGU} \\ \downarrow & \downarrow \\ 1 & 1 \end{array}$$

of sheaves of groups on $X_{\text{ét}}$.

Finally, if the \mathscr{O}_Z -rank of \mathscr{H} is m, then $N(\det(\varphi)) = \lambda^{2m}$. Thus we have an induced homomorphism $\det/\mu^m : \mathbf{GU}(\mathscr{H}, h, \mathscr{L}) \to \mathbf{R}^1_{Z/X}\mathbf{G}_m$, whose sheaf kernel is the *proper general unitary* group $\mathbf{SGU}(\mathscr{H}, h, \mathscr{L})$. In analogy with (4), we have an exact sequence

(28)
$$1 \to \mathbf{SGU}(\mathscr{H}, h, \mathscr{L}) \to \mathbf{GU}(\mathscr{H}, h, \mathscr{L}) \xrightarrow{\det /\mu^m} \mathbf{R}^1_{Z/X} \mathbf{G}_{\mathrm{m}} \to 1.$$

Restricting det $/\mu^m$ to the unitary group yields a homomorphism det : $\mathbf{U}(\mathscr{H}, h, \mathscr{L}) \to \mathbf{R}^1_{Z/X} \mathbf{G}_m$, whose sheaf kernel is the *special unitary* group $\mathbf{SU}(\mathscr{H}, h, \mathscr{L})$. In total, we have a commutative diagram with exact columns and rows



of sheaves of groups on $X_{\text{\acute{e}t}}$, in analogy with (5).

ASHER AUEL

3.3. Involutive Brauer group. Parimala–Srinivas [85] construct an "involutive" Brauer group $\operatorname{Br}^*(X)$ of isomorphism classes $[\mathscr{A}, \sigma]$ of Azumaya \mathscr{O}_X -algebras with involution of the first kind modulo classes of adjoint involutions associated to \mathscr{O}_X -valued symmetric bilinear forms. For an étale quadratic $f: Z \to X$, they also construct a "Z/X-unitary" Brauer group $\operatorname{Br}^*(Z/X)$ of isomorphism classes of Azumaya \mathscr{O}_Z -algebras with Z/X-unitary involution modulo classes of adjoint involutions associated to \mathscr{O}_X -valued symmetric bilinear forms. For an étale sociated to \mathscr{O}_X -valued Z/X-hermitian forms. The following result shows that $\operatorname{Br}^*(X)$ is a refinement of the 2-torsion subgroup $_2\operatorname{Br}(X)$ and provides a new interpretation for $H^2_{\operatorname{\acute{e}t}}(X, \mu_2)$.

We have canonical isomorphisms $H^i_{\text{ét}}(X, \mathbf{R}_{Z/X}\mathbf{G}_m) \simeq H^i_{\text{\acute{e}t}}(Z, \mathbf{G}_m)$ from the degeneration of the Leray spectral sequence associated to the étale morphism $f: Z \to X$. Combined with the long exact sequence of étale cohomology groups associated to (7) and the canonical identification $H^1_{\text{\acute{e}t}}(X, \mathbf{G}_m) =$ Pic(X), we have the exact sequence

$$\cdots \to H^1_{\text{\'et}}(X, \mathbf{R}^1_{Z/X}\mathbf{G}_{\mathrm{m}}) \to \operatorname{Pic}(Z) \xrightarrow{N} \operatorname{Pic}(X) \to H^1_{\text{\'et}}(X, \mathbf{R}^1_{Z/X}\mathbf{G}_{\mathrm{m}}) \to H^2_{\text{\'et}}(Z, \mathbf{G}_{\mathrm{m}}) \to \cdots$$

which defines, for every invertible \mathscr{O}_X -module \mathscr{L} a class $c_1(\mathscr{L}, \mathbf{R}^1_{Z/X} \mathbf{G}_m) \in H^2_{\text{\acute{e}t}}(X, \mathbf{R}^1_{Z/X} \mathbf{G}_m)$.

Theorem 3.9 (Parimala-Srinivas [85, Th. 1, Th. 2]). Let X be a scheme with 2 invertible.

a) There is a canonical homomorphism Ψ : $\operatorname{Br}^*(X) \to H^2_{\operatorname{\acute{e}t}}(X, \mu_2)$ fitting into a commutative diagram

$$Br^{*}(X) \xrightarrow{\Psi} H^{2}_{\text{ét}}(X, \mu_{2})$$

$$\downarrow \qquad \qquad \downarrow$$

$$_{2}Br(X) \longrightarrow _{2}H^{2}_{\text{ét}}(X, \mathbf{G}_{m})$$

where the vertical maps are the natural forgetful maps. If $(\mathscr{E}, b, \mathscr{L})$ is a regular line bundlevalued (skew-)symmetric bilinear form on X, then $\Psi[\mathscr{E}nd(\mathscr{E}), \sigma_b] = c_1(\mathscr{L}, \mu_2)$.

b) If $f: Z \to X$ is étale quadratic, then there is a canonical homomorphism $\Psi_{Z/X} : \operatorname{Br}^*(Z/X) \to H^2_{\operatorname{\acute{e}t}}(X, \mathbf{R}^1_{Z/X} \mathbf{G}_m)$ fitting into a commutative diagram

$$\begin{array}{c} \operatorname{Br}^*(Z/X) \xrightarrow{\Psi_{Z/X}} H^2_{\operatorname{\acute{e}t}}(X, \mathbf{R}^1_{Z/X} \mathbf{G}_m) \\ \downarrow & \downarrow \\ \operatorname{Br}(Z) \xrightarrow{} H^2_{\operatorname{\acute{e}t}}(Z, \mathbf{G}_m) \end{array}$$

where the left vertical map is the natural forgetful map and the right vertical map is induced from cohomology applied to (7). If $(\mathcal{H}, h, \mathcal{L})$ is a regular line bundle-valued Z/X-hermitian form, then $\Psi_{Z/X}[\mathscr{E}nd(\mathcal{H}), \sigma_h] = -c_1(\mathcal{L}, \mathbf{R}^1_{Z/X} \mathbf{G}_m).$

Remark 3.10. The forgetful maps $\operatorname{Br}^*(X) \to {}_2\operatorname{Br}(X)$ and $\operatorname{Br}^*(Z/X) \to \operatorname{ker}(H^2_{\operatorname{\acute{e}t}}(Y, \mathbf{G}_{\mathrm{m}}) \xrightarrow{N} H^2_{\operatorname{\acute{e}t}}(X, \mathbf{G}_{\mathrm{m}}))$ are surjective, since every 2-torsion (resp. trivial norm) Brauer class is represented by an Azumaya algebra with involution of the first kind (resp. unitary involution). This follows from the construction in Knus–Parimala–Srinivas [67], which generalize results of Saltman [88] and Albert [1, X.9 Thm. 19], see also [69, I.3].

Now suppose that the map $_{2}\text{Br}(X) \to _{2}H^{2}_{\text{\acute{e}t}}(X, \mathbf{G}_{m})$ is surjective, i.e. that every 2-torsion class in the cohomological Brauer group is represented by an Azumaya algebra. This is the case when X is a quasi-compact quasi-separated scheme admitting an ample invertible sheaf by de Jong's extension [29] (see also [74, Th. 2.2.2.1]) of a result of Gabber [41]. Then by the above remark and Theorem 3.9, $\text{Br}^{*}(X) \to H^{2}_{\text{\acute{e}t}}(X, \mu_{2})$ is surjective. Furthermore, restricting Theorem 3.9 to the subgroup $\text{Br}^{+}(X) \subset \text{Br}^{*}(X)$ consisting of classes of Azumaya algebras with orthogonal involution yields an isomorphism $\text{Br}^{+}(X) \simeq H^{2}_{\text{\acute{e}t}}(X, \mu_{2})$.

One can reinterpret this in the language of twisted sheaves: while a torsion \mathbf{G}_{m} -twisted sheaf on a sufficiently nice scheme is represented by an Azumaya algebra, a μ_2 -twisted sheaf is represented by an Azumaya algebra with (orthogonal) involution.

Proof of Theorem 3.9. We will review details of the proof inasmuch as they are necessary for our purposes. We will also provide new torsorial proofs of the statements concerning classes arising from adjoint involutions. The sign discrepancy between the final formula of b) and the final formula of Parimala–Srinivas [85, Thm. 2] (the proof of which was left to the reader) is most likely due to different conventions for associating \mathbf{G}_{m} -torsors to invertible sheaves.

As for a), for each $n \geq 1$ and $\epsilon \in H^0_{\text{\acute{e}t}}(X, \mu_2)$, denote by $\mathbf{Isom}_{\epsilon,n}$ the isometry group of the form $\langle 1, \ldots, 1 \rangle$ on connected components where $\epsilon = 1$ and the standard skew-symmetric hyperbolic form of rank n (if n is even) on connected components where $\epsilon = -1$. Similarly to Remark 3.4, the map $H^1_{\text{\acute{e}t}}(X, \mathbf{Isom}_{n,\epsilon}) \to H^1_{\text{\acute{e}t}}(X, \mathbf{PIsom}_{n,\epsilon})$ associated to (24) is interpreted as: the isometry

class of a regular ϵ -symmetric form (\mathscr{E}, b) maps to the isomorphism class of its adjoint involution $(\mathscr{E}nd(\mathscr{E}), \sigma_b)$. In particular, the coboundary map $\Psi_n : H^1_{\acute{e}t}(X, \mathbf{PIsom}_{n,\epsilon}) \to H^2_{\acute{e}t}(X, \mu_2)$ vanishes on adjoint algebras of \mathscr{O}_X -valued ϵ -symmetric bilinear forms and so factors through $\mathrm{Br}^*(X)$. The map $\Psi : \mathrm{Br}^*(X) \to H^2_{\acute{e}t}(X, \mu_2)$ is defined to be the limit of these coboundary maps over classes of Azumaya algebras with involution of fixed degree. The fact that the map is a group homomorphism is verified in [85]. The commutativity of the diagram in a follows from the interpretation on cohomology of the commutative diagram with exact rows



of sheaves of groups on $X_{\text{ét}}$.

As for the final statement of a), first note that since $c_1(\mathscr{L}) = c_1(\mathscr{N}^{\otimes 2} \otimes \mathscr{L})$, the 1st Chern class of the value line bundle is an invariant of the projective similarity class (see §3.1) of a line bundlevalued bilinear form, and is therefore an invariant of the \mathscr{O}_X -algebra isomorphism class of its adjoint involution. Now we apply the Roman IX Lemma 5.17 to the Roman IX rearrangement (Remark 5.18) of diagram (24), which yields the (anti)commutative pentagon

$$H^{1}_{\text{ét}}(X, \mathbf{PSim}_{n,\epsilon}) \xrightarrow{\Psi_{n}} H^{2}_{\text{ét}}(X, \boldsymbol{\mu}_{2})$$

$$H^{1}_{\text{ét}}(X, \mathbf{Sim}_{n,\epsilon}) \xrightarrow{H^{1}_{\text{ét}}(X, \mathbf{G}_{m})} H^{2}_{\text{ét}}(X, \boldsymbol{\mu}_{2})$$

of cohomology sets. Following the similarity class of a line bundle-valued ϵ -symmetric bilinear form $(\mathscr{E}, b, \mathscr{L})$ around the (anti)commuting pentagon yields

$$\Psi_n[\mathscr{E}nd(\mathscr{E}),\sigma_b] = -c_1(\mathscr{L},\boldsymbol{\mu}_2) = c_1(\mathscr{L},\boldsymbol{\mu}_2) \in H^2_{\text{\'et}}(X,\boldsymbol{\mu}_2).$$

A similar technique has been used in Căldăraru [24, Prop. 5.2.4].

As for b), the same argument carries over upon replacing $\operatorname{Isom}_{n,\epsilon}$ by \mathbf{U}_n^Z , the unitary group of the standard Z/X-hermitian form of \mathscr{O}_Z -rank n. In this case, short exact sequence (23) is replaced by

$$1 \to \mathbf{R}^1_{Z/X} \mathbf{G}_{\mathrm{m}} \to \mathbf{U}^Z_n \to \mathbf{P} \mathbf{U}^Z_n \to 1$$

whose coboundary map $\Psi_n^Z : H^1_{\text{ét}}(X, \mathbf{PU}_n^Z) \to H^2_{\text{\acute{et}}}(X, \mathbf{R}^1_{Z/X}\mathbf{G}_m)$ factor through $\operatorname{Br}^*(Z/X)$. We define $\Psi_{Z/X} : \operatorname{Br}^*(Z/X) \to H^1_{\text{\acute{et}}}(X, \mathbf{R}^1_{Z/X}\mathbf{G}_m)$ as a limit of the maps Ψ_n^Z . Diagram (24) is replaced by the corresponding diagram (27), and the the Roman IX Lemma 5.17 and rearrangement (Remark 5.18) yields the anticommutative pentagon

$$H^{1}_{\text{ét}}(X, \mathbf{G}\mathbf{U}_{n}^{Z}) \xrightarrow{\Psi^{2}_{n}} H^{2}_{\text{ét}}(X, \mathbf{R}^{1}_{Z/X}\mathbf{G}_{m})$$
$$H^{1}_{\text{ét}}(X, \mathbf{G}\mathbf{U}_{n}^{Z}) \xrightarrow{H^{1}_{\text{ét}}(X, \mathbf{G}_{m})} H^{2}_{\text{ét}}(X, \mathbf{R}^{1}_{Z/X}\mathbf{G}_{m})$$

of cohomology sets. Then following the similarity class of a regular line bundle-valued Z/X-hermitian form $(\mathcal{H}, h, \mathcal{L})$ around the anticommuting pentagon yields

$$\Psi_{Z/X}[\mathscr{E}nd(\mathscr{H}),\sigma_h] = -c_1(\mathscr{L},\mathbf{R}^1_{Z/X}\mathbf{G}_m) \in H^2_{\mathrm{\acute{e}t}}(X,\mathbf{R}^1_{Z/X}\mathbf{G}_m).$$

Note that constructing a proof of 3.9b) was left to the reader in Parimala–Srinivas [85].

3.4. Involutive Brauer class of the even Clifford algebra. We present a generalization of results of Parimala–Srinivas [85, §2] to even Clifford algebras of line bundle-valued quadratic forms. Given a line bundle-valued quadratic form $(\mathscr{E}, q, \mathscr{L})$, the type of the *canonical involution* (see 1.8) τ_0 on $\mathscr{C}_0(\mathscr{E}, q, \mathscr{L})$ is given as in the classical case over fields.

Proposition 3.11. Let $(\mathscr{E}, q, \mathscr{L})$ be a line bundle-valued quadratic form on X.

- a) If n is odd, then $\mathscr{C}_0(\mathscr{E}, q, \mathscr{L})$ is an Azumaya \mathscr{O}_X -algebra and
 - (a) if $n \equiv 1, 7 \mod 8$ then τ_0 is of orthogonal type, while
 - (b) if $n \equiv 3, 5 \mod 8$ then τ_0 is of symplectic type.
- b) If n is even, then $\mathcal{C}_0(\mathcal{E}, q, \mathcal{L})$ is an Azumaya \mathcal{O}_Z -algebra, where $Z \to X$ is the Arf covering of $(\mathcal{E}, q, \mathcal{L})$.

- (a) if $n \equiv 0 \mod 8$ then τ_0 is of orthogonal type on Z,
- (b) if $n \equiv 4 \mod 8$ then τ_0 is of symplectic type on Z, while
- (c) if $n \equiv 2 \mod 4$ then τ_0 is of Z/X-unitary type.

Proof. Since the type of an involution is local in the étale topology, we may reduce to the case of symmetric bilinear forms on affine schemes, where we can appeal to [85, Lemma 2]. Of course, the same proposition holds for even Clifford algebras of Azumaya algebras with orthogonal involution. \Box

Recall the notation of Example 2.6. For each even $n = 2m \ge 2$ and each étale quadratic $f : Z \to X$, let $(\mathscr{H}, b) = h_n^Z = (f_*\mathscr{O}_Z, h^Z) \perp H_{\mathscr{O}_X}(\mathscr{O}_X^{m-1})$, where $(f_*\mathscr{O}_Z, h^Z)$ is the norm form associated to $f : Z \to X$ (see §5.1).

3.4.1. Canonical involutions of the first type. If $n = 2m \equiv 0 \mod 4$, the even Clifford algebra gives rise, via the involutive Brauer group, to a class

$$[\mathscr{C}_0(\mathscr{E}, q, \mathscr{L}), \tau_0] \in H^2_{\text{ét}}(Z, \mu_2)$$

where $Z \to X$ is the Arf invariant of $(\mathscr{E}, q, \mathscr{L})$. If $Z \to X$ is split, then any choice of global splitting idempotent induces a decomposition of \mathscr{O}_X -algebras with involution $(\mathscr{C}_0(\mathscr{E}, q, \mathscr{L}), \tau_0) \cong$ $(\mathscr{C}_0^+(\mathscr{E}, q, \mathscr{L}), \tau_0^+) \times (\mathscr{C}_0^-(\mathscr{E}, q, \mathscr{L}), \tau_0^-)$, where τ_0^{\pm} are the restrictions of τ_0 . Note that $\mathscr{C}_0^{\pm}(\mathscr{E}, q, \mathscr{L})$ are Azumaya \mathscr{O}_X -algebras of degree 2^{m-1} , where n = 2m.

Recall that there's an isomorphism of group schemes $\mathbf{z}_n^Z \cong \mathbf{R}_{Z/X}\boldsymbol{\mu}_2$ for m even, see [69, VI.26.A] for example. For Z/X split, the choice of an isomorphism $\mathbf{z}_n^Z = \mathbf{z}_{m,m} \cong \boldsymbol{\mu}_2 \times \boldsymbol{\mu}_2$ (as in §2.8) yields a fixed isomorphism $\mathbf{z}_n^Z \cong \mathbf{R}_{Z/X}\boldsymbol{\mu}_2$. We now provide a generalization of [69, VII.31.11] to our setting, which can be viewed as an involutive refinement of the Tits algebra in the ${}^1\mathsf{D}_m$ and ${}^2\mathsf{D}_m$ cases, for m even.

Proposition 3.12. Let X be a scheme with 2 invertible, $f : Z \to X$ étale quadratic, and $n = 2m \equiv 0 \mod 4$. Then the coboundary map

$$H^1_{\text{\'et}}(X, \mathbf{PSO}_n^Z) \to H^2_{\text{\'et}}(X, \mathbf{z}_n^Z) \cong H^2_{\text{\'et}}(X, \mathbf{R}_{Z/X} \boldsymbol{\mu}_2) \cong H^2_{\text{\'et}}(Z, \boldsymbol{\mu}_2)$$

has the following interpretation: a class associated to the adjoint involution of $(\mathcal{E}, q, \mathcal{L})$ (for some choice of orientation) maps to the involutive Brauer class $[\mathcal{C}_0(\mathcal{E}, q, \mathcal{L}), \tau_0]$.

Proof. As in Remark 1.10, the natural homomorphism $\mathbf{PSO}_n^Z \to \mathbf{R}_{Z/X} \mathbf{PIso}(\widetilde{\mathscr{C}}_0(h_n^Z), \tau_0)$ induces a map

$$H^1_{\text{\'et}}(X, \mathbf{PSO}^Z_n) \to H^1_{\text{\'et}}(Z, \mathbf{PIso}(\mathscr{C}_0(h^Z_n), \tau_0))$$

with the following interpretation: a class associated to the adjoint involution of $(\mathscr{E}, q, \mathscr{L})$ maps to the \mathscr{O}_Z -isomorphism class of its even Clifford algebra $\mathscr{C}_0(\mathscr{E}, q, \mathscr{L})$. This homomorphism fits into a commutative diagram

$$\begin{split} 1 & \longrightarrow \mathbf{z}_{n}^{Z} & \longrightarrow \mathbf{Spin}_{n}^{Z} & \longrightarrow \mathbf{PSO}_{n}^{Z} & \longrightarrow 1 \\ & \downarrow & \downarrow & \downarrow \\ 1 & \longrightarrow \mathbf{R}_{Z/X} \boldsymbol{\mu}_{2} & \longrightarrow \mathbf{R}_{Z/X} \mathbf{Iso}(\widetilde{\mathcal{C}_{0}}(h_{n}^{Z}), \tau_{0}) & \longrightarrow \mathbf{R}_{Z/X} \mathbf{PIso}(\widetilde{\mathcal{C}_{0}}(h_{n}^{Z}), \tau_{0}) & \longrightarrow 1 \end{split}$$

of sheaves of groups on $X_{\acute{e}t}$ with exact rows. The proposition then follows from chasing the class of $(\mathscr{E}nd(\mathscr{E}), \sigma_q)$ around the diagram

$$\begin{array}{c} H^1_{\mathrm{\acute{e}t}}(X,\mathbf{PSO}^Z_n) \longrightarrow H^2_{\mathrm{\acute{e}t}}(X,\mathbf{z}^Z_n) \\ & \swarrow \\ H^1_{\mathrm{\acute{e}t}}(Z,\mathbf{PIso}(\mathscr{C}_0(h^Z_n),\tau_0)) \longrightarrow H^2_{\mathrm{\acute{e}t}}(Z,\boldsymbol{\mu}_2) \end{array}$$

of coboundary maps. The statement (and proof) of the proposition hold, more generally, for even Clifford algebras of Azumaya algebras of degree $\equiv 0 \mod 4$ with orthogonal involution.

Finally, we come the main result of this section. We will denote the following composition

$$H^{2}_{\text{\acute{e}t}}(X,\boldsymbol{\kappa}_{n}^{Z}) \cong H^{2}_{\text{\acute{e}t}}(X,\mathbf{R}_{Z/X}^{1}\boldsymbol{\mu}_{4}) \to H^{2}_{\text{\acute{e}t}}(X,\mathbf{R}_{Z/X}\boldsymbol{\mu}_{4}) \cong H^{2}_{\text{\acute{e}t}}(Z,\boldsymbol{\mu}_{4})$$

by φ^2 , where the first isomorphism is induced from Example 2.6. Denote by $i: \mu_2 \to \mu_4$ the canonical homomorphism.

Theorem 3.13. Let X be a scheme with 2 invertible, $f : Z \to X$ étale quadratic, and $n = 2m \equiv 0 \mod 4$. Let $(\mathscr{E}, q, \mathscr{L})$ be a regular quadratic form of rank n and Arf invariant [Z/X]. Then we have

$$\varphi^2 gc(\mathscr{E}, q, \mathscr{L}) = i^2 [\mathscr{C}_0(\mathscr{E}, q, \mathscr{L}), \tau_0] + c_1(f^* \mathscr{L}, \boldsymbol{\mu}_4)$$

in $H^{2}_{\text{ét}}(Z, \mu_{4})$.

Proof. This proof may be seen as a refinement of [69, VII Exer. 15a]. Let $\tilde{\kappa}_n^Z$ be the subgroup scheme of $\mathbf{S}\Gamma_n^Z$ defined by

$$\tilde{\boldsymbol{\kappa}}_n^Z(U) = \{ x \in \mathbf{S} \boldsymbol{\Gamma}_n^Z(U) : N(x), r(x) \in \boldsymbol{\mu}_2(U) \}.$$

Note that $\tilde{\kappa}_n^Z$ is the kernel in the short exact sequence of group schemes

(29)
$$1 \to \tilde{\kappa}_n^Z \to \mathbf{S} \mathbf{\Gamma}_n^Z \to \mathbf{G} \mathbf{S} \mathbf{O}_n^Z / \boldsymbol{\mu}_2 \to 1$$

on $X_{\text{\acute{e}t}}$. Restricting the homomorphisms N and $s = N \cdot r$ to $\tilde{\kappa}_n^Z$ yields exact sequences

$$1 \to \mathbf{z}_n^Z \xrightarrow{\tilde{\mathbf{i}}_z} \tilde{\mathbf{\kappa}}_n^Z \xrightarrow{N} \mu_2 \to 1$$
$$1 \to \mathbf{\kappa}_n^Z \xrightarrow{\tilde{\mathbf{i}}_{\mathbf{\kappa}}} \tilde{\mathbf{\kappa}}_n^Z \xrightarrow{s} \mu_2 \to 1$$

fitting into commutative diagrams with exact rows and columns



of sheaves of groups on $X_{\text{\acute{e}t}}$. The existence of these diagrams is guaranteed by the Nine Lemma 5.16.

Proposition 3.14. Let X be a scheme with 2 invertible and $(\mathscr{E}, q, \mathscr{L})$ a regular \mathscr{L} -valued quadratic form of rank n on X. The groupoid of $\mathbf{GO}(\mathscr{E}, q, \mathscr{L})/\mu_2$ -torsors is equivalent to the category of 2torsion data with involution, whose objects are tuples $(\mathscr{A}, \sigma, \mathscr{V}, b, \mathscr{M}, \varphi)$, where (\mathscr{A}, σ) is an Azumaya \mathscr{O}_X -algebra of degree n with orthogonal involution, $(\mathscr{V}, b, \mathscr{M})$ is a regular \mathscr{M} -valued quadratic form of rank n^2 , and $\varphi : (\mathscr{A}, \sigma) \otimes (\mathscr{A}, \sigma) \to (\mathscr{E}nd(\mathscr{V}), \sigma_b)$ is an isomorphism of \mathscr{O}_X -algebras with involution and whose morphisms between objects $(\mathscr{A}, \sigma, \mathscr{V}, b, \mathscr{M}, \varphi)$ and $(\mathscr{A}', \sigma', \mathscr{V}', b', \mathscr{M}', \varphi')$ are tuples (ψ, g, λ) consisting of an isomorphism $\psi : (\mathscr{A}, \sigma) \to (\mathscr{A}', \sigma')$ of \mathscr{O}_X -algebras with involution and a similarity $(g, \lambda) : (\mathscr{V}, b, \mathscr{M}) \to (\mathscr{V}', b', \mathscr{M}')$ satisfying $\varphi' \circ (\psi \otimes \psi) = \operatorname{Ad}_g \circ \varphi$.

Proof. This is similar (but slightly more involved) to the description of \mathbf{GL}_n/μ_2 -torsors in terms of 2-torsion data, see 4.2 and Auel [4].

The map of pointed sets $\alpha : H^1_{\acute{e}t}(X, \mathbf{GO}) \to H^1_{\acute{e}t}(X, \mathbf{GO}/\mu_2)$ induced from the quotient homomorphism has the following interpretation: the class of $(\mathscr{E}, q, \mathscr{L})$ maps to the tuple $(\mathscr{E}nd(\mathscr{E}), \sigma_q, \mathscr{E} \otimes \mathscr{E}, q \otimes q, \mathscr{L}^{\otimes 2}, \varphi_{\mathscr{E}})$, where $\varphi_{\mathscr{E}} : \mathscr{E}nd(\mathscr{E}) \otimes \mathscr{E}nd(\mathscr{E}) \to \mathscr{E}nd(\mathscr{E}\otimes \mathscr{E})$ is the canonical \mathscr{O}_X -algebra isomorphism.

The map of pointed sets $\beta : H^1_{\text{ét}}(X, \mathbf{PGO}) \cong H^1_{\text{\acute{et}}}(X, \mathbf{PO}) \to H^1_{\text{\acute{et}}}(X, \mathbf{GO}/\mu_2)$ has the following interpretation: the class of (\mathscr{A}, σ) maps to the tuple $(\mathscr{A}, \sigma, \mathscr{A}, \operatorname{Trd}_{(\mathscr{A}, \sigma)}, \mathscr{O}_X, \varphi_{\mathscr{A}})$, where $\operatorname{Trd}_{(\mathscr{A}, \sigma)}$ is the trace form

$$\begin{aligned} \operatorname{Trd}_{(\mathscr{A},\sigma)} &: \mathscr{A} \otimes \mathscr{A} \to \mathscr{O}_X \\ & a \otimes b \quad \mapsto \operatorname{Trd}_{\mathscr{A}}(a\,\sigma(b)). \end{aligned}$$

Consider the commutative diagram with exact rows and columns



of sheaves of groups on $X_{\text{\acute{e}t}}$. There's an induced action of $\operatorname{Pic} X = H^1_{\text{\acute{e}t}}(X, \mathbf{G}_m)$ on $H^1_{\text{\acute{e}t}}(X, \mathbf{GO}/\mu_2)$, which is described by

$$\mathscr{L} \cdot (\mathscr{A}, \sigma, \mathscr{V}, b, \mathscr{M}, \varphi) = (\mathscr{A}, \sigma, \mathscr{L} \otimes \mathscr{V}, \mathrm{id} \otimes b, \mathscr{L}^{\otimes 2} \otimes M, \varphi)$$

where we consider $\varphi : (\mathscr{A}, \sigma) \otimes (\mathscr{A}, \sigma) \to (\mathscr{E}nd(\mathscr{V}), \sigma_b) = (\mathscr{E}nd(\mathscr{L} \otimes \mathscr{V}), \sigma_{\mathrm{id} \otimes b})$ via the canonical identification $\mathscr{E}nd(\mathscr{L} \otimes \mathscr{V}) = \mathscr{E}nd(\mathscr{V})$.

By Remark 3.4, the map of pointed sets $H^1_{\text{\acute{e}t}}(X, \mathbf{GO}) \to H^1_{\text{\acute{e}t}}(X, \mathbf{PGO}) \cong H^1_{\text{\acute{e}t}}(X, \mathbf{PO})$ induced from the quotient homomorphism has the following interpretation: the class $(\mathscr{E}, q, \mathscr{L})$ maps to its adjoint involution $(\mathscr{E}nd(\mathscr{E}), \sigma_q)$.

Claim 3.15. Let $(\mathscr{E}, q, \mathscr{L})$ be a regular symmetric bilinear form. We have the following formula

$$\beta(\mathscr{E}nd(\mathscr{E}),\sigma_q) = \mathscr{L}^{\vee} \cdot \alpha(\mathscr{E},q,\mathscr{L})$$

Proof. By a generalization of [69, II Prop. 11.4], the composition of \mathcal{O}_X -module isomorphisms

$$\mathscr{E}\otimes\mathscr{E}\otimes\mathscr{L}^\vee\xrightarrow{\psi_q^{-1}}\mathscr{E}\otimes\mathscr{E}^\vee\xrightarrow{\mathrm{can}}\mathscr{E}nd(\mathscr{E})$$

is an isometry of the symmetric bilinear forms

$$(\mathscr{E} \otimes \mathscr{E} \otimes \mathscr{L}^{\vee}, q \otimes q \otimes \mathrm{id}_{\mathscr{L}^{\vee}}, \mathscr{O}_X) \cong (\mathscr{E}nd(\mathscr{E}), \mathrm{Trd}_{(\mathscr{E}nd(\mathscr{E}), \sigma_q)}, \mathscr{O}_X),$$

where

$$q \otimes q \otimes \operatorname{id}_{\mathscr{L}^{\vee}} : \left(\mathscr{E} \otimes \mathscr{E} \otimes \mathscr{L}^{\vee} \right) \otimes \left(\mathscr{E} \otimes \mathscr{E} \otimes \mathscr{L}^{\vee} \right) \to \mathscr{L}^{\otimes 2} \otimes (\mathscr{L}^{\vee})^{\otimes 2} \xrightarrow{\operatorname{ev}} \mathscr{O}_X.$$

Through this isometry, the map β applied to the adjoint involution of $(\mathscr{E}, q, \mathscr{L})$ is identified with $(\mathscr{E}nd(\mathscr{E}), \sigma_q, \mathscr{E} \otimes \mathscr{E} \otimes \mathscr{L}^{\vee}, q \otimes q \otimes \operatorname{id}_{\mathscr{L}^{\vee}}, \mathscr{O}_X, \varphi_{\mathscr{E}})$, where

$$\varphi_{\mathscr{E}}: \mathscr{E}nd(\mathscr{E}) \otimes \mathscr{E}nd(\mathscr{E}) \to \mathscr{E}nd(\mathscr{E} \otimes \mathscr{E}) \cong \mathscr{E}nd(\mathscr{E} \otimes \mathscr{E} \otimes \mathscr{L}^{\vee}).$$

But this is precisely α applied to $(\mathscr{E}, q, \mathscr{L})$ scaled by the class of \mathscr{L}^{\vee} .

We leave it as an exercise to the reader to make the adjustments to Proposition 3.14, Claim 3.15, and the above cohomological interpretations necessary to describe torsors corresponding to \mathbf{PGSO}/μ_2 (all objects and morphisms are oriented, preserving the discriminant form or center of the Clifford algebra).

Let $\tilde{gc}(\mathscr{A}, \sigma, \mathscr{V}, b, \mathscr{M}, \varphi) \in H^2_{\text{\acute{e}t}}(X, \tilde{\kappa}^Z_n)$ be the coboundary map arising from sequence (29) applied to a choice of Z/X-orientation on the tuple $(\mathscr{A}, \sigma, \mathscr{V}, b, \mathscr{M}, \varphi)$. Similarly as in Definition 2.8, by Lemma 5.15, the invariant \tilde{gc} is independent of the choice of orientation. Considering the central rows of the right-hand diagram (30), we have $\tilde{gc} \circ \alpha = \tilde{\imath}^2_{\kappa} \circ gc$. Hence, by Proposition 3.12 and the commutativity two left columns of the left-hand diagram (30), we have

(31)
$$\widetilde{gc}(\mathscr{L}^{\vee} \cdot \alpha(\mathscr{E}, q, \mathscr{L})) = \widetilde{gc}(\beta(\mathscr{E}nd(\mathscr{E}), \sigma_q)) = \widetilde{i}_{\mathbf{z}}^2[\mathscr{C}_0(\mathscr{E}, q, \mathscr{L}), \tau_0]$$

in $H^2_{\text{ét}}(X, \tilde{\kappa}_n^Z)$.

Now we identify the effect of scaling by $H^1_{\text{\acute{e}t}}(X, \mathbf{G}_m)$ on the coboundary map of sequence (29). Restricting the homomorphism r to $\tilde{\kappa}_n^Z$ yields an exact sequence

$$1 \to \boldsymbol{\mu}_4 \xrightarrow{\imath_{\boldsymbol{\mu}_4}} \tilde{\boldsymbol{\kappa}}_n^Z \xrightarrow{r} \boldsymbol{\mu}_2 \to 1$$

fitting into the commutative diagram with exact rows and columns



of sheaves of groups on $X_{\text{\acute{e}t}}$. For $\xi \in H^1_{\text{\acute{e}t}}(X, \mathbf{GSO}/\mu_2)$ and $\mathscr{L} \in H^1_{\text{\acute{e}t}}(X, \mathbf{G}_m)$, we then have

$$\widetilde{gc}(\mathscr{L}^{\vee}\cdot\xi) = c_1(\mathscr{L}^{\vee},\boldsymbol{\mu}_4) + \widetilde{gc}(\xi)$$

in $H^2_{\text{ét}}(X, \tilde{\kappa}^Z_n)$, by an application of Lemma 5.19 to the above diagram. Applying this to (31) yields

(32)
$$\tilde{\imath}^{2}_{\boldsymbol{\mu}_{4}}c_{1}(\mathscr{L}^{\vee},\boldsymbol{\mu}_{4}) + \tilde{g}c(\alpha(\mathscr{E},q,\mathscr{L})) = \tilde{\imath}^{2}_{\mathbf{z}}\left[\mathscr{C}_{0}(\mathscr{E},q,\mathscr{L}),\tau_{0}\right]$$

in $H^2_{\text{\acute{e}t}}(X, \tilde{\kappa}^Z_n)$. This is, in fact, a refinement of the statement of the theorem. To finally finish the proof, we construct an embedding $j : \tilde{\kappa}^Z_n \hookrightarrow \mathbf{R}_{Z/X} \boldsymbol{\mu}_4$, so that applying $j^2 : H^2_{\text{\acute{e}t}}(X, \tilde{\kappa}^Z_n) \to H^2_{\text{\acute{e}t}}(X, \mathbf{R}_{Z/X} \boldsymbol{\mu}_4) \cong H^2_{\text{\acute{e}t}}(Z, \boldsymbol{\mu}_4)$ to (32) yields the statement of the theorem.

To this end, first note that the diagram



is a pushout in the category of sheaves of abelian groups on $X_{\text{\acute{e}t}}$. Indeed, the internal product of the maps $\tilde{\imath}_{\kappa}$ and $\tilde{\imath}_{\mathbf{z}}$ fits into an exact sequence

$$1 \to \boldsymbol{\mu}_2 \xrightarrow{\Delta} \boldsymbol{\kappa}_n^Z \times \mathbf{z}_n^Z \xrightarrow{\tilde{\imath}_{\boldsymbol{\kappa}} \cdot \tilde{\imath}_{\mathbf{z}}} \tilde{\boldsymbol{\kappa}}_n^Z \to 1$$

presenting $\tilde{\kappa}_n^Z$ as the explicit pushout of the canonical inclusions of $\mu_2 \hookrightarrow \mathbf{S} \Gamma_n^Z$ into κ_n^Z and \mathbf{z}_n^Z . Considering the homomorphisms $\kappa_n^Z \cong \mathbf{R}_{Z/X}^1 \mu_4 \to \mathbf{R}_{Z/X} \mu_4$ and $\mathbf{z}_n^Z \cong \mathbf{R}_{Z/X} \mu_2 \to \mathbf{R}_{Z/X} \mu_4$, the universal property of the pushout supplies a unique homomorphism $j : \tilde{\kappa}_n^Z \to \mathbf{R}_{Z/X} \mu_4$.

Now denoting by N^2 the composition of homomorphisms $\mathbf{R}_{Z/X} \mu_4 \xrightarrow{N} \mu_4 \xrightarrow{2} \mu_2$, we can complete the diagrams provided by the universal property of the pushout to the following commutative diagrams with exact rows and columns

$$1 \longrightarrow \kappa_n^Z \xrightarrow{\tilde{\imath}_{\kappa}} \tilde{\kappa} \xrightarrow{\tilde{\imath}_{\kappa}} \mu_2 \longrightarrow 1 \qquad 1 \longrightarrow \mathbf{z}_n^Z \xrightarrow{\tilde{\imath}_z} \tilde{\kappa} \xrightarrow{N} \mu_2 \longrightarrow 1 \qquad 1 \longrightarrow \mathbf{z}_n^Z \xrightarrow{\tilde{\imath}_z} \tilde{\kappa} \xrightarrow{N} \mu_2 \longrightarrow 1 \qquad 1 \longrightarrow \mathbf{z}_n^Z \xrightarrow{\tilde{\imath}_z} \tilde{\kappa} \xrightarrow{N} \mu_2 \longrightarrow 1 \qquad 1 \longrightarrow \mathbf{z}_n^Z \xrightarrow{\tilde{\imath}_z} \tilde{\kappa} \xrightarrow{N} \mu_2 \longrightarrow 1 \qquad 1 \longrightarrow \mathbf{z}_n^Z \xrightarrow{1} \mu_2 \longrightarrow$$

of sheaves of abelian groups on $X_{\text{\acute{e}t}}$. Similarly, we have the commutative diagram with exact rows and columns



of sheaves of abelian groups on $X_{\text{ét}}$, the bottom square of which commutes since $x^2 \iota(x)^2 = (x^2/\iota(x)^2)\iota(x)^4 = x^2/\iota(x)^2$ for sections x of $\mathbf{R}_{Z/X}\boldsymbol{\mu}_4$, where ι is the nontrivial automorphism of $Z \to X$ (see §2.2). In particular, the compositions $j \circ \tilde{\imath}_{\kappa}$, $j \circ \tilde{\imath}_{\mathbf{z}}$, and $j \circ \tilde{\imath}_{\mu_4}$ are identified with the canonical inclusions of the respective subgroups into $\mathbf{R}_{Z/X}\boldsymbol{\mu}_4$. As a consequence, we have identifications

$$j^2\circ \tilde{\imath}^2_{\pmb{\kappa}}=\varphi^2,\quad j^2\circ \tilde{\imath}^2_{\mathbf{z}}=i^2,\quad j^2\circ \tilde{\imath}^2_{\pmb{\mu}_4}=f^*$$

of the respective maps after taking $H_{\text{\acute{e}t}}^2$. Finally, applying j^2 to (32) yields the statement of the theorem.

3.4.2. Canonical involutions of the second type. If $n = 2m \equiv 2 \mod 4$, the even Clifford algebra gives rise, via the Z/X-unitary involutive Brauer group, to a class

$$[\mathscr{C}_0(\mathscr{E}, q, \mathscr{L}), \tau_0] \in H^2_{\text{ét}}(X, \mathbf{R}^1_{Z/X} \mathbf{G}_{\mathrm{m}})$$

where $Z \to X$ is the Arf covering of $(\mathscr{E}, q, \mathscr{L})$. If $Z \to X$ is split (equivalently, $d_{\pm}(\mathscr{E}, q, \mathscr{L})$ is trivial), then any choice of global splitting idempotent induces a decomposition of \mathscr{O}_X -algebras $\mathscr{C}_0(\mathscr{E}, q, \mathscr{L}) \cong$ $\mathscr{C}_0^+(\mathscr{E}, q, \mathscr{L}) \times \mathscr{C}_0^-(\mathscr{E}, q, \mathscr{L})$. By a globalization of Bichsel [18, §5.3] (also see [68, IV Thm. 9.2.2] or [69, I Prop. 2.14]), the canonical involution τ_0 induces an \mathscr{O}_X -algebra isomorphism $\mathscr{C}_0^+(\mathscr{E}, q, \mathscr{L})^{\mathrm{op}} \cong$ $\mathscr{C}_0^-(\mathscr{E}, q, \mathscr{L})$, and so $\mathscr{C}_0(\mathscr{E}, q, \mathscr{L})$ is a hyperbolic ring. There's an isomorphism of group schemes $\mathbf{z}_n^Z \cong \mathbf{R}_{Z/X}^1 \boldsymbol{\mu}_4$ for m odd, see [69, VI.26.A]. In fact, for

There's an isomorphism of group schemes $\mathbf{z}_n^Z \cong \mathbf{R}_{Z/X}^2 \boldsymbol{\mu}_4$ for m odd, see [69, VI.26.A]. In fact, for Z/X split, the choice of an isomorphism $\mathbf{z}_n^Z = \mathbf{z}_{m,m} \cong \boldsymbol{\mu}_4$ (as in §2.8) yields a fixed isomorphism $\mathbf{z}_n^Z \cong \mathbf{R}_{Z/X}^1 \boldsymbol{\mu}_4$ (using the method of Remark 2.6). We now provide a generalization of [69, VII.31.9] to our setting, which can be viewed as a unitary involutive refinement of the Tits algebra in the ${}^1\mathsf{D}_m$ and ${}^2\mathsf{D}_m$ cases, for m odd.

Proposition 3.16. Let X be a scheme with 2 invertible, $f : Z \to X$ étale quadratic, and $n = 2m \equiv 2 \mod 4$. Then the composition of maps

$$H^1_{\text{\'et}}(X, \mathbf{PSO}_n^Z) \to H^2_{\text{\'et}}(X, \mathbf{z}_n^Z) \cong H^2_{\text{\'et}}(X, \mathbf{R}^1_{Z/X} \boldsymbol{\mu}_4) \to H^2_{\text{\'et}}(X, \mathbf{R}^1_{Z/X} \mathbf{G}_m)$$

has the following interpretation: a class associated to the adjoint involution of $(\mathcal{E}, q, \mathcal{L})$ maps to the Z/X-unitary involutive Brauer class $[\mathcal{C}_0(\mathcal{E}, q, \mathcal{L}), \tau_0]$.

Proof. The argument is similar to the proof of Proposition 3.12. As in Remark 1.10, the natural homomorphism $\mathbf{PSO}_n^Z \to \mathbf{PU}(\widetilde{\mathscr{C}}_0(h_n^Z), \tau_0)$ induces a map

$$H^1_{\text{\'et}}(X, \mathbf{PSO}_n^Z) \to H^1_{\text{\'et}}(X, \mathbf{PU}(\mathscr{C}_0(h), \tau_0))$$

with the following interpretation: a class associated to the adjoint involution of $(\mathscr{E}, q, \mathscr{L})$ maps to the \mathscr{O}_Z -algebra $(\widetilde{\mathscr{C}}_0(\mathscr{E}, q, \mathscr{L}), \tau_0)$ with Z/X-unitary involution.

This homomorphism fits into a commutative diagram

of sheaves of groups on $X_{\text{ét}}$ with exact rows. The proposition then follows by chasing the class of $(\mathscr{E}nd(\mathscr{E}), \sigma_q)$ around the diagram

$$\begin{array}{c} H^1_{\mathrm{\acute{e}t}}(X,\mathbf{PSO}^Z_n) \longrightarrow H^2_{\mathrm{\acute{e}t}}(X,\mathbf{z}^Z_n) \\ & \swarrow \\ H^1_{\mathrm{\acute{e}t}}(Z,\mathbf{PU}(\mathscr{C}_0(h^Z_n),\tau_0)) \longrightarrow H^2_{\mathrm{\acute{e}t}}(X,\mathbf{R}^1_{Z/X}\mathbf{G}_m) \end{array}$$

of coboundary maps. The statement (and proof) of the proposition hold, more generally, for even Clifford algebras of Azumaya algebras of degree $\equiv 2 \mod 4$ with orthogonal involution.

Finally, we come the main result of this section, which is analogous to, but weaker than, Theorem 3.13. We will denote the following composition

$$H^{2}_{\text{\acute{e}t}}(X, \boldsymbol{\kappa}_{n}^{Z}) \cong H^{2}_{\text{\acute{e}t}}(X, \mathbf{R}_{Z/X}\boldsymbol{\mu}_{2}) \cong H^{2}_{\text{\acute{e}t}}(Z, \boldsymbol{\mu}_{2}) \to H^{2}_{\text{\acute{e}t}}(Z, \mathbf{G}_{m})$$

by ϕ^2 , where the first isomorphism is induced from Example 2.6. Denote by $i: \mu_2 \to \mathbf{G}_m$ the canonical homomorphism.

Theorem 3.17. Let X be a scheme with 2 invertible, $f : Z \to X$ étale quadratic, and $n = 2m \equiv 2 \mod 4$. Let $(\mathscr{E}, q, \mathscr{L})$ be a regular quadratic form of rank n and Arf invariant [Z/X]. Then we have

$$\phi^2 gc(\mathscr{E}, q, \mathscr{L}) = [\widetilde{\mathscr{C}_0}(\mathscr{E}, q, \mathscr{L})]$$

in $H^2_{\text{ét}}(Z, \mathbf{G}_m)$.

Proof. The proof is similar to that of Theorem 3.13. Let $\widetilde{\mathbf{G}}_n^Z$ be the subgroup scheme of $\mathbf{S}\mathbf{\Gamma}_n^Z$ defined by

$$\widetilde{\mathbf{G}}_n^Z(U) = \{ x \in \mathbf{S} \mathbf{\Gamma}_n^Z(U) : r(x) \in \boldsymbol{\mu}_2(U) \}.$$

Note that $\widetilde{\mathbf{G}}_n^Z$ is the kernel in the short exact sequence of group schemes

(33)
$$1 \to \widetilde{\mathbf{G}}_n^Z \to \mathbf{S} \mathbf{\Gamma}_n^Z \to \mathbf{P} \mathbf{S} \mathbf{O}_n^Z \to 1$$

on $X_{\text{\acute{e}t}}$. Restricting the homomorphisms N and $s = N \cdot r$ to $\widetilde{\mathbf{G}}_n^Z$ yields exact sequences

$$\begin{split} 1 &\to \mathbf{z}_n^Z \xrightarrow{\tilde{\imath}_{\mathbf{z}}} \widetilde{\mathbf{G}}_n^Z \xrightarrow{N} \mathbf{G}_{\mathbf{m}} \to 1 \\ 1 &\to \kappa_n^Z \xrightarrow{\tilde{\imath}_{\mathbf{\kappa}}} \widetilde{\mathbf{G}}_n^Z \xrightarrow{s} \mathbf{G}_{\mathbf{m}} \to 1 \end{split}$$

fitting into commutative diagrams with exact rows and columns



of sheaves of groups on $X_{\text{\acute{e}t}}$. The inclusion of the scalar torus $\mathbf{G}_{\mathrm{m}} \to \widetilde{\mathbf{G}}_{n}^{Z}$ can be completed to an exact sequence

$$1 \to \mathbf{G}_{\mathrm{m}} \xrightarrow{i_{\mathbf{G}_{\mathrm{m}}}} \widetilde{\mathbf{G}}_{n}^{Z} \to \boldsymbol{\mu}_{2} \to 1$$

which fits into the commutative diagram with exact rows and columns



of sheaves of groups on $X_{\text{ét}}$. The existence of these diagrams is guaranteed by the Nine Lemma 5.16.

Let $t: H^1_{\text{\'et}}(X, \mathbf{PSO}_n^Z) \cong H^1_{\text{\'et}}(X, \mathbf{PGO}_n^Z) \to H^2_{\text{\'et}}(X, \mathbf{z}_n^Z)$ be the coboundary map of (18). We will write $t(\mathscr{E}, q, \mathscr{L}) = t(\mathscr{E}nd(\mathscr{E}), \sigma_q)$. Now chasing the class of $(\mathscr{E}nd(\mathscr{E}), \sigma_q)$ through the upper rows of the right-hand diagram and the left-most columns of the left-hand diagram (34), we have

(35)
$$\tilde{\imath}_{\kappa}^2 gc(\mathscr{E}, q, \mathscr{L}) = \tilde{\imath}_{\mathbf{z}}^2 t(\mathscr{E}, q, \mathscr{L})$$

in $H^2_{\text{\acute{e}t}}(X, \widetilde{\mathbf{G}}_n^Z)$. This is, in fact, a refinement of the statement of the theorem. To finally finish the proof, we construct an embedding $j : \widetilde{\mathbf{G}}_n^Z \to \mathbf{R}_{Z/X}\mathbf{G}_m$, so that applying $j^2 : H^2_{\text{\acute{e}t}}(X, \widetilde{\mathbf{G}}_n^Z) \to H^2_{\text{\acute{e}t}}(X, \mathbf{R}_{Z/X}\mathbf{G}_m) \cong H^2_{\text{\acute{e}t}}(Z, \mathbf{G}_m)$ to (35) yields the statement of the theorem.

To this end, first note that the diagram



is a pushout in the category of sheaves of abelian groups on $X_{\text{\acute{e}t}}$. Indeed, the internal product of the maps $\tilde{\imath}_{\kappa}$ and $\tilde{\imath}_{\mathbf{G}_{m}}$ fits into an exact sequence

$$1 \to \boldsymbol{\mu}_2 \xrightarrow{\Delta} \boldsymbol{\kappa}_n^Z \times \mathbf{G}_{\mathrm{m}} \xrightarrow{\tilde{\imath}_{\boldsymbol{\kappa}} \cdot \tilde{\imath}_{\mathbf{G}_{\mathrm{m}}}} \widetilde{\mathbf{G}}_n^Z \to 1$$

presenting $\widetilde{\mathbf{G}}_n^Z$ as the explicit pushout of the canonical inclusions of $\boldsymbol{\mu}_2 \hookrightarrow \mathbf{S} \boldsymbol{\Gamma}_n^Z$ into $\boldsymbol{\kappa}_n^Z$ and \mathbf{G}_m . Considering the homomorphisms $\boldsymbol{\kappa}_n^Z \cong \mathbf{R}_{Z/X} \boldsymbol{\mu}_2 \to \mathbf{R}_{Z/X} \mathbf{G}_m$ and $\mathbf{G}_m \to \mathbf{R}_{Z/X} \mathbf{G}_m$, the universal property of the pushout supplies a unique homomorphism $j : \widetilde{\mathbf{G}}_n^Z \to \mathbf{R}_{Z/X} \mathbf{G}_m$.

Now denoting by $(\mathrm{id}/\iota)^2$ the composition of homomorphisms $\mathbf{R}_{Z/X}\mathbf{G}_{\mathrm{m}} \xrightarrow{\mathrm{id}/\iota} \mathbf{R}_{Z/X}^1\mathbf{G}_{\mathrm{m}} \xrightarrow{2} \mathbf{R}_{Z/X}^1\mathbf{G}_{\mathrm{m}}$, we can complete the diagrams provided by the universal property of the pushout to the following commutative diagrams with exact rows and columns

$$1 \longrightarrow \kappa_n^Z \xrightarrow{\tilde{\imath}_{\kappa}} \widetilde{\mathbf{G}}_n^Z \xrightarrow{s} \mathbf{G}_m \longrightarrow 1 \qquad 1 \longrightarrow \mathbf{G}_m \xrightarrow{\tilde{\imath}_{\mathbf{G}_m}} \widetilde{\mathbf{G}}_n^Z \xrightarrow{s} \mu_2 \longrightarrow 1 \\ \downarrow^{\tilde{\imath}_{\mathbf{G}_m}} \downarrow^{j} \downarrow^{j}$$

of sheaves of abelian groups on $X_{\text{\acute{e}t}}$. Similarly, we have the commutative diagram with exact rows and columns



of sheaves of abelian groups on $X_{\text{\acute{e}t}}$. In particular, the compositions $j \circ \tilde{\imath}_{\kappa}$, $j \circ \tilde{\imath}_{\mathbf{G}_{m}}$, and $j \circ \tilde{\imath}_{\mathbf{z}}$ are identified with the canonical inclusions of the respective subgroups into $\mathbf{R}_{Z/X}\mathbf{G}_{m}$. As a consequence, we have identifications

$$j^2\circ \tilde{\imath}^2_{\boldsymbol{\kappa}}=\phi^2,\quad j^2\circ \tilde{\imath}^2_{\mathbf{G}_{\mathrm{m}}}=f^*,\quad j^2\circ \tilde{\imath}^2_{\mathbf{z}}=\phi^2_{\mathbf{z}}$$

of the respective maps after taking $H^2_{\text{\'et}}$. Here, $\phi^2_{\mathbf{z}}$ is the composition of maps $H^2_{\text{\'et}}(X, \mathbf{z}^Z_n) \cong H^2_{\text{\'et}}(X, \mathbf{R}^1_{Z/X} \boldsymbol{\mu}_4) \rightarrow H^2_{\text{\'et}}(X, \mathbf{R}^1_{Z/X} \mathbf{G}_m)$ from Proposition 3.16, which satisfies $\phi^2_{\mathbf{z}} t(\mathscr{E}, q, \mathscr{L}) = [\widetilde{\mathscr{C}}_0(\mathscr{E}, q, \mathscr{L}), \tau_0]$ in $H^2_{\text{\'et}}(Z, \mathbf{G}_m)$. Finally, applying j^2 to (35) yields the statement of the theorem.

4. Relationship to the Clifford bimodule

As usual, let X be a noetherian separated scheme with 2 invertible and let \mathscr{L} be a fixed invertible \mathscr{O}_X -module. While the results in §3 comparing the (refined) Tits algebra to (involutive Brauer classes associated to) the even Clifford algebra were quite neat, the results comparing the similarity Clifford invariant to the even Clifford algebra were less so. In this section, we shall see that the similarity Clifford invariant much more elegantly reflects involutive structures arising from the Clifford bimodule.

4.1. Clifford bimodule. As in the case of central simple algebras with orthogonal involution, line bundle-valued quadratic forms do not generally enjoy a "full" Clifford algebra, of which the even Clifford algebra (see §1.8) is the even degree part. As a replacement, algebras with orthogonal involution have a Clifford bimodule (see [69, II §9]) while line bundle-valued quadratic forms enjoy a refinement of this notion of Clifford bimodule. If $(\mathscr{E}, q, \mathscr{L})$ is a regular line bundle-valued quadratic form, then there exists a locally free \mathscr{O}_X -module $\mathscr{C}_1(\mathscr{E}, q, \mathscr{L})$ with the structure of an invertible $\mathscr{C}_0(\mathscr{E}, q, \mathscr{L})$ -bimodule. The corresponding Clifford bimodule defined in [69] is then isomorphic to $\mathscr{E} \otimes \mathscr{C}_1(\mathscr{E}, q, \mathscr{L})$.

Let $(\mathscr{E}, q, \mathscr{L})$ be a quadratic form on X and $\mathscr{C}'(\mathscr{E}, q, \mathscr{L})$ its generalized Clifford algebra (see §1.8). Define the *Clifford bimodule* $\mathscr{C}_1(\mathscr{E}, q, \mathscr{L})$ of $(\mathscr{E}, q, \mathscr{L})$ to be the degree 1 submodule of $\mathscr{C}'(\mathscr{E}, q, \mathscr{L})$. It's also possible to construct the Clifford bimodule via a direct tensorial construction or by gluing the odd degree parts of Clifford algebras over a Zariski cover trivializing \mathscr{L} , as in §1.8.

Functorial properties. The Clifford bimodule has the following functorial properties (assuming that $(\mathscr{E}, q, \mathscr{L})$ is a regular quadratic form of rank n on X):

a) There is a canonical embedding of locally free \mathscr{O}_X -modules

$$i: \mathscr{E} \to \mathscr{C}_1(\mathscr{E}, q, \mathscr{L}).$$

b) Via multiplication in the generalized Clifford algebra, $\mathscr{C}_1(\mathscr{E}, q, \mathscr{L})$ is an invertible $\mathscr{C}_0(\mathscr{E}, q, \mathscr{L})$ bimodule and there's a canonical isomorphism

$$m: \mathscr{C}_1(\mathscr{E}, q, \mathscr{L}) \otimes_{\mathscr{C}_0(\mathscr{E}, q, \mathscr{L})} \mathscr{C}_1(\mathscr{E}, q, \mathscr{L}) \xrightarrow{\sim} \mathscr{C}_0(\mathscr{E}, q, \mathscr{L}) \otimes_{\mathscr{O}_X} \mathscr{L}$$

of $\mathscr{C}_0(\mathscr{E}, q, \mathscr{L})$ -bimodules (with trivial action on \mathscr{L}) satisfying

$$m(i(v) \otimes i(v)) = 1 \otimes q(v)$$

for a section v of \mathscr{E} . See [18, §2] or [17, Lemma 3.1].

c) Any similarity transformation $(\varphi, \lambda) : (\mathscr{E}, q, \mathscr{L}) \to (\mathscr{E}', q', \mathscr{L}')$ induces an \mathscr{O}_X -module isomorphism

$$\mathscr{C}_1(\varphi,\lambda):\mathscr{C}_1(\mathscr{E},q,\mathscr{L})\to\mathscr{C}_1(\mathscr{E}',q',\mathscr{L}').$$

that is $\mathscr{C}_0(\varphi, \lambda)$ -semilinear according to the diagram

See [18, Prop. 2.6].

- d) If n = 2m is even, and \mathscr{Z} is the center of $\mathscr{C}_0(\mathscr{E}, q, \mathscr{L})$ then $\mathscr{C}_1(\mathscr{E}, q, \mathscr{L})$ is a \mathscr{Z} -bimodule satisfying $x \cdot z = \iota(z) \cdot x$ for sections z of \mathscr{Z} and x of $\mathscr{C}_1(\mathscr{E}, q, \mathscr{L})$, where ι is the nontrivial element of $\operatorname{Gal}(\mathscr{Z}/\mathscr{O}_X)$. See [68, IV Prop. 4.3.1(4)].
- e) There is an \mathscr{O}_X -module isomorphism $\tau_1 : \mathscr{C}_1(\mathscr{E}, q, \mathscr{L}) \to \mathscr{C}_1(\mathscr{E}, q, \mathscr{L})$ of order 2 satisfying

$$\tau_1(axb) = \tau_0(b)\tau_1(x)\tau_0(a), \quad (\tau_0 \otimes \mathrm{id}_{\mathscr{L}})(m(x \otimes y)) = m(\tau_1(y) \otimes \tau_1(x)), \quad \tau_1(i(v)) = v$$

for sections a, b of $\mathscr{C}_0(\mathscr{E}, q, \mathscr{L}), x, y$ of $\mathscr{C}_1(\mathscr{E}, q, \mathscr{L}), and v$ of \mathscr{E} .

f) Any regular bilinear form $(\mathcal{N}, n, \mathcal{N}^{\otimes 2})$ of rank 1, induces an \mathcal{O}_X -algebra isomorphism

$$\mathscr{C}_1(n \otimes \mathrm{id}_{\mathscr{E}}) : \mathscr{C}_1(\mathscr{E}, q, \mathscr{L}) \otimes \mathscr{N} \to \mathscr{C}_1(\mathscr{N} \otimes \mathscr{E}, n \otimes q, \mathscr{N}^{\otimes 2} \otimes \mathscr{L})$$

g) For any morphism of schemes $g: X' \to X$, there's a canonical \mathcal{O}_X -module isomorphism

$$g^* \mathscr{C}_1(\mathscr{E}, q, \mathscr{L}) \cong \mathscr{C}_1(g^*(\mathscr{E}, q, \mathscr{L})).$$

When a regular line bundle-valued quadratic form $(\mathscr{E}, q, \mathscr{L})$ of rank n is fixed, we will write \mathscr{C}_0 for the even Clifford algebra (as an \mathscr{O}_X -algebra) and \mathscr{C}_1 for the Clifford bimodule (as an \mathscr{O}_X -module). When n is even, write $Z = \operatorname{Spec} \mathscr{Z}$ with structure morphism $f: Z \to X$, and denote by $\widetilde{\mathscr{C}}_0$ and $\widetilde{\mathscr{C}}_1$ the associated \mathscr{O}_Z -algebra (see §1.8) and $\widetilde{\mathscr{C}}_0$ -bimodule, respectively. Note that $\widetilde{\mathscr{C}}_1$ has a left and right \mathscr{O}_Z -module structure, which are interchanged by twisting by ι .

Given \mathscr{O}_Z -bimodules \mathscr{P} and \mathscr{P}' , take care that $\mathscr{P} \otimes_{\mathscr{O}_Z} \mathscr{P}'$ stands for the tensor product with respect to the right \mathscr{O}_Z -action of \mathscr{P} and the left \mathscr{O}_Z -action of \mathscr{P}' , and is again an \mathscr{O}_Z -bimodule.

Given an étale quadratic $f : Z \to X$ with Galois group generated by ι and an \mathscr{O}_Z -module \mathscr{B} , consider the naïve "switch" morphism $s_{\mathscr{B}} : \mathscr{B} \otimes_{\mathscr{O}_Z} \mathscr{B} \to \mathscr{B} \otimes_{\mathscr{O}_Z} \mathscr{B}$ and the ι -semilinear "switch" morphism

$$s_{\mathscr{B}}^{\iota}:\mathscr{B}\otimes_{\mathscr{O}_{Z}}\iota_{*}\mathscr{B}\to\iota_{*}(\mathscr{B}\otimes_{\mathscr{O}_{Z}}\iota_{*}\mathscr{B})$$
$$a\otimes\iota_{*}b\mapsto\iota_{*}(b\otimes\iota_{*}a).$$

Furthermore, if $\beta : \mathscr{B} \to \mathscr{B}$ is an \mathscr{O}_Z -module isomorphism, denote by $\operatorname{Ad}_{\beta} : \mathscr{E}nd(\mathscr{B}) \to \mathscr{E}nd(\mathscr{B})$ the induced \mathscr{O}_Z -algebra isomorphism $\varphi \mapsto \tau \circ \varphi \circ \tau^{-1}$; if β is a ι -semilinear isomorphism $\mathscr{B} \to \iota_*\mathscr{B}'$, denote by

$$\operatorname{Ad}_{\beta}^{\iota}: \mathscr{E}\!nd_{\mathscr{O}_{Z}}(\mathscr{B}) \xrightarrow{\operatorname{Ad}_{\beta}} \mathscr{E}\!nd_{\mathscr{O}_{Z}}(\iota_{*}\mathscr{B}') \cong \iota_{*}\mathscr{E}\!nd(\mathscr{B}')$$

the induced ι -semilinear \mathscr{O}_Z -algebra automorphism.

4.2. Torsion data. The notion of a torsion datum on an Azumaya algebra is a way of rigidifying it's realizability as a torsion class in the Brauer group. This has many benefits when working in the groups $H^2_{\text{ét}}(X, \mu_n)$ as opposed to ${}_n\text{Br}(X)$. A torsion datum should also be thought of as a "preinvolution."

2-torsion data. Consider the cokernel of the exact sequence of sheaves of groups

(36)
$$1 \rightarrow \boldsymbol{\mu}_2 \rightarrow \mathbf{GL}_n \rightarrow \mathbf{GL}_n / \boldsymbol{\mu}_2 \rightarrow 1,$$

on $X_{\text{\acute{e}t}}$. Since 2 is invertible on X, the sheaf of groups \mathbf{GL}_n/μ_2 is representable by a smooth affine reductive group scheme on X. Note that $\mathbf{GL}_1/\mu_2 \cong \mathbf{G}_m$ via the Kummer sequence, but that in general, \mathbf{GL}_n/μ_2 is not isomorphic to \mathbf{GL}_n for $n \geq 2$. The category of \mathbf{GL}_n/μ_2 -torsors on $X_{\text{\acute{e}t}}$ is equivalent to the category of 2-torsion data, see Knus [68, III §9.3].

Definition 4.1. A 2-torsion datum of degree n on X is a triple $(\mathscr{A}, \mathscr{P}, \varphi)$, consisting of an Azumaya \mathscr{O}_X -algebra \mathscr{A} of degree n, a locally free \mathscr{O}_X -module \mathscr{P} of rank n^2 , and an \mathscr{O}_X -algebra isomorphism $\varphi : \mathscr{A} \otimes \mathscr{A} \cong \mathscr{E}nd(\mathscr{P})$. In particular, the class of \mathscr{A} in the Brauer group has period ≤ 2 . A morphism of 2-torsion data is a pair $(\psi, \beta) : (\mathscr{A}, \mathscr{P}, \varphi) \to (\mathscr{A}', \mathscr{P}', \varphi')$, where $\psi : \mathscr{A} \to \mathscr{A}'$ is an \mathscr{O}_X -algebra isomorphism and $\beta : \mathscr{P} \to \mathscr{P}'$ is an \mathscr{O}_X -module isomorphism satisfying $\varphi' \circ (\psi \otimes \psi) = \operatorname{Ad}_\beta \circ \varphi$.

To every locally free \mathscr{O}_X -module \mathscr{V} of rank n, we associate a *split datum* $(\mathscr{E}nd(\mathscr{V}), \mathscr{V} \otimes \mathscr{V}, \varphi_{\mathscr{V}})$ of degree n, where

$$\varphi_{\mathscr{V}}: \mathscr{E}\!\mathit{nd}(\mathscr{V})\otimes \mathscr{E}\!\mathit{nd}(\mathscr{V}) \simeq \mathscr{E}\!\mathit{nd}(\mathscr{V}\otimes \mathscr{V})$$

is the canonical \mathscr{O}_X -algebra isomorphism.

Proposition 4.2. Let X be a scheme with 2 invertible. The category of 2-torsion data of degree n on X is equivalent to the category of \mathbf{GL}_n/μ_2 -torsors for the étale topology.

Proof. The proof involves completing the exercise [69, VII Exer. 5] on $X_{\acute{e}t}$. See Auel [4].

The interpretation of exact sequence (36) on cohomology

$$H^1_{\text{\acute{e}t}}(X, \mathbf{GL}_n) \to H^1_{\text{\acute{e}t}}(X, \mathbf{GL}_n/\boldsymbol{\mu}_2) \to H^2_{\text{\acute{e}t}}(X, \boldsymbol{\mu}_2)$$

is as follows: the class associated to a locally free \mathscr{O}_X -module \mathscr{V} of rank n maps to the class of the associated split datum $(\mathscr{E}nd(\mathscr{V}), \mathscr{V} \otimes \mathscr{V}, \varphi_{\mathscr{V}})$; the class of 2-torsion datum $(\mathscr{A}, \mathscr{P}, \varphi)$ maps (by the 2nd coboundary) to a class $[\mathscr{A}, \mathscr{P}, \varphi] \in H^2_{\text{\acute{e}t}}(X, \mu_2)$, which we call the *involutive class* associated to $(\mathscr{A}, \mathscr{P}, \varphi)$, and which we shall now describe.

Given a 2-torsion datum $(\mathscr{A}, \mathscr{P}, \varphi)$, Knus–Parimala–Srinivas [67, Thm. 4.1] construct a canonical involution σ_{φ} of the first kind (in fact, of orthogonal type) on the \mathscr{O}_X -algebra $\mathscr{E}nd_{\mathscr{A}}(\mathscr{A} \oplus \mathscr{P})$, hence an involutive Brauer class.

Lemma 4.3. We have an equality of classes $[\mathscr{A}, \mathscr{P}, \varphi] = \Psi[\mathscr{E}nd_{\mathscr{A}}(\mathscr{A} \oplus \mathscr{P}), \sigma_{\varphi}]$ in $H^2_{\text{\acute{e}t}}(X, \mu_2)$.

Proof. Restricting the homomorphism H from the proof of Theorem 2.11 to \mathbf{GL}_n yields $\mathbf{GL}_n \to \mathbf{O}_{n,n}$ fitting into a morphism of short exact sequences

of sheaves of groups on $X_{\text{\acute{e}t}}$. The map $H^1_{\acute{e}t}(X, \mathbf{GL}_n/\mu_2) \to H^1_{\acute{e}t}(X, \mathbf{PO}_{n,n})$ has the following interpretation: the class associated to a 2-torsion datum $(\mathscr{A}, \mathscr{P}, \varphi)$ of rank n is mapped to the class of the \mathscr{O}_X -algebra with orthogonal involution $(\mathscr{E}nd(\mathscr{A} \oplus \mathscr{P}), \sigma_{\varphi})$. This is stated, for example, in [69, VII Exer. 6]. A comparison of the coboundary maps finishes the proof.

Z/X-torsion data. Now, let $f : Z \to X$ be étale quadratic. Via the central inclusion $\mathbf{R}_{Z/X} \mathbf{G}_{\mathrm{m}} \to \mathbf{R}_{Z/X} \mathbf{G} \mathbf{L}_n$, consider the cokernel of the exact sequence of sheaves of groups

(37)
$$1 \to \mathbf{R}_{Z/X}^{1} \mathbf{G}_{\mathrm{m}} \to \mathbf{R}_{Z/X} \mathbf{G} \mathbf{L}_{n} \to \mathbf{R}_{Z/X} \mathbf{G} \mathbf{L}_{n} / \mathbf{R}_{Z/X}^{1} \mathbf{G}_{\mathrm{m}} \to 1,$$

on $X_{\text{\acute{e}t}}$. Note that $\mathbf{R}_{Z/X}\mathbf{GL}_1/\mathbf{R}_{Z/X}^1\mathbf{G}_m \cong \mathbf{G}_m$ via the norm map.

Recall the definition of the norm or corestriction \mathscr{O}_X -module (resp. algebra) $N_{Z/X}(\mathscr{A})$ of an \mathscr{O}_Z -module (resp. algebra) \mathscr{A} . The norm is a functor from the category of \mathscr{O}_Z -modules (resp. algebras) to the category of \mathscr{O}_X -modules (resp. algebras).

Definition 4.4 ([69, VII Exer. 7]). A Z/X-torsion datum of degree n on X is a triple $(\mathscr{A}, \mathscr{P}, \varphi)$, consisting of an Azumaya \mathscr{O}_Z -algebra \mathscr{A} of degree n, a locally free \mathscr{O}_X -module \mathscr{P} of rank n^2 , and an \mathscr{O}_X -algebra isomorphism $\varphi : N_{Z/X}(\mathscr{A}) \cong \mathscr{E}nd(\mathscr{P})$. A morphism of Z/X-torsion data is a pair $(\psi, \beta) : (\mathscr{A}, \mathscr{P}, \varphi) \to (\mathscr{A}', \mathscr{P}', \varphi')$, where $\psi : \mathscr{A} \to \mathscr{A}'$ is an \mathscr{O}_Z -algebra isomorphism and $\beta : \mathscr{P} \to \mathscr{P}'$ is an \mathscr{O}_X -module isomorphism satisfying $\varphi' \circ N_{Z/X}(\psi) = \operatorname{Ad}_\beta \circ \varphi$.

To every locally free \mathscr{O}_Z -module \mathscr{V} of rank n, we associate a split datum $(\mathscr{E}nd(\mathscr{V}), N_{Z/X}(\mathscr{V}), \varphi_{\mathscr{V}})$, where

$$\varphi_{\mathscr{V}}: N_{Z/X}(\mathscr{E}nd(\mathscr{V})) \cong \mathscr{E}nd(N_{Z/X}(\mathscr{V}))$$

is the canonical \mathcal{O}_X -algebra isomorphism.

Proposition 4.5. Let X be a scheme with 2 invertible and $f: Z \to X$ étale quadratic. The category of Z/X-torsion data of degree n is equivalent to the category of $\mathbf{R}_{Z/X}\mathbf{GL}_n/\mathbf{R}_{Z/X}^1\mathbf{G}_m$ -torsors on $X_{\text{\acute{e}t}}$.

Proof. The proof involves completing the exercise [69, VII Exer. 7] on $X_{\text{ét}}$. See Auel [4].

The interpretation of exact sequence (37) on cohomology

$$H^1_{\text{\acute{e}t}}(Z, \mathbf{GL}_n) \to H^1_{\text{\acute{e}t}}(X, \mathbf{R}_{Z/X}\mathbf{GL}_n/\mathbf{R}^1_{Z/X}\mathbf{G}_m) \to H^2_{\text{\acute{e}t}}(X, \mathbf{R}^1_{Z/X}\mathbf{G}_m)$$

is as follows: the class associated to a locally free \mathscr{O}_X -module \mathscr{V} of rank n maps to the class of the associated split datum $(\mathscr{E}nd(\mathscr{V}), \mathscr{V} \otimes \mathscr{V}, \varphi_{\mathscr{V}})$; the class of 2-torsion datum $(\mathscr{A}, \mathscr{P}, \varphi)$ maps (by

the 2nd coboundary) to a class $[\mathscr{A}, \mathscr{P}, \varphi] \in H^2_{\text{\acute{e}t}}(X, \mathbf{R}^1_{Z/X} \mathbf{G}_m)$, which we call the *involutive class* associated to $(\mathscr{A}, \mathscr{P}, \varphi)$, and which we shall now describe.

Given a Z/X-torsion datum ($\mathscr{A}, \mathscr{P}, \varphi$), Knus–Parimala–Srinivas [67, Thm. 4.2] construct a canonical Z/X-unitary involution σ_{φ} on the \mathscr{O}_Z -algebra $\mathscr{E}nd_{\mathscr{A}}(\mathscr{A} \oplus \mathscr{P})$, hence a Z/X-unitary involutive Brauer class.

Lemma 4.6. We have an equality of classes $[\mathscr{A}, \mathscr{P}, \varphi] = \Psi_{Z/X}[\mathscr{E}nd_{\mathscr{A}}(\mathscr{A} \oplus \mathscr{P}), \sigma_{\varphi}]$ in $H^2_{\text{\acute{e}t}}(X, \mathbf{R}^1_{Z/X}\mathbf{G}_m)$.

Proof. A Z/X-hermitian variant of the homomorphism H from the proof of Theorem 2.11 gives rise to $\mathbf{R}_{Z/X}\mathbf{GL}_n \to \mathbf{U}_{n,n}$, which fits into a morphism of short exact sequences

of sheaves of groups on $X_{\text{\acute{e}t}}$, where $\mathbf{U}_{n,n}$ denotes the unitary group of the rank 2n hyperbolic Z/Xhermitian form. The map $H^1_{\text{\acute{e}t}}(X, \mathbf{R}^1_{Z/X}\mathbf{GL}_n/\mathbf{R}^1_{Z/X}\mathbf{G}_m) \to H^1_{\text{\acute{e}t}}(X, \mathbf{PU}_{n,n})$ has the following interpretation: the class associated to a Z/X-torsion datum $(\mathscr{A}, \mathscr{P}, \varphi)$ of rank n is mapped to the class
of the \mathscr{O}_Z -algebra with Z/X-unitary involution $(\mathscr{E}nd(\mathscr{A} \oplus \mathscr{P}), \sigma_{\varphi})$. This is stated, for example, in
[69, VII Exer. 7]. A comparison of the coboundary maps finishes the proof.

Remark 4.7. Any 2-torsion datum $(\mathscr{A}_0, \mathscr{P}_0, \varphi_0)$ on X defines a Z/X-torsion datum $(f^*\mathscr{A}_0, \mathscr{P}_0, \varphi_0)$ with respect to an isomorphism $N_{Z/X}(f^*\mathscr{A}_0) \cong \mathscr{A}_0 \otimes \mathscr{A}_0$. Conversely, it is *not* true that a Z/X-torsion datum $(\mathscr{A}, \mathscr{P}, \varphi)$ with $\mathscr{A} \cong f^*\mathscr{A}_0$ is equivalent to a 2-torsion datum on X, see Knus–Parimala– Srinivas [67].

4.3. Clifford data. Axiomatizing the properties of the even Clifford algebra and Clifford bimodule of a line bundle-valued quadratic form lead to the notion of a *Clifford datum*, a mixture of having an involution and a torsion datum.

Definition 4.8. Let n = 2m be even. A *Clifford datum* of rank *n* consists of a tuple $(Z/X, \mathscr{A}, \sigma, \mathscr{P}, \varphi)$, where:

- (if m is odd) $Z \to X$ is étale quadratic, (\mathscr{A}, σ) is an Azumaya \mathscr{O}_Z -algebra of degree 2^{m-1} with Z/X-unitary involution, and $(\mathscr{A}, \mathscr{P}, \varphi)$ is a 2-torsion datum of degree 2^{m-1} on Z
- (if m is even) $Z \to X$ is étale quadratic, (\mathscr{A}, σ) is an Azumaya \mathscr{O}_Z -algebra of degree 2^{m-1} with involution of the first kind of type $(-1)^{m/2}$, and $(\mathscr{A}, \mathscr{P}, \varphi)$ is a Z/X-torsion datum of degree 2^{m-1} .

A morphism of Clifford data of rank n = 2m is a tuple $(g, \psi, \beta) : (Z/X, \mathscr{A}, \sigma, \mathscr{P}, \varphi) \to (Z'/X, \mathscr{A}', \sigma', \mathscr{P}', \varphi')$, where $g : Z \to Z'$ is an X-isomorphism, $\psi : (\mathscr{A}, \sigma) \to g^*(\mathscr{A}', \sigma')$ is an \mathscr{O}_Z -algebra isomorphism preserving the involutions, and $(\psi, \beta) : (\mathscr{A}, \mathscr{P}, \varphi) \to g^*(\mathscr{A}', \mathscr{P}', \varphi')$ is a morphism of torsion data.

Given a regular line bundle-valued quadratic form $(\mathscr{E}, q, \mathscr{L})$ of rank n = 2m and Arf covering $f: \mathbb{Z} \to X$, we construct an associated Clifford datum $(\mathbb{Z}/X, \widetilde{\mathscr{C}}_0, \tau_0, \widetilde{\mathscr{B}}_1, \nu)$ of rank n as follows:

For m odd, by Proposition 3.11, $(\widehat{\mathscr{C}}_0, \tau_0)$ is an Azumaya \mathscr{O}_Z -algebra with Z/X-unitary Clifford involution; let $\widetilde{\mathscr{B}}_1 = \widetilde{\mathscr{C}}_1$, then by the functorial properties in §4.1, $\widetilde{\mathscr{B}}_1$ is an invertible $\widetilde{\mathscr{C}}_0$ -bimodule (in particular is locally free of rank 2^{n-2}) with an \mathscr{O}_Z -module automorphism $\tau_1 : \widetilde{\mathscr{B}}_1 \to \widetilde{\mathscr{B}}_1$. The $\widetilde{\mathscr{C}}_0$ -bimodule structure on $\widetilde{\mathscr{B}}_1$ yields a canonical map

$$\begin{split} \nu : \widetilde{\mathscr{C}_0} \otimes_{\mathscr{O}_Z} \widetilde{\mathscr{C}_0} &\to \mathscr{E}nd_{\mathscr{O}_Z}(\widetilde{\mathscr{B}}_1) \\ a \otimes b \mapsto x \mapsto \tau_1(b\,\tau_1(ax)) = ``a\,x\,\tau_0(b)" \end{split}$$

which is an \mathscr{O}_Z -algebra isomorphism (since its a nontrivial homomorphism between Azumaya algebras of the same degree). Thus $(Z/X, \widetilde{\mathscr{C}}_0, \tau_0, \widetilde{\mathscr{B}}_1, \nu)$ is a 2-torsion datum on Z of degree n.

For *m* even, by Proposition 3.11, $(\widetilde{\mathscr{C}}_0, \tau_0)$ is an Azumaya \mathscr{O}_Z -algebra with Clifford involution of the first kind of type $(-1)^{m/2}$; by the functorial properties in §4.1, $\widetilde{\mathscr{C}}_1$ is an invertible $\widetilde{\mathscr{C}}_0$ -bimodule (in particular is locally free of rank 2^{n-2}) with an (ι -semilinear) \mathscr{O}_Z -module automorphism $\tau_1 : \widetilde{\mathscr{C}}_1 \to \iota_* \widetilde{\mathscr{C}}_1$. The $\widetilde{\mathscr{C}}_0$ -bimodule structure on $\widetilde{\mathscr{C}}_1$ yields a canonical map

$$\widetilde{\nu}: \widetilde{\mathscr{C}_0} \otimes_{\mathscr{O}_Z} \iota_* \widetilde{\mathscr{C}_0} \to \mathscr{E}nd_{\mathscr{O}_Z}(\widetilde{\mathscr{C}_1})$$
$$a \otimes \iota_* b \mapsto x \mapsto \tau_1(\iota_* b \, \tau_1(ax)) = ``a \, x \, \tau_0(b)"$$

which is an \mathcal{O}_Z -algebra isomorphism (since its a nontrivial homomorphism between Azumaya algebras of the same degree). The following compatibility condition

$$\iota_*\nu\circ s^{\iota}_{\widetilde{\mathscr{C}}_0}=\mathrm{Ad}^{\iota}_{\tau_1}\circ\widetilde{\nu}$$

holds, ensuring that $\tilde{\nu}$ descends to an \mathscr{O}_X -algebra isomorphism

$$\nu: N_{Z/X}(\mathscr{C}_0) \to \mathscr{E}nd_{\mathscr{O}_X}(\mathscr{Sym}(\mathscr{C}_1, \tau_1))$$

where $\mathscr{Sym}(\widetilde{\mathscr{C}}_1, \tau_1)$ is the descent of the \mathscr{O}_Z -submodule of $\widetilde{\mathscr{C}}_1$ of elements fixed by τ_1 , see [68, V Lemma 4.2.4] (and its preceding discussion) or (the Claim contained in the proof of) [69, II Thm. 9.12]. Thus, letting $\widetilde{\mathscr{B}}_1 = \mathscr{Sym}(\widetilde{\mathscr{C}}_1, \tau_1)$, the tuple $(Z/X, \widetilde{\mathscr{C}}_0, \tau_0, \widetilde{\mathscr{B}}_1, \nu)$ is a Z/X-torsion datum of degree n.

4.4. Clifford invariants and Clifford data. In this section, we provide some general interpretations of the similarity Clifford invariant in terms of Clifford data. Our main result of this section shows that the similarity Clifford invariant naturally captures the "torsion data" part of the Clifford data of a regular line bundle-valued quadratic form, while the results of §3.4 show that (the refinement of) the Tits algebra naturally captures the "involution" part. Recall from §4.2 that a torsion datum $(\mathscr{A}, \mathscr{P}, \varphi)$ has an associated involutive class $[\mathscr{A}, \mathscr{P}, \varphi]$.

Theorem 4.9. Let X be a scheme with 2 invertible and $(\mathscr{E}, q, \mathscr{L})$ a regular line bundle-valued quadratic form of even rank n = 2m and with Arf cover $f : Z \to X$. We have:

- a) if m is odd then $gc(\mathscr{E}, q, \mathscr{L}) = [\widetilde{\mathscr{C}}_0, \widetilde{\mathscr{B}}_1, \nu]$ in $H^2_{\text{\'et}}(X, \kappa_n^Z) \cong H^2_{\text{\'et}}(Z, \mu_2)$
- b) if m is even then $i^2gc(\mathscr{E}, q, \mathscr{L}) = [\widetilde{\mathscr{C}}_0, \widetilde{\mathscr{B}}_1, \nu]$ in $H^2_{\text{\'et}}(X, \mathbf{R}^1_{Z/X}\mathbf{G}_m)$

where $i: \kappa_n^Z \cong \mathbf{R}^1_{Z/X} \boldsymbol{\mu}_4 \to \mathbf{R}^1_{Z/X} \mathbf{G}_m$.

Proof. We have a graded isomorphism of (full) Clifford \mathcal{O}_X -algebras

$$\mathscr{C}(h_n^Z) \cong \mathscr{C}(h_{n-2}) \widehat{\otimes} \mathscr{C}(h_2^Z) \cong \mathscr{M}_{2^{m-1}}(\mathscr{O}_X) \widehat{\otimes} \operatorname{\mathscr{E}nd}_{\mathscr{O}_X}(f_* \mathscr{O}_Z)$$

see [68, IV Prop. 2.11, V §2.3], in particular, we have an \mathscr{O}_Z -algebra isomorphism $\widetilde{\mathscr{C}}_0(h_n^Z) \cong \mathscr{M}_{2^{m-1}}(\mathscr{O}_Z)$. After pulling back to Z, a choice of half-spin representation

$$\rho^+: f^{-1}\Gamma_n^Z \cong \Gamma_{m,m} \to \mathscr{E}nd_{\mathscr{O}_Z}(f^*\widetilde{\mathscr{C}}_0(h_n^Z)) \cong \mathbf{GL}_{2^{m-1}}$$

induces, by the universal property of Weil restriction, a homomorphism $\rho: \Gamma_n^Z \to \mathbf{R}_{Z/X}\mathbf{GL}_{2^{m-1}}$.

For m odd, ρ fits into a commutative diagram with exact rows (but not exact columns)

on $X_{\mathrm{\acute{e}t}}.$ The interpretation of the right vertical maps on cohomology

$$H^1_{\text{\'et}}(X, \mathbf{GSO}_n^Z) \to H^1_{\text{\'et}}(Z, \mathbf{GL}_{2^{m-1}}/\mu_2) \to H^1_{\text{\'et}}(Z, \mathbf{PO}_{2^{m-1}, 2^{m-1}})$$

is as follows: the class of a regular line bundle-valued quadratic form $(\mathscr{E}, q, \mathscr{L})$ of rank n and Arf invariant [Z/X] maps to the 2-torsion datum $(\widetilde{\mathscr{E}}_0, \widetilde{\mathscr{B}}_1, \nu)$ on Z; a 2-torsion datum $(\mathscr{A}, \mathscr{P}, \varphi)$ of degree n on Z maps to the involutive Brauer class of $(\mathscr{E}nd_{\mathscr{A}}(\mathscr{A} \oplus \mathscr{P}), \sigma_{\varphi})$. Finally, chasing the class of $(\mathscr{E}, q, \mathscr{L})$ around the diagram of coboundary maps yields the statement in a).

Similarly for m even, ρ fits into a commutative diagram with exact rows (but not exact columns)



on $X_{\text{ét}}$. The interpretation of the right vertical maps on cohomology

$$H^{1}_{\text{\acute{e}t}}(X, \mathbf{GSO}_{n}^{Z}) \to H^{1}_{\text{\acute{e}t}}(X, \mathbf{R}_{Z/X}\mathbf{GL}_{2^{m-1}}/\mathbf{R}_{Z/X}^{1}\mathbf{G}_{m}) \to H^{1}_{\text{\acute{e}t}}(X, \mathbf{PU}_{2^{m-1}, 2^{m-1}})$$

is as follows: the class of a regular line bundle-valued quadratic form $(\mathscr{E}, q, \mathscr{L})$ of rank n and Arf invariant [Z/X] maps to the Z/X-torsion datum $(\widetilde{\mathscr{C}}_0, \widetilde{\mathscr{B}}_1, \nu)$; a Z/X-torsion datum $(\mathscr{A}, \mathscr{P}, \varphi)$ of degree n maps to the Z/X-unitary involutive Brauer class of $(\mathscr{E}nd_{\mathscr{A}}(\mathscr{A} \oplus \mathscr{P}), \sigma_{\varphi})$. Finally, chasing the class of $(\mathscr{E}, q, \mathscr{L})$ around the diagram of coboundary maps yields the statement in b). \Box

In the case of where *m* is even, we can further describe the similarity Clifford invariant. We will first need a presentation of the group $H^2_{\text{\acute{e}t}}(X, \mathbf{R}^1_{Z/X}\boldsymbol{\mu}_4)$ analogous to the involutive Brauer group(s). For such a description when X is the spectrum of a field, see [69, VII Prop. 30.13] or Colliot-Thélène–Gille-Parimala [28, Prop. 2.10]. Given an Azumaya \mathcal{O}_X -algebra (\mathcal{B}, σ) with orthogonal involution, the Azumaya \mathcal{O}_Z -algebra $f^*\mathcal{B} = \mathcal{O}_Z \otimes_{f^{-1}\mathcal{O}_X} f^*\mathcal{A}$ has a natural Z/X-unitary involution $\sigma_{Z/X} = \iota^{\sharp} \otimes \sigma$ (see §3.2).

Proposition 4.10. Let X be a scheme with 2 invertible and $f: Z \to X$ étale quadratic. Denote by $\operatorname{Br}_4^*(Z/X)$ the abelian group of pairs $([\mathscr{A}, \tau], [\mathscr{B}, \sigma]) \in \operatorname{Br}^*(Z/X) \times \operatorname{Br}^*(X)$ such that $[\mathscr{A}, \tau]^{\otimes 2} = [f^*\mathscr{B}, \sigma_{Z/X}]$ Then there is a canonical homomorphism

$$\Psi_{Z/X,4}: \operatorname{Br}_4^*(Z/X) \to H^2_{\operatorname{\acute{e}t}}(X, \mathbf{R}^1_{Z/X}\boldsymbol{\mu}_4)$$

which fits into a commutative diagram with exact rows

$$\begin{array}{cccc} H^{1}_{\mathrm{\acute{e}t}}(X, \mathbf{R}^{1}_{Z/X} \mathbf{G}_{m}) & \longrightarrow & \mathrm{Br}^{*}_{4}(Z/X) & \longrightarrow & \mathrm{Br}^{*}(Z/X) & \xrightarrow{4} & \mathrm{Br}^{*}(Z/X) \\ & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ H^{1}_{\mathrm{\acute{e}t}}(X, \mathbf{R}^{1}_{Z/X} \mathbf{G}_{m}) & \longrightarrow & H^{2}_{\mathrm{\acute{e}t}}(X, \mathbf{R}^{1}_{Z/X} \boldsymbol{\mu}_{4}) & \longrightarrow & H^{2}_{\mathrm{\acute{e}t}}(X, \mathbf{R}^{1}_{Z/X} \mathbf{G}_{m}) & \xrightarrow{4} & H^{2}_{\mathrm{\acute{e}t}}(X, \mathbf{R}^{1}_{Z/X} \mathbf{G}_{m}) & \xrightarrow{4} & \mathrm{Br}^{*}(Z/X) \\ & & & & & & & \\ H^{1}_{\mathrm{\acute{e}t}}(X, \mathbf{R}^{1}_{Z/X} \mathbf{G}_{m}) & \longrightarrow & H^{2}_{\mathrm{\acute{e}t}}(X, \mathbf{R}^{1}_{Z/X} \mathbf{G}_{m}) & \xrightarrow{4} & H^{2}_{\mathrm{\acute{e}t}}(X, \mathbf{R}^{1}_{Z/X} \mathbf{G}_{m}) & \xrightarrow{4} & \mathrm{Br}^{*}(Z/X) \\ & & & & & & \\ \end{array}$$

of abelian groups. Finally, $\Psi_{Z/X,4}$ is surjective if and only if every 4-torsion element of ker $(N_{Z/X} : H^2_{\text{ét}}(Z, \mathbf{G}_m) \to H^2_{\text{ét}}(X, \mathbf{G}_m))$ is represented by an Azumaya \mathscr{O}_Z -algebra and every 2-torsion element of $H^2_{\text{ét}}(X, \mathbf{G}_m)$ is represented by an Azumaya \mathscr{O}_X -algebra.

Proof. Applying étale cohomology to the commutative diagram with exact rows

yields a commutative a diagram of abelian groups with exact rows

of which the central square is cartesian by a simple diagram chase. Composing the projections from $\operatorname{Br}_4^*(Z/X)$ with the maps Ψ and $\Psi_{Z/X}$ (from Theorem 3.9), the universal property of cartesian squares provides the homomorphism $\Psi_{Z/X,4}$. Moreover, since $\operatorname{Br}_4^*(Z/X)$ is defined as a fiber product, the maps Ψ , $\Psi_{Z/X}$, and $\Psi_{Z/X,4}$ define a morphism of cartesian squares. In particular, this extends to the sought after commutative diagram. The final statement of the Proposition follows from considerations similar to Remark 3.10.

Remark 4.11. Similar to Remark 3.10, if we consider the fiber $\operatorname{Br}_4^*(Z/X)^+$ of $\operatorname{Br}_4^*(Z/X)$ over $\operatorname{Br}^+(X)$ via the projection, then the restriction $\Psi_{Z/X,4}^+$ is an isomorphism whenever the conditions stated at the end of Proposition 4.10 hold.

By Theorem 3.9*a*), $\Psi[\mathscr{E}nd(\mathscr{E}), \sigma_b] = c_1(\mathscr{L}, \mu_2) \in H^2_{\text{\'et}}(X, \mu_2)$ for any regular \mathscr{L} -valued ϵ -symmetric bilinear form. By abuse of notation, we will write $c_1(\mathscr{L}, \mu_2) \in \text{Br}^*(X)$.

Theorem 4.12. Let X be a scheme with 2 invertible and $(\mathscr{E}, q, \mathscr{L})$ a regular line bundle-valued quadratic form of rank $n = 2m \equiv 0 \mod 4$ and Arf cover $f : Z \to X$. Then

$$gc(\mathscr{E}, q, \mathscr{L}) = \Psi_{Z/X, 4} \big([\mathscr{C}_0, \mathscr{B}_1, \nu], c_1(\mathscr{L}, \boldsymbol{\mu}_2) \big)$$

in $H^2_{\text{\'et}}(X, \kappa_4^Z) \cong H^2_{\text{\'et}}(X, \mathbf{R}^1_{Z/X}\boldsymbol{\mu}_4).$

Proof. This follows by a combination of Theorem 4.9b), Theorem 2.10a), and the construction of $\operatorname{Br}_4^*(Z/X)$ as a cartesian square in Proposition 4.10.

CLIFFORD INVARIANTS

5. Low dimensional exceptional isomorphisms

In the context of quadratic forms over schemes, *low rank* usually means of rank ≤ 6 . In this interval, the isomorphisms of Dynkin diagrams $A_1 = B_1 = C_1$, $D_2 = A_1^2$, $B_2 = C_2$, and $A_3 = D_3$, have beautiful reverberations in the theory of quadratic forms of rank 3, 4, 5, and 6, respectively. Over rings, these isomorphisms were deeply investigated by Knus, Ojanguren, Parimala, Paques, and Sridharan in the 1980s and 1990s. Now, a standard reference on this work is Knus [68, Ch. V]. Over fields, a wonderful reference is [69, IV §15]. Over general schemes, much of the theory over rings can be globalized, but a unified treatment did not yet exist in the literature until Auel [4].

As usual, let X be a noetherian separated scheme with 2 invertible and let \mathscr{L} be a fixed invertible \mathscr{O}_X -module.

5.1. Norm forms. A normed algebra or composition algebra (\mathscr{A}, n) on X is a locally free \mathscr{O}_X -algebra \mathscr{A} of finite rank together with a multiplicative quadratic form $n : \mathscr{A} \to \mathscr{O}_X$. A globalization of Hurwitz's theorem (see Petersson [86, Prop. 1.7d]) states that if the norm n is regular (as a quadratic form) then \mathscr{A} is either \mathscr{O}_X , an étale quadratic \mathscr{O}_X -algebra, or an Azumaya quaternion \mathscr{O}_X -algebra (or a generalized octonion algebra if \mathscr{A} is not assumed to be associative). An involution σ of the first kind on an \mathscr{O}_X -algebra \mathscr{A} is called a standard involution if the multiplicative map

$$n_{\sigma}: \mathscr{A} \xrightarrow{\Delta} \mathscr{A} \times \mathscr{A} \xrightarrow{\operatorname{id} \times \sigma} \mathscr{A} \times \mathscr{A}^{\operatorname{op}} \xrightarrow{m} \mathscr{A}$$

has image in the identity subalgebra $\mathscr{O}_X \hookrightarrow \mathscr{A}$. In this case, $(\mathscr{A}, n_{\sigma})$ is a normed algebra. For results about the existence and uniqueness of standard involutions in the affine setting, see Voight [95].

For a fixed normed algebra (\mathscr{A}, n) , a norm form of type $(\mathscr{A}, n_{\mathscr{A}})$ is a tuple $(\mathscr{P}, q, \mathscr{M})$ consisting of a right \mathscr{A} -module \mathscr{P} , an \mathscr{O}_X -module \mathscr{M} , and an \mathscr{M} -valued quadratic form $q : \mathscr{P} \to \mathscr{M}$ that is *n*-semilinear for the action of \mathscr{A} on \mathscr{P} , i.e. there's a commutative diagram of maps



or equivalently, that q(xa) = q(x) n(a) locally on sections. A morphism $(\psi, \lambda) : (\mathscr{P}, q, \mathscr{M}) \to (\mathscr{P}', q', \mathscr{M}')$ between norm forms of type (\mathscr{A}, σ) consists of \mathscr{O}_X -module morphisms $\psi : \mathscr{P} \to \mathscr{P}'$ and $\lambda : \mathscr{M} \to \mathscr{M}'$ such that $q' \circ \psi = \lambda \circ q$.

Theorem 5.1. Let (\mathscr{A}, n) be a normed algebra. Then for any right \mathscr{A} -module \mathscr{P} , there exists an \mathscr{O}_X -module $\mathscr{N}(\mathscr{P})$ and a norm form $(\mathscr{P}, n_{\mathscr{P}}, \mathscr{N}(\mathscr{P}))$, universal for the property that given any norm form $(\mathscr{P}, q, \mathscr{M})$ there exists an \mathscr{O}_X -module morphism $\psi : \mathscr{N}(\mathscr{P}) \to \mathscr{M}$ such that $q = \psi \circ n_{\mathscr{P}}$. Furthermore:

- a) if $\mathscr{P} = \mathscr{A}$ with the standard right action, then then there's a canonical isomorphism $(\mathscr{P}, n_{\mathscr{P}}, \mathscr{N}(\mathscr{P})) \cong (\mathscr{A}, n, \mathscr{O}_X),$
- b) the construction of the triple $(\mathscr{P}, n_{\mathscr{P}}, \mathscr{N}(\mathscr{P}))$ commutes with arbitrary base change,
- c) if \mathscr{P} is an invertible \mathscr{A} -module then $\mathscr{N}(\mathscr{P})$ is an invertible \mathscr{O}_X -module.

Proof. Bichsel [18, 3.1] and Knus [68, III.7] give a tensorial construction in the affine setting, which can be globalized. Knus–Ojanguren–Sridharan [62] give a construction by faithfully flat descent in the affine setting, which can also be globalized. \Box

5.2. Forms of rank 2. In this section, we will review the classification of regular line bundle-valued quadratic forms of rank 2 in terms of norm forms associated to quadratic normed algebras and relate the Clifford invariant to this classification.

Any Clifford datum of rank 2 is isomorphic to $(Z/X, \mathscr{O}_Z, \iota^{\sharp}, \mathscr{P}, <1>)$, where $f : Z \to X$ is étale quadratic, \mathscr{P} is an invertible \mathscr{O}_Z -module, and where we consider $<1>: \mathscr{O}_Z \otimes_{\mathscr{O}_Z} \mathscr{O}_Z \to \mathscr{O}_Z \cong$ $\mathscr{E}nd_{\mathscr{O}_Z}(\mathscr{P})$. We will denote this Clifford datum simply by $(Z/X, \mathscr{P})$.

Conversely, given $(Z/X, \mathscr{P})$, the norm form construction produces a unique similarity class of regular line bundle-valued quadratic form of rank 2. Indeed, the \mathscr{O}_X -algebra $f_*\mathscr{O}_Z$ has a unique standard involution induced from ι , with corresponding norm n_f . The locally free \mathscr{O}_X -module $f_*\mathscr{M}$ is an invertible $f_*\mathscr{O}_Z$ -module, and so defines a universal norm form $(f_*\mathscr{M}, n_{f_*\mathscr{M}}, \mathscr{N}(f_*\mathscr{M}))$ of type $(f_*\mathscr{O}_Z, n_f)$.

Remark 5.2. This recaptures (in the case of locally free \mathcal{O}_X -algebras of rank 2) a general "norm functor" construction due to Ferrand [36, §5.3].

Theorem 5.3 (Kneser [60, §6 Prop. 2], Ferrand [37, §5.4], Bichsel–Knus [17, §4.2]). Let X be a scheme with 2 invertible. The norm form and Clifford datum define inverse equivalence functors between the following categories:

- objects are Clifford data $(Z/X, \mathscr{P})$ of rank 2 and morphisms are isomorphisms of Clifford data
- objects are regular line bundle-valued quadratic forms $(\mathcal{E}, q, \mathcal{L})$ of rank 2 and morphisms are similarity transformations,

Under this equivalence, [Z/X] maps to $d_{\pm}(\mathscr{E}, q, \mathscr{L})$ under $H^1_{\text{\acute{e}t}}(X, \mathbb{Z}/2\mathbb{Z}) \to H^1_{\text{\acute{e}t}}(X, \mu_2)$ from Proposition 1.13.

Remark 5.4. The generalization of Theorem 5.3 to flat quadratic coverings and nondegenerate forms is established by Ferrand [37] over an arbitrary scheme with 2 invertible.

The norm for associated to $(Z/X, \mathscr{O}_Z)$ is our standard form h_2^Z and there's a canonical isomorphism of short exact sequences

given by the right multiplication map $\rho : \mathbf{R}_{Z/X} \mathbf{G}_{\mathbf{m}} \to \mathbf{GSO}_2^Z$. Then since h_2^Z is of type $(f_* \mathscr{O}_Z, n_f)$, we verify that $n_{\mathscr{O}_Z}(\rho_z(v)) = n_{\mathscr{O}_Z}(v \cdot z) = n_{\mathscr{O}_Z}(v)n_f(z)$, so that indeed ρ_z is a similarity transformation with multiplier $n_f(z)$, and the diagram commutes. The fact that ρ is an isomorphism may be checked locally, the affine case being verified in [17, §6.1].

Also, there's a canonical isomorphism of short exact sequences

which follows from the above, and from [68, V Lemma 2.5.2].

Proposition 5.5. Let X be a scheme with 2 invertible, $f : Z \to X$ étale quadratic, \mathscr{P} and invertible \mathscr{O}_Z -module, and $(\mathscr{E}, q, \mathscr{L})$ the corresponding norm form. Then we have

$$gc(\mathscr{E},q,\mathscr{L}) = c_1(\mathscr{P},\boldsymbol{\mu}_2)$$

in $H^2_{\text{\acute{e}t}}(X, \kappa_2^Z) \cong H^2_{\text{\acute{e}t}}(Z, \mu_2).$

Proof. This follows from Theorem 4.9, or directly by chasing a class associated to a regular line bundle-valued quadratic form $(\mathscr{E}, q, \mathscr{L})$ around the commutative diagram of coboundary maps induced from diagram 38.

Note that Theorem 2.10b is then a consequence of the isomorphism of exact sequences



and Proposition 5.5. Using the classification of regular line bundle-valued forms of rank 2, we can immediately give a generalization of [68, V Ex. 9.2.3].

Corollary 5.6. If a regular quadratic form $(\mathscr{E}, q, \mathscr{L})$ of rank 2 has trivial signed discriminant, then it's similar to an \mathscr{L} -valued hyperbolic form $H_{\mathscr{L}}(\mathscr{V})$ for an invertible \mathscr{O}_X -module \mathscr{V} . Furthermore, gc(q) is trivial if and only if \mathscr{L} and \mathscr{V} are both squares in $\operatorname{Pic}(X)$.

Proof. If $(\mathscr{E}, q, \mathscr{L})$ has trivial discriminant, then by Theorem 5.3, it's a norm form $(Z/X, \mathscr{P})$ with Z/X split and $\mathscr{P} \cong (\mathscr{P}_1, \mathscr{P}_2)$ in $\operatorname{Pic}(Z) \cong \operatorname{Pic}(X) \times \operatorname{Pic}(X)$. Then $\mathscr{E} \cong f_* \mathscr{P} \cong \mathscr{P}_1 \oplus \mathscr{P}_2$ and $\mathscr{L} \cong \mathscr{N}_f(\mathscr{P}) \cong \det(f_* \mathscr{P}) \cong \mathscr{P}_1 \otimes \mathscr{P}_2$. It's then easy to verify, using Remark 5.2, that $(\mathscr{E}, q, \mathscr{L})$ is similar to the form $(\mathscr{P}_1 \oplus \mathscr{P}_2, \otimes, \mathscr{P}_1 \otimes \mathscr{P}_2)$, which in turn is similar to the hyperbolic form $H_{\mathscr{P}_1 \otimes \mathscr{P}_2}(\mathscr{P}_1)$. This proves the first claim, which is a generalization of [68, IV.9.2.3 Ex.].

By Proposition 5.5, gc(q) is trivial if and only \mathscr{P} is a square in Pic(Z) if and only if \mathscr{P}_1 and \mathscr{P}_2 are squares in Pic(X). But this is also equivalent to both $\mathscr{V} \cong \mathscr{P}_1$ and $\mathscr{L} \cong \mathscr{P}_1 \otimes \mathscr{P}_2$ being squares.

5.3. Forms of rank 4. In this section, we will review the classification of regular line bundle-valued quadratic forms of rank 4 in terms of norm forms associated to Azumaya quaternion algebras and relate the Clifford invariant to this classification.

Any symplectic involution on an Azumaya quaternion \mathscr{O}_Z -algebra \mathscr{A} is isomorphic to the standard involution $\sigma_{\mathscr{A}}$ (this follows from Pumplün [87, Theorem 2.6], for example), and thus any Clifford datum of rank 4 is isomorphic to $(Z/X, \mathscr{A}, \sigma_{\mathscr{A}}, \mathscr{P}, \varphi)$, which we will denote simply by $(Z/X, \mathscr{A}, \mathscr{P}, \varphi)$. Thus a Clifford datum of rank 4 is equivalent to a 2-torsion datum on Z.

Every regular line bundle-valued quadratic form gives rise to a Clifford datum of rank 4. Conversely, a "twisted" version of the universal norm form functor can reconstruct the similarity class of a quadratic form of rank 4 from its associated Clifford datum. This functor is essentially described in [65, §10] or [68, V §4.2] and can be viewed as a refinement of [69, IV.15.B]. For a precise definition of the twisted norm form, see Auel [4].

Theorem 5.7. Let X be a scheme with 2 invertible. The twisted norm form and Clifford datum define inverse equivalence functors between the following categories:

- objects are Clifford data (Z/X, A, P, φ) of rank 4 and morphisms are isomorphisms of Clifford data
- objects are regular line bundle-valued quadratic forms $(\mathcal{E}, q, \mathcal{L})$ of rank 4 and morphisms are similarity transformations.

Under this equivalence, [Z/X] is mapped to $d_{\pm}(\mathscr{E}, q, \mathscr{L})$ under $H^1_{\text{\acute{e}t}}(X, \mathbb{Z}/2\mathbb{Z}) \cong H^1_{\text{\acute{e}t}}(X, \mu_2)$ from Proposition 1.13.

To compute the Clifford invariant in terms of the classification in Theorem 5.7, we use the following torsorial reinterpretation. Recall the notation $h_4^Z = H_{\mathscr{O}_X}(\mathscr{O}_X) \perp (f_*\mathscr{O}_Z, h^Z)$. There's a canonical isomorphism $\mathbf{Spin}_4^Z \cong \mathbf{R}_{Z/X}\mathbf{SL}_1(\widetilde{\mathscr{O}_0}(h_4^Z))$ by [68, V.4.4.1] (see also [69, IV Prop. 15.10]), which generalizes immediately from the affine case. In particular, we have an isomorphism $\mathbf{Spin}_4^Z \cong \mathbf{R}_{Z/X}\mathbf{SL}_2$ for any choice of \mathscr{O}_Z -algebra isomorphism $\widetilde{\mathscr{O}_0} \cong M_2(\mathscr{O}_Z)$ (a canonical choice is furnished from the graded \mathscr{O}_X -algebra isomorphism $\mathscr{C}(h_2^Z) \cong \mathscr{C}(h_2) \widehat{\otimes} \mathscr{C}(h_4^Z)$ of (full) Clifford algebras).

The Weil restriction functor induces the following exact sequence of groups schemes

$$1 \to \mathbf{R}_{Z/X} \mathbf{SL}_2 \to \mathbf{R}_{Z/X} \mathbf{GL}_2 \xrightarrow{\mathbf{R}_{Z/X} \det} \mathbf{R}_{Z/X} \mathbf{G}_{\mathrm{m}} \to 1$$

on $X_{\text{\acute{e}t}}$. Set $\mathbf{G} = \mathbf{R}_{Z/X} \mathbf{G} \mathbf{L}_2 / \mathbf{R}_{Z/X}^1 \mathbf{G}_m$ (see §4.2) and denote by Γ the sheaf kernel of the composite epimorphism $\mathbf{R}_{Z/X} \mathbf{G} \mathbf{L}_2 \to \mathbf{R}_{Z/X} \mathbf{G}_m \xrightarrow{\mathrm{id}/\iota} \mathbf{R}_{Z/X}^1 \mathbf{G}_m$. Hence we have exact sequences

$$1 \to \mathbf{R}_{Z/X}^{1} \mathbf{G}_{\mathrm{m}} \to \mathbf{R}_{Z/X}^{1} \mathbf{G}_{\mathbf{L}_{2}} \to \mathbf{G} \to 1 \qquad 1 \to \mathbf{\Gamma} \to \mathbf{R}_{Z/X}^{1} \mathbf{G}_{\mathbf{L}_{2}} \xrightarrow{\mathrm{id}/\iota} \mathbf{R}_{Z/X}^{1} \mathbf{G}_{\mathrm{m}} \to 1$$

of sheaves of groups on $X_{\text{\acute{e}t}}$.

Finally, restricting the quotient map $\mathbf{R}_{Z/X}\mathbf{GL}_2 \to \mathbf{G}$ to Γ yields a homomorphism $s: \Gamma \to \mathbf{G}$, restricting $\mathbf{R}_{Z/X}$ det to Γ yields a homomorphism $\Gamma \to \mathbf{G}_m$, and restricting $N_{Z/X} \circ \mathbf{R}_{Z/X}$ det to \mathbf{G} yields a homomorphism $\mathbf{G} \to \mathbf{G}_m$.

Theorem 5.8. There are canonical isomorphisms

$$\Gamma \cong \mathbf{S}\Gamma_4^Z, \qquad \mathbf{R}_{Z/X}\mathbf{S}\mathbf{L}_2 \cong \mathbf{Spin}_4^Z$$

and

$$\mathbf{G} \cong \mathbf{GSO}_4^Z, \qquad (\mathbf{R}_{Z/X}\mathbf{SL}_2)/\boldsymbol{\mu}_2 \cong \mathbf{SO}(h_{Z/X}^2)$$

making the even fundamental diagram (15) associated to the form h_4^Z isomorphic to



Proof. See Auel [4]. We have the following commutative diagram



of sheaves of groups on $X_{\text{\acute{e}t}}$. In particular, the morphism $\Gamma \to \mathbf{G}$ can be identified with the cokernel of the inclusion $\mathbf{R}^1_{Z/X} \mu_4 \to \Gamma$.

Proposition 5.9. Let X be a scheme with 2 invertible, $f : Z \to X$ étale quadratic, $(Z/X, \mathscr{A}, \mathscr{P}, \varphi)$ a Clifford datum of rank 4, and $(\mathscr{E}, q, \mathscr{L})$ the associated twisted norm form. Then

$$gc(\mathscr{E}, q, \mathscr{L}) = \Psi_{Z/X, 4}([\mathscr{A}, \mathscr{P}, \varphi], c_1(\mathscr{L}, \boldsymbol{\mu}_2))$$

in $H^2_{\text{\'et}}(X, \boldsymbol{\kappa}_4^Z) \cong H^2_{\text{\'et}}(X, \mathbf{R}^1_{Z/X} \boldsymbol{\mu}_4).$

Proof. This is a consequence of Theorem 4.12.

To every locally free \mathscr{O}_Z -module \mathscr{V} of rank 2, we can associate the *split* Clifford datum $(Z/X, \mathscr{E}nd(\mathscr{V}), \mathscr{V} \otimes \iota_*\mathscr{V}, \varphi_{\mathscr{V}})$ of rank 4, where $\varphi_{\mathscr{V}}$ is the composition of canonical \mathscr{O}_Z -algebra isomorphisms

$$\mathscr{E}nd(\mathscr{V}) \otimes_{\mathscr{O}_{Z}} \iota_{*} \mathscr{E}nd(\mathscr{V}) \xrightarrow{\operatorname{can}} \mathscr{E}nd(\mathscr{V}) \otimes_{\mathscr{O}_{Z}} \mathscr{E}nd(\iota_{*}\mathscr{V}) \to \mathscr{E}nd(\mathscr{V} \otimes \iota_{*}\mathscr{V}).$$

The map

$$H^1_{\text{\acute{e}t}}(Z, \mathbf{GL}_2) \cong H^1_{\text{\acute{e}t}}(X, \mathbf{R}_{Z/X}\mathbf{GL}_2) \to H^1_{\text{\acute{e}t}}(X, \mathbf{G}) \cong H^1_{\text{\acute{e}t}}(X, \mathbf{GSO}_4^Z)$$

has the following interpretation: a locally free \mathscr{O}_Z -module of rank 2 is sent to the twisted norm form of its split Clifford datum.

The twisted norm form of the split Clifford datum associated to \mathscr{V} is isometric to $N_{Z/X}(\mathscr{V}, \wedge, \det \mathscr{V}) = (N_{Z/X}(\mathscr{V}), N_{Z/X}(\wedge), N_{Z/X}(\det \mathscr{V}))$, which we can think of as the "bilinear form norm" applied the canonical skew-symmetric form $\wedge : \mathscr{V} \otimes \mathscr{V} \to \det \mathscr{V}$ on Z. When $f : Z \to X$ is split, then a locally free \mathscr{O}_Z -module of rank 2 can be though of as a pair of locally free \mathscr{O}_X -modules $(\mathscr{V}, \mathscr{W})$ of rank 2, and the twisted norm for of the associated Clifford datum is similar to the tensor product $\mathscr{V} \otimes \mathscr{W} \to \det \mathscr{V} \otimes \det \mathscr{W}$ of the canonical skew-symmetric wedging forms.

Corollary 5.10. With the notations of Proposition 5.9, the class $gc(\mathscr{E}, q, \mathscr{L})$ vanishes if and only if $(\mathscr{E}, q, \mathscr{L})$ is similar to $N_{Z/X}(\mathscr{V}, \wedge, \det \mathscr{V})$ for some locally free \mathscr{O}_Z -module of rank 2 such that $\det \mathscr{V} \cong \iota_* \det \mathscr{V}$.

Proof. This follows from the above description of twisted norm forms of split Clifford data of rank 4, along with a diagram chase involving the cohomology of diagram (39).

When $f: Z \to X$ is split, then $\mathbf{R}_{Z/X}\mathbf{GL}_2 = \mathbf{GL}_2 \times \mathbf{GL}_2$ and $\Gamma = \mathbf{GL}_2 \times_{\mathbf{G}_m} \mathbf{GL}_2$ (pairs of invertible transformations with equal determinant), making the even fundamental diagram is isomorphic to



compare with Knus [68, V §4.5–4.6]. Furthermore, Clifford data of rank 4 (with Z/X split) can be viewed as tuples $(\mathscr{A}, \mathscr{B}, \mathscr{P}, \varphi)$ of degree 2, where \mathscr{A} and \mathscr{B} are Azumaya quaternion \mathscr{O}_X -algebras, \mathscr{P} is a locally free \mathscr{O}_X -module of rank 4, and $\varphi : \mathscr{A} \otimes \mathscr{B} \to \mathscr{E}nd(\mathscr{P})$ is an \mathscr{O}_X -algebra isomorphism.

The category of such tuples are $(\mathbf{GL}_2 \times \mathbf{GL}_2)/\mathbf{G}_m$ -torsors on $X_{\text{\acute{e}t}}$. In this case, the twisted norm form is just the universal or reduced norm form, and is studied in [18], [65], [64], and [17].

5.4. Forms of rank 6. In this section, we will review the classification of regular line bundle-valued quadratic forms of rank 6 in terms of the pfaffian forms (or discriminant algebras) associated to Azumaya algebras of degree 4 relate the Clifford invariant to this classification.

Any regular line bundle-valued quadratic form of rank 6 determines a Clifford datum of rank 6. Conversely, a "twisted" version of the reduced pfaffian form functor can reconstruct the quadratic form from the associated Clifford datum. This functor is essentially described in [61], [65, §7], [66, §5], [68, V §5.5], and can be viewed as a refinement of [69, IV.15.D]. For a precise definition of the twisted pfaffian form, see Auel [4].

Theorem 5.11. Let X be a scheme with 2 invertible. The twisted pfaffian form and Clifford datum define inverse equivalence functors between the following categories:

- objects are Clifford data $(Z/X, \mathscr{A}, \sigma, \mathscr{P}, \varphi)$ of rank 6 and morphisms are isomorphisms of Clifford data
- objects are regular line bundle-valued quadratic forms $(\mathcal{E}, q, \mathcal{L})$ of rank 6 and morphisms are similarity transformations.

Under this equivalence, [Z/X] maps to $d_{\pm}(\mathscr{E}, q, \mathscr{L})$ under $H^{1}_{\text{\acute{e}t}}(X, \mathbb{Z}/2\mathbb{Z}) \cong H^{1}_{\text{\acute{e}t}}(X, \mu_{2})$ from Proposition 1.13.

To compute the Clifford invariant in terms of the classification in Theorem 5.11, we use the following torsorial reinterpretation. Recall the notation $h_6^Z = H_{\mathscr{O}_X}(\mathscr{O}_X^2) \perp (f_*\mathscr{O}_Z, h^Z)$. There are canonical isomorphisms $\operatorname{\mathbf{Spin}}_6^Z \cong \operatorname{\mathbf{SU}}_{2,2}$ and $\Gamma_6^Z \cong \operatorname{\mathbf{SGU}}_{2,2}$, by [68, V §5.6] (see also [69, IV Prop. 15.27]), where $\operatorname{\mathbf{GU}}_{2,2}$ is the general unitary group of the hyperbolic Z/X-hermitian form of rank 4.

Theorem 5.12. The above isomorphisms make the even fundamental diagram (15) associated to the form h_6^Z isomorphic to



Proposition 5.13. Let X be a scheme with 2 invertible, $f : Z \to X$ étale quadratic, $(Z/X, \mathscr{A}, \tau, \mathscr{P}, \varphi)$ a Clifford datum of rank 6, and $(\mathscr{E}, q, \mathscr{L})$ the associated twisted pfaffian form. Then

$$gc(\mathscr{E}, q, \mathscr{L}) = [\mathscr{A}, \mathscr{P}, \varphi]$$

 $in \ H^2_{\text{\'et}}(X, \boldsymbol{\kappa}_6^Z) \cong H^2_{\text{\'et}}(X, \mathbf{R}_{Z/X} \boldsymbol{\mu}_2) \cong H^2_{\text{\'et}}(Z, \boldsymbol{\mu}_2).$

Proof. This is a consequence of Theorem 4.9a).

To every regular line bundle-valued Z/X-hermitian form $(\mathcal{H}, h, \mathcal{L})$ of rank 4, we can associate a split Clifford datum $(Z/X, \mathscr{E}nd(\mathcal{H}), \tau_h, \mathcal{H} \otimes \mathcal{H}, \varphi_{\mathcal{H}})$ of rank 6, where $(\mathscr{E}nd(\mathcal{H}), \tau_h)$ is the associated Z/X-unitary adjoint involution (see §3.2), and $\varphi_{\mathcal{H}}$ is the canonical \mathcal{O}_Z -algebra isomorphism

$$\mathscr{E}nd(\mathscr{H})\otimes \mathscr{E}nd(\mathscr{H}) \to \mathscr{E}nd(\mathscr{H}\otimes \mathscr{H}).$$

The map

$$H^1_{\text{\'et}}(X, \mathbf{GU}_{2,2}) \to H^1_{\text{\'et}}(X, \mathbf{SGU}_{2,2}/\mathbf{R}_{Z/X}\boldsymbol{\mu}_2) \cong H^1_{\text{\'et}}(X, \mathbf{GSO}(h_6^Z))$$

has the following interpretation: a regular line bundle-valued Z/X-hermitian form is sent to the twisted pfaffian form of its split Clifford datum.

Corollary 5.14. With the notations of Proposition 5.13, the class $gc(\mathcal{E}, q, \mathcal{L})$ vanishes if and only if there exists a regular line bundle-valued Z/X-hermitian form $(\mathcal{H}, h, \mathcal{M})$ of rank 4 with trivial hermitian discriminant such that $f^*(\mathcal{E}, q, \mathcal{L})$ is similar (with the pullback Z/X-hermitian structure) to the canonical quadratic form $(\bigwedge^2 \mathcal{H}, \wedge, \det \mathcal{H})$.

When $f : Z \to X$ is split, then $\mathbf{GU}_{2,2} \cong \mathbf{G}_{\mathrm{m}} \times \mathbf{GL}_4$, $\mathbf{SGU}_{2,2} \cong \mathbf{G}_{\mathrm{m}} \times_{\mathbf{G}_{\mathrm{m}}} \mathbf{GL}_{2,2}$ (pairs (a, A) with det $A = a^2$), and $\mathbf{SU}_{2,2} \cong \mathbf{SL}_4$. We also have $\mathbf{SGU}_{2,2}/(\mu_2 \times \mu_2) \cong \mathbf{GL}_4/\mu_2$. Hence the even fundamental diagram is isomorphic to



and diagram (21), which defines the oriented invariants, is isomorphic to



In particular the oriented invariants gc^{\pm} are described as in Example 2.20. Furthermore, Clifford data of rank 6 (with Z/X split) can be viewed simply as 2-torsion data ($\mathscr{A}, \mathscr{P}, \varphi$) of degree 4 on X. In this case, the twisted pfaffian form is just the reduced pfaffian form, which is studied in [61], [65], and [17]. Actually, not only can the similarity class of a quadratic form be recovered from a Clifford datum, but also a choice of orientation.

Appendix

Lemma 5.15. Consider a commutative diagram with exact rows

$$1 \longrightarrow \kappa \longrightarrow \Gamma^{+} \longrightarrow G^{+} \longrightarrow 1$$
$$\| \qquad \downarrow \qquad \downarrow \qquad \downarrow$$
$$1 \longrightarrow \kappa \longrightarrow \Gamma \longrightarrow G \longrightarrow 1$$

of sheaves of groups on a site X. Suppose that $\kappa \to \Gamma^+$ is central and $\mathbf{G}^+ \to \mathbf{G}$ is a normal monomorphism. Then the coboundary map $H^1(X, \mathbf{G}^+) \to H^2(X, \kappa)$ is constant on the fibers of the map $H^1(X, \mathbf{G}^+) \to H^1(X, \mathbf{G})$.

Proof. If $g : \mathbf{G}^+ \to \mathbf{G}$ is a normal monomorphism, then so is $\Gamma^+ \to \Gamma$ and there's a natural homomorphism $\Gamma/\Gamma^+ \to \mathbf{G}/\mathbf{G}^+$ of sheaves of groups, which is an isomorphism by the Nine Lemma 5.16.

First we show that the coboundary map is constant on the kernel of g^1 . We have a commutative diagram

$$\begin{split} H^0(X, \mathbf{\Gamma}/\mathbf{\Gamma}^+) & \xrightarrow{\sim} H^0(X, \mathbf{G}/\mathbf{G}^+) \\ & \downarrow & \downarrow^{\delta_{\mathbf{G}}} \\ H^1(X, \mathbf{\Gamma}^+) & \longrightarrow H^1(\mathbf{G}^+) & \xrightarrow{\delta} H^2(\mathbf{k}) \\ & \downarrow & \downarrow^{g^1} & \parallel \\ H^1(X, \mathbf{\Gamma}) & \longrightarrow H^1(\mathbf{G}) & \longrightarrow H^2(\mathbf{k}) \end{split}$$

of pointed sets. In particular, $\ker(g^1)$ equals the image of $\delta_{\mathbf{G}}$. By the commutativity of the top square in the diagram, $\ker(g^1)$ can be lifted to $H^1(X, \Gamma^+)$, i.e. the coboundary map δ is trivial (hence constant) on $\ker(g^1)$.

Now, for each fixed $\xi \in H^1(X, \mathbf{G})$ we can reduce to the previous case, where ξ is the neutral element. Indeed, if ξ is not in the image of g^1 the statement of the lemma is vacuous, otherwise, let $\xi^+ \in H^1(X, \mathbf{G}^+)$ map to ξ . If $\xi^+_{ij} \in Z^1(\mathscr{U}, \mathbf{G}^+)$ is a Čech 1-cocycle representing ξ^+ for a cover \mathscr{U} of X, then $\xi_{ij} \in Z^1(\mathscr{U}, \mathbf{G})$, where $\xi_{ij} = g(\xi^+_{ij})$, is a Čech 1-cocycle representing ξ . Finally, twisting the exact sequences by ξ^+_{ij} and ξ_{ij} accomplishes the reduction.

Lemma 5.16 (Nine Lemma). Consider a commutative diagram



of sheaves of groups on a site X. If all columns and the two bottom rows are exact then the top row is exact; if all columns and the two top rows are exact then the bottom row is exact.

Lemma 5.17 (Roman IX Lemma). Consider a commutative diagram with exact rows and diagonals



of sheaves of groups on a site X. Assume that $\mu \to \mathbf{A}$ and $\mu \to \mathbf{C}$ are central. Then the induced diagram on cohomology

is commutative, except for the right hand pentagon, which is anticommutative.

Remark 5.18 (Roman IX rearrangement). Given a commutative diagram



of sheaves of groups on a site X, whose top two rows and left two columns are exact, then by the Nine Lemma 5.16, β and γ are isomorphisms, and there's a rearranged commutative diagram with exact rows and diagonals



where $f = \beta \circ \alpha$ and $g = \gamma \circ \delta$. Hence the Roman IX Lemma 5.17 applies.

Lemma 5.19. Let



be a commutative diagram of sheaves of groups on a site X with central and exact rows and central columns. Let δ and δ' be the second coboundary maps associated to the first and second rows, respectively. Then for each $\xi \in H^1(X, \mathbf{G})$ and $\xi' \in H^1(X, \mathbf{G}')$, we have

$$\delta'(\xi \cdot \xi') = \delta(\xi) + \delta'(\xi')$$

where \cdot and + represent the action of $H^1(X, \mathbf{G})$ on $H^1(X, \mathbf{G}')$ and of $H^2(X, \mathbf{k})$ on $H^2(X, \mathbf{k}')$, respectively.

Proof. Let $\mathscr{U} = \{U_i \to X\}_{i \in I}$ be a cover of X. Let $\xi_{ij} \in Z^1(\mathscr{U}, \mathbf{G})$ and $\xi'_{ij} \in Z^1(\mathscr{U}, \mathbf{G}')$ be 1-cocycles, and $\gamma_{ij} \in C^1(\mathscr{U}, \Gamma)$ and $\gamma'_{ij} \in C^1(\mathscr{U}, \Gamma')$ be 1-cochains lifting ξ_{ij} and ξ'_{ij} , respectively. Then we have

$$\delta'(\xi_{ij} \cdot \xi'_{ij}) = (\gamma_{ij\underline{k}} \cdot \gamma'_{ij\underline{k}}) (\gamma_{\underline{i}jk} \cdot \gamma'_{\underline{i}jk}) (\gamma_{\underline{i}j\underline{k}} \cdot \gamma'_{\underline{i}j\underline{k}})^{-1}$$
$$= \gamma_{ij\underline{k}}\gamma_{\underline{i}j\underline{k}}\gamma_{\underline{i}j\underline{k}}^{-1} + \gamma'_{ij\underline{k}}\gamma'_{\underline{i}j\underline{k}}\gamma'_{\underline{i}j\underline{k}}^{-1}$$
$$= \delta(\xi_{ij}) + \delta'(\xi'_{ij})$$

since $\Gamma \to \Gamma'$ is central. Also see Giraud [47, IV Cor. 3.3.4(ii)].

CLIFFORD INVARIANTS

References

- A. Adrian Albert, Structure of Algebras, American Mathematical Society Colloquium Publications, vol. 24, American Mathematical Society, New York, 1939.
- 2. Asher Auel, (Non)surjectivity of the Clifford invariant for curves over local fields, in preparation.
- 3. _____, Cohomological invariants of line bundle-valued symmetric bilinear forms, Ph.D. thesis, University of Pennsylvania, Philadelphia, 2009.
- 4. _____, Line bundle-valued quadratic forms of low rank, preprint, 2011.
- Thiruvalloor Eesanaipaadi Venkata Balaji, Line-bundle valued ternary quadratic forms over schemes, J. Pure Appl. Algebra 208 (2007), 237–259.
- 6. Paul Balmer, Derived Witt groups of a scheme, J. Pure Appl. Algebra 141 (1999), 101–129.
- 7. _____, Triangular Witt groups. I. The 12-term exact sequence, K-Theory 19 (2000), 311–363.
- 8. _____, Triangular Witt groups. II. From usual to derived, Math. Z. 236 (2001), 351–382.
- 9. Paul Balmer, Witt groups, Handbook of K-theory. Vol. 1, 2, Springer, Berlin, 2005, pp. 539–576.
- 10. Paul Balmer and Baptiste Calmès, Geometric description of the connecting homomorphism for Witt groups, preprint, 2007.
- Grégory Berhuy, Cohomological invariants of quaternionic skew-Hermitian forms, Arch. Math. (Basel) 88 (2007), no. 5, 434–447. MR 2316889 (2008d:12005)
- 12. Marcello Bernardara and Michele Bolognesi, *Derived categories and rationality of conic bundles*, preprint arXiv:1010.2417v3, 2010.
- Manjul Bhargava, Higher composition laws. I. A new view on Gauss composition, and quadratic generalizations, Ann. of Math. (2) 159 (2004), no. 1, 217–250.
- <u>—</u>, Higher composition laws. II. On cubic analogues of Gauss composition, Ann. of Math. (2) 159 (2004), no. 2, 865–886.
- <u>—</u>, Higher composition laws. III. The parametrization of quartic rings, Ann. of Math. (2) 159 (2004), no. 3, 1329–1360.
- 16. _____, The density of discriminants of quartic rings and fields, Ann. of Math. (2) 162 (2005), no. 2, 1031–1063.
- 17. W. Bichsel and M.-A. Knus, Quadratic forms with values in line bundles, Contemp. Math. 155 (1994), 293–306.
- 18. Walter Bichsel, Quadratische raeume mit werten in invertierbaren moduln, Ph.D. thesis, ETH Zürich, 1985.
- A. Bondal and D. Orlov, *Derived categories of coherent sheaves*, Proceedings of the International Congress of Mathematicians, Vol. II (Beijing, 2002) (Beijing), Higher Ed. Press, 2002, pp. 47–56.
- Alexei Bondal and Dmitri Orlov, Semiorthogonal decomposition for algebraic varieties, preprint arXiv:alggeom/9506012v1, 1995.
- C. J. Bushnell, Modular quadratic and Hermitian forms over Dedekind rings. I, J. Reine Angew. Math. 286/287 (1976), 169–186.
- Modular quadratic and Hermitian forms over Dedekind rings. II, J. Reine Angew. Math. 288 (1976), 24–36.
- Stefaan Caenepeel and Freddy van Oystaeyen, Quadratic forms with values in invertible modules, K-Theory 7 (1993), 23–40.
- Andrei Căldăraru, Derived categories of twisted sheaves on Calabi-Yau manifolds, Ph.D. thesis, Cornell University, Ithica, NY, May 2000.
- Baptiste Calmès and Jens Hornbostel, Witt motives, transfers, and dévissage, preprint, K-theory preprint archive, 2006.
- Baptiste Calmès and Jens Hornbostel, Push-forwards for Witt groups of schemes, preprint, arXiv:0806.0571v1 [math.AG], 2008.
- Philippe Cassou-Noguès, Boaz Erez, and Martin J. Taylor, Invariants of a quadratic form attached to a tame covering of schemes, J. Th. des Nombres de Bordeaux 12 (2000), 597–660.
- J.-L. Colliot-Thélène, P. Gille, and R. Parimala, Arithmetic of linear algebraic groups over 2-dimensional geometric fields, Duke Math. J. 121 (2004), no. 2, 285–341.
- 29. A.J. de Jong, A result of Gabber, preprint, 2003.
- Pierre Deligne, Cohomologie à supports propres, exposé XVII of SGA 4, Théorie des Topos et Cohomologie Etale des Schémas, Tome 3, Lecture Notes in Mathematics, vol. 305, Springer-Verlag, Berlin, 1973, pp. 250–480.
- Les constantes locales de l'équation fonctionnelle de la fonction L d'Artin d'une représentation orthogonale, Invent. Math. 35 (1976), 299–316.
- 32. A. Delzant, Définition des classes de Stiefel-Whitney d'un module quadratique sur un corps de caractéristique différente de 2, C. R. Acad. Sci. Paris 255 (1962), 1366–1368.
- 33. Michel Demazure and Pierre Gabriel, Groupes algébriques. Tome I: Géométrie algébrique, généralités, groupes commutatifs. Avec un appendice, Corps de classes local, par Michiel Hazewinkel, Masson & Cie, Éditeur, Paris; North-Holland Publishing Company, Amsterdam, 1970.
- Hélène Esnault, Bruno Kahn, and Eckart Viehweg, Coverings with odd ramification and Stiefel-Whitney classes, J. reine Angew. Math. 441 (1993), 145–188.
- Hélène Esnault, Bruno Kahn, Marc Levine, and Eckart Viehweg, The Arason invariant and mod 2 algebraic cycles, J. Amer. Math. Soc. 11 (1998), no. 1, 73–118.
- 36. Daniel Ferrand, Un foncteur norme, Bull. Soc. Math. France 126 (1998), no. 1, 1–49.
- 37. _____, Note on double coverings and binary quadratic forms, arXiv:0906.4084, 2009.
 38. A. Fröhlich, On the K-theory of unimodular forms over rings of algebraic integers, Quart. J. Math. Oxford Ser.
- (2) 22 (1971), 401–423.
 39. A. Fröhlich, Orthogonal representations of Galois groups, Stiefel-Whitney classes and Hasse-Witt invariants, J. Reine Angew. Math. 360 (1985), 84–123.
- William Fulton, Schubert varieties in flag bundles for the classical groups, Proceedings of Conference in Honor of Hirzebruch's 65th Birthday, Bar Ilan, 1993, vol. 9, Israel Mathematical Conference Proceedings, 1995.
- Ofer Gabber, Some theorems on Azumaya algebras, The Brauer group (Sem., Les Plans-sur-Bex, 1980), Lecture Notes in Math., vol. 844, Springer, Berlin, 1981, pp. 129–209.

- Ryan Garibaldi, Jean-Pierre Tignol, and Adrian R. Wadsworth, Galois cohomology of special orthogonal groups, Manuscripta Math. 93 (1997), no. 2, 247–266. MR 1464368 (98k:11039)
- Skip Garibaldi, Alexander Merkurjev, and Jean-Pierre Serre, Cohomological invariants in Galois cohomology, University Lecture Series, vol. 28, AMS, Providence, R.I., 2003.
- W.-D. Geyer, G. Harder, M. Knebusch, and W. Scharlau, Ein Residuensatz f
 ür symmetrische Bilinearformen, Invent. Math. 11 (1970), 319–328.
- 45. Stefan Gille, A transfer morphism for Witt groups, J. Reine Angew. Math. 564 (2003), 215–233.
- Stefan Gille, A Gersten-Witt complex for Hermitian Witt groups of coherent algebras over schemes. II. Involution of the second kind, J. K-Theory 4 (2009), no. 2, 347–377.
- 47. Jean Giraud, Cohomologie non abélienne, Springer-Verlag, Berlin, 1971.
- Alexander Grothendieck, Dix exposés sur la cohomologie des schémas, ch. VIII Classes de Chern et représentations linéaires des groupes discretsbala, pp. 215–305, North-Holland Publishing Company, Amsterdam; Masson & Cie, Éditeur, Paris, 1968.
- 49. Wei Ho, Orbit parametrizations of curves, Ph.D. thesis, Princeton University, September 2009.
- 50. Yogish I. Holla and Nitin Nitsure, Characteristic classes for GO(2N, C), Asian J. Math. 5 (2001), no. 1, 169–182.
- 51. Yogish I. Holla and Nitin Nitsure, Topology of quadric bundles, Internat. J. Math. 12 (2001), no. 9, 1005–1047.
- 52. N. Jacobson, Clifford algebras for algebras with involution of type D, J. Algebra 1 (1964), 288–300.
- J. F. Jardine, Universal Hasse-Witt classes, Algebraic K-theory and algebraic number theory, no. 83, Amer. Math. Soc., Providence, RI, 1989, pp. 83–100.
- 54. _____, Higher spinor classes, Mem. Amer. Math. Soc. 528 (1994).
- Kirti Joshi, Musings on Q(1/4): arithmetic spin structures on elliptic curves, Math. Res. Lett. 17 (2010), no. 6, 1013–1028.
- 56. Teruo Kanzaki, On bilinear module and Witt ring over a commutative ring, Osaka J. Math. 8 (1971), 485–496.
- M.M. Kapranov, On the derived categories of coherent sheaves on some homogeneous spaces, Invent. Math. 92 (1988), 479–508.
- M. Knebusch, Symmetric bilinear forms over algebraic varieties, Conference on Quadratic Forms-1976 (Kingston, Ont.) (G. Orzech, ed.), Queen's Papers in Pure and Appl. Math., no. 46, Queen's Univ., 1977, pp. 103–283.
- Manfred Knebusch, Grothendieck- und Wittringe von nichtausgearteten symmetrischen Bilinearformen, S.-B. Heidelberger Akad. Wiss. Math.-Natur. Kl. 1969/70 (1969/1970), 93–157.
- 60. Martin Kneser, Composition of binary quadratic forms, J. Number Theory 15 (1982), no. 3, 406-413.
- 61. M.-A. Knus, Pfaffians and quadratic forms, Adv. in Math. 71 (1988), 1–20.
- M.-A. Knus, M. Ojanguren, and R. Sridharan, Quadratic forms and Azumaya algebras, J. Reine Angew. Math. 303/304 (1978), 231–248.
- 63. M.-A. Knus and A. Paques, Quadratic spaces with trivial Arf invariant, J. Algebra 93 (1985), no. 2, 267–291.
- M.-A. Knus, R. Parimala, and R. Sridharan, On rank 4 quadratic spaces with given Arf and Witt invariants, Math. Ann. 274 (1986), no. 2, 181–198.
- M.-A. Knus, R. Parimala, and R. Sridharan, A classification of rank 6 quadratic spaces via pfaffians, J. reine angew. Math. 398 (1989), 187–218.
- M.-A. Knus, R. Parimala, and R. Sridharan, On compositions and triality, J. Reine Angew. Math. 457 (1994), 45–70.
- 67. M.-A. Knus, R. Parimala, and V. Srinivas, Azumaya algebras with involutions, J. Algebra 130 (1990), no. 1, 65-82.
- 68. Max-Albert Knus, Quadratic and hermitian forms over rings, Springer-Verlag, Berlin, 1991.
- Max-Albert Knus, Alexander Merkurjev, Markus Rost, and Jean-Pierre Tignol, *The book of involutions*, Colloquium Publications, vol. 44, AMS, 1998.
- Max-Albert Knus and Manuel Ojanguren, The Clifford algebra of a metabolic space, Arch. Math. (Basel) 56 (1991), no. 5, 440–445.
- Alexander Kuznetsov, Derived categories of quadric fibrations and intersections of quadrics, Adv. Math. 218 (2008), no. 5, 1340–1369.
- O. Laborde, Classes de Stiefel-Whitney en cohomologie étale, Bull. Soc. Math. France Suppl. Mem. (1976), no. 48, 47–51, Colloque sur les Formes Quadratiques (Montpellier, 1975).
- T. Y. Lam, The algebraic theory of quadratic forms, Benjamin/Cummings Publishing Co. Inc. Advanced Book Program, Reading, Mass., 1980, Revised second printing, Mathematics Lecture Note Series.
- 74. Max Lieblich, Moduli of twisted sheaves, Duke Math. J. 138 (2007), no. 1, 23–118.
- Ottmar Loos, Bimodule-valued Hermitian and quadratic forms, Arch. Math. (Basel) 62 (1994), no. 2, 134–142. MR 1255637 (95d:11040)
- J. S. Milne, Étale cohomology, Princeton Mathematical Series, no. 33, Princeton University Press, Princeton, N.J., 1980.
- 77. David Mumford, Theta characteristics of an algebraic curve, Ann. Sci. École Norm. Sup. 4 (1971), no. 4, 181–192.
- 78. Alexander Nenashev, Gysin maps in Balmer-Witt theory, preprint, K-theory preprint archive, 2005.
- 79. _____, Projective push-forwards in the Witt theory of algebraic varieties, preprint, K-theory preprint archive, 2007.
- M. Ojanguren, R. Parimala, and R. Sridharan, Ketu and the second invariant of a quadratic space, K-Theory 7 (1993), no. 6, 501–515.
- R. Parimala and W. Scharlau, On the canonical class of a curve and the extension property for quadratic forms, Recent advances in real algebraic geometry and quadratic forms (Berkeley, CA, 1990/1991; San Francisco, CA, 1991), Contemp. Math., vol. 155, Amer. Math. Soc., Providence, RI, 1994, pp. 339–350.
- 82. R. Parimala and R. Sridharan, Graded Witt ring and unramified cohomology, K-Theory 6 (1992), no. 1, 29-44.
- Monsurjectivity of the Clifford invariant map, Proc. Indian Acad. Sci. Math. Sci. 104 (1994), no. 1, 49–56, K. G. Ramanathan memorial issue.
- R. Parimala and R. Sridharan, Reduced norms and pfaffians via Brauer-Severi schemes, Contemp. Math. 155 (1994), 351–363.

- R. Parimala and V. Srinivas, Analogues of the Brauer group for algebras with involution, Duke Math. J. 66 (1992), no. 2, 207–237.
- Holger P. Petersson, Composition algebras over algebraic curves of genus zero, Trans. Amer. Math. Soc. 337 (1993), no. 1, 473–493.
- 87. S. Pumplün, Quaternion algebras over elliptic curves, Comm. Algebra 26 (1998), no. 12, 4357–4373.
- 88. David J. Saltman, Azumaya algebras with involution, J. Algebra 52 (1978), no. 2, 526–539.
- Jean-Pierre Serre, Revêtements à ramification impaire et thêta caractéristiques, C. R. Acad. Sci. Paris Sér. I Math. 311 (1990), no. 9, 547–552.
- 90. _____, Cohomologie galoisienne, 5th ed., 2nd print., Lecture Notes in Math., vol. 5, Springer-Verlag, Berlin, 1997.
- Jean-Pierre Serre, Cohomological invariants, Witt invariants, and trace forms, Cohomological invariants in Galois cohomology, Univ. Lecture Ser., vol. 28, Amer. Math. Soc., Providence, RI, 2003, Notes by Skip Garibaldi, pp. 1– 100.
- 92. Richard G. Swan, K-theory of quadric hypersurfaces, Ann. of Math. (2) 122 (1985), no. 1, 113–153.
- 93. J. Tits, Formes quadratiques, groupes orthogonaux et algèbres de Clifford, Invent. Math. 5 (1968), 19–41. MR 0230747 (37 #6307)
- 94. John Voight, Characterizing quaternion rings over an arbitrary base, J. Reine Angew. Math., to appear, 2010.
- 95. _____, Rings of low rank with a standard involution, Illinois J. Math., to appear, 2010.
- 96. Charles Walter, Grothendieck-Witt groups of triangulated categories, preprint, K-theory preprint archive, 2003.
- 97. Melanie Matchett Wood, Moduli spaces for rings and ideals, Ph.D. thesis, Princeton University, 2009.

EMORY UNIVERSITY, DEPARTMENT OF MATHEMATICS & CS, EMORY UNIVERSITY, 400 DOWMAN DRIVE NE W401, ATLANTA, GA 30322

E-mail address: auel@mathcs.emory.edu